

This electronic version (PDF) was scanned by the International Telecommunication Union (ITU) Library & Archives Service from an original paper document in the ITU Library & Archives collections.

La présente version électronique (PDF) a été numérisée par le Service de la bibliothèque et des archives de l'Union internationale des télécommunications (UIT) à partir d'un document papier original des collections de ce service.

Esta versión electrónica (PDF) ha sido escaneada por el Servicio de Biblioteca y Archivos de la Unión Internacional de Telecomunicaciones (UIT) a partir de un documento impreso original de las colecciones del Servicio de Biblioteca y Archivos de la UIT.

(ITU) نتاج تصوير بالمسح الضوئي أجراه قسم المكتبة والمحفوظات في الاتحاد الدولي للاتصالات (PDF)هذه النسخة الإلكترونية نقلاً من وثيقة ورقية أصلية ضمن الوثائق المتوفرة في قسم المكتبة والمحفوظات.

此电子版(PDF 版本)由国际电信联盟(ITU)图书馆和档案室利用存于该处的纸质文件扫描提供。

Настоящий электронный вариант (PDF) был подготовлен в библиотечно-архивной службе Международного союза электросвязи путем сканирования исходного документа в бумажной форме из библиотечно-архивной службы МСЭ.



Documentos de la Conferencia Mundial de Radiocomunicaciones (CMR-95) (Ginebra, 1995)

A fin de reducir el tiempo de carga, el Servicio de Biblioteca y Archivos de la UIT ha repartido los documentos de conferencias en varias secciones.

- Este PDF comprende los Documentos DL N° 1 a 17.
- La serie completa de documentos de la Conferencia comprende los Documentos N° 1 a 327, DL N° 1-17, DT N° 1-114.

UNIÓN INTERNACIONAL DE TELECOMUNICACIONES



CMR-95

CONFERENCIA MUNDIAL DE RADIOCOMUNICACIONES

Documento DL/1-S 23 de octubre de 1995 Original: inglés

Documento

GINEBRA, 23 DE OCTUBRE – 17 DE NOVIEMBRE DE 1995

ORDEN DEL DÍA DE LA REUNIÓN DE JEFES DE DELEGACIÓN

Lunes 23 de octubre de 1995 a las 9.30 horas (Sala II)

1.	Apertura por el Secretario General y designación del Presidente de la reunión	-
2.	Aprobación del orden del día de la reunión	DL/1
3.	Propuestas para la elección del Presidente de la Conferencia	. -
4.	Propuestas para la elección de los Vicepresidentes de la Conferencia	-
5.	Estructura de la Conferencia	DT/2
6.	Propuestas para la elección de los Presidentes y Vicepresidentes de las Comisiones	-
7.	Proyecto de orden del día de la primera sesión plenaria	DT/3
8.	Atribución de documentos a las Comisiones	DT/4
9.	Otros asuntos	-

Pekka TARJANNE Secretario General

UNIÓN INTERNACIONAL DE TELECOMUNICACIONES



CMR-95

CONFERENCIA MUNDIAL DE RADIOCOMUNICACIONES

Documento DL/2-S 23 de octubre de 1995 Original: español francés inglés

GINEBRA, 23 DE OCTUBRE – 17 DE NOVIEMBRE DE 1995

COMISION DE DIRECCIÓN

Proyecto

PROGRAMA GENERAL DE TRABAJO DE LA CONFERENCIA

Primera semana (23 - 27 de octubre)

Organización y comienzo de los trabajos de las Comisiones y Grupos de trabajo

Segunda semana (30 octubre - 3 noviembre)

Continuación de los trabajos de las Comisiones y Grupos de trabajo

Viernes 3

Fin de los trabajos del Grupo de trabajo de la Plenaria

Tercera semana (6 - 10 noviembre)

Continuación de los trabajos de las Comisiones y Grupos de trabajo

Viernes 10

Fin de los trabajos de los Grupos de trabajo de la Comisión 4

Cuarta semana (13 - 17 noviembre)

Lunes 13

Fin de los trabajos de los Grupos de trabajo de la Comisión 5

Martes 14

Fin de los trabajos de la Comisión 4 Fin de los trabajos de la Comisión 5

Informe de la Comisión 2 Informe de la Comisión 3

Miércoles 15

Primera lectura por la Plenaria de los últimos textos de las

Actas Finales

Jueves 16

Segunda lectura por la Plenaria de los últimos textos de las

Actas Finales

Viernes 17

Ceremonia de la firma y clausura

Nota 1 - Las sesiones plenarias se fijarán cada semana de la Conferencia, según sea necesario.

Nota 2 - Este programa puede cambiarse durante el curso de los trabajos de la Conferencia.



WRC-95

WORLD
RADIOCOMMUNICATION
CONFERENCE

Document DL/3(Rev.1)-E 29 October 1995 Original: English only

GENEVA.

23 OCTOBER

17 NOVEMBER 1995

Chairman, Drafting Group 4B1

SUMMARY OF THE CURRENT STATUS OF THE WORK OF 4B1 (RESOLUTION 46)

1. Footnote 1 in the Title of the Annex to Resolution 46

1.1 Application to earth stations in relation to terrestrial stations

It is understood that this part of the footnote ("where a pfd limit at the border of the territory of another administration (for an earth station) ... is exceeded") might have been intented to refer to the content of footnotes RR 608A and 608B, where the only reference to a pfd limit applicable to earth stations under Resolution 46 appears.

However, these two footnotes (608A and 608 B) are worded as absolute limits, rather than coordination thresholds.

Consequently, there is no scope for the application of the provision concerning earth stations in the footnote of the Title to the Annex to Resolution 46, and it should therefore be deleted.

There is agreement that the need for coordination between two administrations concerning the use of a frequency assignment by an earth station should be based either on a coordination distance (as currently implemented in Section III of Resolution 46, with possibly a reduced value for this distance) or on a method similar to that in Appendix 28. UK and USA were requested to develop a proposal along these lines, taking into account that Appendix 28 currently does not apply under 1 Ghz and that the need for a coordination distance would still exist in the case of aeronautical mobile earth stations.

1.2. Application to space stations

The remaining part of the footnote, relating to the space station coordination, should be transferred to Section II, under paragraph 2.5. In order to simplify Article 8, elimination of the sentence "coordination of space stations of the MSS with respect to terrestrial services is required only if the pfd prodiuced at the surface of the Earth's surface exceeds the limits in N°2566" should be envisaged (Liaison with Committee 5 needed).

2. Case of no response under 2.8B b)

The BR's explanation that the RRB rule of procedure on this point was based on similar provisions in Article 14 and Appendix 30.

It was also noted, however, that the text under 2.8B b) (in the CPM report) was taken out of context from a provision in Article 11 where the assistance of the Bureau has been requested and no reply was provided to the Bureau by the affected administration. On the other hand, how could assistance of the Bureau be requested when the list of affected administration is openned?

The possible solution to this problem, as contained in J/22/70) was noted. However, it would lead to twice the same process as under the current procedure, with no change in the type of warning sent to the affected administrations.

Another solution was identified in which a circular telegram would be sent to all administrations one month prior to the expiry date in order to attract their attention on the consequence of a lack of reply. This solution was retained as te nominal one for the development of the procedure.

The need to provide assurance on both sides was highlighted.

Taking into account the impossibility for the "terrestrial" administration to assess the potential for interference into the space station, the need for the "space" administration to make this assessment was mentionned.

3. Applicability of Resolution 46 to all the services sharing the same band with equal status

The RRB Rule of procedure on this point was supported, on the basis of the need for equity to all services sharing the same band with an equal status.

4. Treatment of GSO in the Title and in Section 2.2

The rationale provided in EUR/5/46-47 was presented. There was general agreement on this point, but the Russian Federation reserved its position until more progress has been made on the development of the new procedure.

5. Time limits

4B1 reviewed the time limits used in establishing the relative priorities between the various earth and terrestrial stations that may be involved in the coordination procedure. The conclusion was as follows:

- the provisions between earth stations should be separated from those between earth and terrestrial stations.
- a 3 years time limit should be included in 3.7 and 3.7.1 in all cases concerning earth stations
- concerning terrestrial stations, a 3 months time limit should apply in 3.7 and 3 years in 3.7.1.

It was agreed that 2.5.5 should be completed as proposed in EUR/5/47. to start the 3 year period in which terrestrial stations can be considered in the coordination at the time of publication of the RES46/C circular, which is referred to in 2.7.2.

Application of time limits to terrestrial stations under section II might be envisaged in the same fashion as currently foreseen under 3.7. and 3.7.1.

It was agreed that the implications of footnote 1 under 1.1, which refers to RR 1550, should be reviewed since RR 1550 refers back to Article 11, and not to Resolution 46. This could be addressed under Section V, where an alternative to RR 1550 should be included, taking into account CR/39, which supersedes the December 1994 Rule of Procedure on this point.

Upon receipt of the RES46/C special section, a six month period is foreseen in the current procedure in order for the administrations affected to make themselves known. Under the Article S9 procedure, a uniform period of 4 months is being considered. In order to allow smooth incorporation of Resolution 46 within the generic procedure of Article S9 at the appropriate time, the same period should be considered [four months].

6. Application of the methods and criteria adopted by the Radiocommunication Assembly

6.1. Coordination Region (doc 8/45)

The Japanese proposal in J/22/63 was noted, and it was agreed that such a reference might be inserted under paragraph 2.5, taking into account the comments from the BR on the applicability of Annex 1 to document 8/45.

6.2. Identification in the affected administrations within the Res46/C special section

In order to reduce the unnecessary workload for the Bureau, and as a simplification to the flowchart given in page 104 of the CPM report, it was agreed that, in the footnote to 2.7.2, the Bureau should only be requested to indicate the administrations having recorded assignments overlaping with the assignment proposed. This would allow to apply the pfd, FDP and SCP calculations only to those assignments for which administrations will have raised concerns during the relevant period after the RES46/C publication.

6.3. Application of the pfd, FDP and SCP methods

At the end of the [four] months period mentionned in 2.8, any administration which considers itself as affected should:

- either base its disagreement on its assignments which have been recorded in the MIFR
- or provide the specific parameters of its assignments currently in use or planned to be used within the next three years.

If these specific parameters are not communicated, for example because the exact location is not known or because the information is classified, it shall be assumed, for the purpose of assessing the need for coordination, that these assignments can be implemented anywhere on the territory of that administration, with the typical parameters mentionned in the relevant ITU-R recommendation.

On this basis, the BR will apply the pfd, FDP and/or SCP methods, as appropriate, to determine the need to effect coordination with that administration.

7. Assistance of the Bureau

In the CPM text, assistance of the Bureau is dealt with under ADD 1. by referring to the applicable parts of Article 11. However, these parts cross-reference other parts of Article 11 rather than the corresponding parts of Resolution 46. Two options are therefore possible:

- consider the general text proposed in Article S13, when this has been approved by WG 4B, or
- insert the complete text of Article 11 at the appropriate four places under Sections I, II, III and IV, with the appropriate cross-references.

8. Need to restart the Advance publication

Paragraph 1.2 will need updating on the basis of the decision to be taken by WG 4B in relation to the corresponding point under Article S9.

PROPOSED NEW DRAFT TO ANNEX TO RESOLUTION No. 46 (WARC-92) **COMMENTS** Interim Procedures for the Coordination and Notification of Frequency Assignments of [Non-Geostationary] Satellite Networks in Certain Space Services and the Other The text of Footnote I should be Services to Which the Bands are Allocated 1 reviewed by WRC-95. Footnote not applicable to earth Sections I. II and III apply to terrestrial services only in the case where a power-fluxstations (RR 608A and 608B are the density limit at the surface of the Earth (for a space station) or at the border of the territory of only cases of pfd limits for earth another administration (for an earth station) specified in a provision of the Radio Regulations isstations and contain absolute limits), exceeded. relocated in Section 2.5 for space stations. Section A. General Information

..1 The assistance of the IFRB BR can be requested in the application of the provisions of this annex.

ADD1:

The following provisions shall apply: RR 1054 to 1054C under Section I, RR 1088 to 1103 under Section II, RR1130 to 1144 under Section III in relation to terrestrial stations and earth stations operated in the opposite direction of transmission, and RR 1168 to 1181 under Section IV.

- A.2 In the absence of specific provisions relating to the evaluation of the interference, the calculation methods and the criteria should be based on relevant CCIR_ITU-R-Recommendations agreed by the administrations concerned either as a result of Resolution 703 (Rev.WARC-92) or otherwise. In the event of disagreement on a CCIR_ITU-R Recommendation or in the absence of such Recommendations, the methods and criteria shall be agreed between the administrations concerned. Such agreements shall be concluded without prejudice to other administrations.
- A.3 When applying the provisions of this Resolution for non-geostationary-satellite networks, administrations should provide the following information in addition to that of Appendix 3 or \ppendix 4:

MOD 1 :

i) right ascension of the ascending node;

ii) argument of perigee;

iii) active service arc.

- i) Orientation of the satellite transmitting and receiving antenna beams and their radiation pattern.
- ii) Type of modulation and multiple access.
- iii) Appropriate information required to calculate the affected region due to the MSS space stations [as defined in Annex 7].
- [iv] Maximum and average beam-peak e.i.r.p./4 kHz and e.i.r.p./1 MHz for each beam.]
- v) The satellite antenna gain $G(\Theta_e)$ as a function of elevation angle at a fixed point on the Earth. (To be provided either as part of Appendix 3 or as a formula to convert existing Appendix 3 data.)
- vi) The spreading loss (for a non-GSO satellite) as a function of elevation angle. (To be determined by equations or provided in graphical form.)

Reference to Article 11 sections where assistance to the BR is described in more

specific terms. (Type 2). To be reviewed

after adoption of Article S13

(Type 6)

To be referred to Committee 5 (unclear)

	elements required to properly characterize non-GSO satellites:	(Type 6)
$N_p =$	Number of orbital planes	
<u> </u>	Number of satellites in each orbital plane	
$\underline{\Omega_j}$	= Right ascension of the ascending node for the <i>j</i> -th orbital plane, measured counter clockwise in the equatorial plane from the direction of the vernal equinox to the point where the satellite makes its south-to-north crossing of the equator $(0^{\circ} \le \Omega_{j} < 360^{\circ})$.	
<u>i</u> j	= Inclination angle for the <i>j</i> -th orbital plane with respect to the reference plane, which is taken to be the Earth's equatorial plane $(0^{\circ} \le i_j < 180^{\circ})$.	
<u> </u>	= Initial phase angle of the <i>i</i> -th satellite in its orbital plane at reference time t=0, measured from the point of ascending node $(0^{0} \le \omega_{i} \le 360^{0})$.	
a =	Semi-major axis.	
e=	eccentricity (for circular orbit, e =0).	
<u>ω</u>	= argument of perigee, measured in the orbital plane, in the direction of motion, from the ascending node to perigee ($0^{\circ} \le \omega_p < 360^{\circ}$)	
Publication of Inform 1.1 An administrat intends to bring into use procedure described in p BureauBoard, not earlier	tion (or one acting on behalf of a group of named administrations) which a satellite network within a satellite system shall, prior to the coordination paragraphs 2.1 and 2.2, send to the International Frequency Registration than six years 1 and preferably not later than two years before the date of each satellite network, the information listed in Appendix 4.	Reference to RR 1550 is inadequate since it refers back to RR 1044 rather than this provision. Modification of
all also be sent to the	nts to the information sent in accordance with the provisions of paragraph 1.1 BureauBoard as soon as they become available. Modifications which are of ge significantly the character of the network may require recommencing the cedure. []	Section V should also be considered in relation to this point. An addition to this point may need to
BureauBoard shall publi also, when the weekly c telegram. The circular te geostationary satellite, the	the complete information sent under paragraphs 1.1 and 1.2, the ish it in a special section of its weekly circular within three months and shall ircular contains such information, so advise all administrations by circular elegram shall indicate the frequency bands to be used and, in the case of a he orbital location of the space station. When the Bureau_ —is not in a the time limit referred to above, it shall periodically so inform the the reasons therefor.	be added following WG 4B conclusions on the similar provision in Article S9.

Comments on Published Information If, after studying the information published under paragraph 1.3, any administration is of the opinion that interference which may be unacceptable may be caused to assignments of its existing or planned satellite networks or to assignments to its existing or planned terrestrial radiocommunication stations, it shall, within four months after the date of the weekly circular containing the complete information listed in Appendix 4, send the administration concerned its comments on the particulars of the interference to its existing or planned satellite systems or to its existing or planned terrestrial stations. A copy of these comments shall also be sent to the BureauBoard. If no such comments are received from an administration within the period mentioned above, it may be assumed that the administration has no basic objections to the planned satellite network(s) of the system on which details have been published. An administration sending information under paragraphs 1.1 and 1.2 shall, if requested by 1.4A an administration receiving information published under paragraph 1.3, provide the technical methods and criteria it proposes to use for the evaluation of the interference. An administration receiving information published under paragraph 1.3, may provide to the administration sending information under paragraphs 1.1 and 1.2 the technical methods and criteria it oposes to use for the evaluation of the interference. Resolution of Difficulties An administration receiving comments sent in accordance with paragraph 1.4 and administrations sending such comments shall endeavour to resolve any difficulties that may arise and shall provide any additional information that may be available. 1.5A In case of difficulties arising, the administration responsible for the planned network shall first explore all possible means of meeting its requirements without considering the possibility of adjustment to stations or networks of other administrations. If no such means can be found, the administration concerned may then request other administrations, either bilaterally or multilaterally, to mutually help resolve these difficulties. 1.5B An administration receiving a request under paragraph 1.5A shall, in consultation with the questing administration, explore all possible means of meeting the latter's requirements.

1.5C If, after following the procedure described in paragraphs 1.5A and 1.5B, there are unresolved difficulties, the administrations concerned shall jointly make every possible effort to

resolve these difficulties by means of mutually acceptable adjustments.

Results of Advance Publication

- 1.6 An administration on behalf of which details of planned satellite networks have been published in accordance with the provisions of paragraphs 1.1 to 1.3 shall, after the period of four months specified in paragraph 1.4, inform the <u>BureauBoard</u> whether or not comments provided for in paragraph 1.4 have been received and of the progress made in resolving any difficulties. Additional information on the progress made in resolving any remaining difficulties shall be sent to the <u>BureauBoard</u> at intervals not exceeding six months prior to the commencement of coordination or the sending of the notices to the <u>BureauBoard</u>. The <u>BureauBoard</u> shall publish this information in the special section of its weekly circular.
- 1.7 When, upon expiry of a period of six years plus the extension provided for in No. 1550 after the date of the publication of the special section referred to in paragraph 1.3, the administration responsible for the network has not submitted the Appendix 3 information for coordination under paragraph 2.1 or paragraph 2.2 or notification under No. 1488, as appropriate, the information published under paragraph 1.3 shall be cancelled after the administration concerned has been informed.

commencement of Coordination or Notification Procedures

- 1.8 When communicating to the <u>BureauBoard</u> the information referred to in paragraph 1.1, an administration may, at the same time or at a later time, communicate:
 - 1.8A the information required for the network coordination of a frequency assignment to a station of a satellite network in accordance with the provisions of paragraph 2.6, or
 - 1.8B the information required for notification of a frequency assignment to a station of a satellite network when coordination for that assignment is not required.
 - 1.8C Such coordination or notification information, as the case may be, shall be considered as having been received by the <u>BureauBoard</u> not earlier than six months after the date of receipt of the information referred to in paragraph 1.1.

Section II. Coordination of Frequency Assignments to a Station of a Satellite Network

Requirement for Coordination

2.1 Before an administration (or one acting on behalf of one or more named administrations) notifies to the <u>BureauBoard</u> or brings into use any frequency assignment to a station of a [non-geostationary-] satellite network, it shall effect coordination of the assignment with any other administration whose assignment to a station in a geostationary-satellite network, or whose assignment to a station of a non-geostationary-satellite network or whose assignment to a terrestrial station might be affected.

This paragraph 2.1 could be made general for the sake of simplifying the footnotes calling Resolution 46.

MOD 2:

2.2 Before an administration (or one acting on behalf of one or more named administrations) notifies to the <u>BureauBoard</u> or brings into use any frequency assignment to a station of a geostationary-satellite network, it shall effect coordination of the assignment with any other administration whose assignment to a station of a non-geostationary-satellite network might be affected.

This paragraph 2.2 could be merged with 2.1 for the sake of simplifying the footnotes calling Resolution 46.

- 2.3 Coordination under paragraphs 2.1 and 2.2 may be effected for a satellite network using the information relating to the space station, including its service area, and the parameters of one or more pical earth stations which may be located in all or part of the space station service area.
- 2.4 If a frequency assignment is brought into use before the commencement of the coordination procedure of paragraphs 2.1 or 2.2, when this coordination is required, the operation in advance of the receipt by the <u>BureauBoard</u> of the Appendix 3 information shall in no way afford any priority of the date.
- 2.5 Frequency assignments to be taken into account in the application of paragraphs 2.1 and 2.2 are those with a frequency overlap with the planned assignment, pertaining to the same service or to another service to which the band is allocated with equal rights, or a higher category of allocation (see Nos. 420 to 425 and 435), and which:

for space services, are:

- 2.5.1 in conformity with No. 1503, and
- 2.5.2 either recorded in the Master Register, or coordinated under the provisions of this Section or of Section II of Article 11, or
- 2.5.3 included in the coordination procedure with effect from the date of receipt by the BureauBoard, in accordance with paragraph 2.6 or No. 1074 or 1074A of Article 11, of the relevant information as specified in Appendix 3 and paragraph A3 above, as appropriate;
- or, for terrestrial services, are:
- 2.5.4 recorded in the Master Register with a favourable finding with respect to No. 1240, or
- 2.5.5 not notified but in use or planned to be brought into use within the next three years following the date of the publication referred to in paragraph 2.7.2.

The date from which this three year period is calculated should be clearly indicated (see Article 11).

<u> ADD 2 :</u>

2.5.6 Coordination of space services (space-to-Earth) with the terrestrial services of an administration is required only if the power flux density radiated over the territory of this administration exceeds the threshold levels appearing in [RR 2566][Article 28]

Relocation of the footnote in the Title of the Annex to Resolution 46, and of similar text in the Article 8 footnotes calling for Resolution 46.

Coordination of space services (Earth to space space-to-Earth) with the Fixed Service of an administration [in the bands [...]] is not required only if:]*

[2.5.-6.7] the frequency assignments referred to in 2.5.4 and or 2.5.5 use analogue modulation and the power flux density (pfd) radiated over the territory of this administration into these frequency assignments does not exceeds the thresholds specified in [RR 2566.1], or

(type 4).

[2.5.7-8. the frequency assignments referred to in 2.5.4 and or 2.5.5 use digital modulation and the Fractional degradation in Performance (FDP) caused into reference digital Fixed Service assignments located in the territory of this administration does not exceeds the thresholds specified in [RR 2566.1], or

[2.5.89. the application of the simulation method described in [SCP-] to Fixed Service frequency assignments located within the territory of this administration, leads to an interference level which does not exceeds the thresholds specified in [1]

* The WRC-95 would need to consider these modifications from the viewpoint of the need for these provisions and, if needed, the way to express the reference either as a provision of the radio Regulations, or as an ITU-R Recommendation.

The application of 2.5.8 could lead to difficulties for developing courntries. The difficulties that might be experienced by developing countries in the application of 2.5.9 can be overcome by submitting to the BR the specific parameters of the assignments concerned, or when those cannot be made available to the BR, by having the BR applying the method on typical parameters.

ADD 3:

2.5.109 No coordination under 2.1 or 2.2 is required:

Additions from RR 1066 to 1071 (type 1/2).

- a) when an administration proposes to notify or bring into use, within the service area of a satellite network, a typical earth station or an earth station which would not cause or suffer interference of a level greater than the typical earth station;
- [This paragraph 2.5.10 should be reviewed to make sure that these provisions, taken from Article 11, also apply in the case of coordinating with terrestrial services.]
- b) when the interference resulting from a modification to a frequency assignment which has previously been coordinated will not exceed that value agreed during coordination;
- c) when an administration proposes to notify or bring into use a new earth station which would not cause or suffer interference of a level greater than that which would be caused by an earth station belonging to the same satellite network and whose characteristics have been published in accordance with 2.7.2, or notified to the Bureau without coordination in those cases where coordination was not required;
- d) when, for a new frequency assignment to a receiving station, the notifying administration states that it accepts the interference resulting from the frequency assignments referred to in 2.5.1 to 2.5.3.
- e) between earth stations using frequency assignments in the same direction (either Earth-to-space or space-to-Earth).

Coordination Data

- 2.6 The administration seeking coordination shall send to the <u>BureauBoard</u> the information listed in Appendix 3 and paragraph A3 above, as appropriate.
- 2.7 On receipt of the complete information referred to in paragraph 2.6, the <u>BureauBoard</u> shall:
 - 2.7.1 examine this information with respect to its conformity with No. 1503; the date of its receipt shall be considered as the date from which the assignment will be taken into account for coordination;
 - 2.7.2 publish in the special section of its weekly circular, within three months, the information received under paragraph 2.6 and the result of the examination under paragraph 2.7.1¹. When the <u>BureauBoard</u> is not in a position to comply with the time limit referred to above, it shall periodically so inform the administrations giving the reasons therefor.
 - 2.7.3 include in the special section mentionned in paragraph 2.7.2 the names of the administrations having frequency assignments complying with the provisions of paragraphs 2.5 and 2.5.1, 2.5.2, 2.5.3 and 2.5.4, as applicable.

Purpose: avoid unnecessary
workload for the Bureau by
postponing the pfd, FDP and SCP
calculations to the next phase of the
procedure, after the concerned
administrations have commented.

MOD 3:

To help administrations identify services that may be affected, the <u>Bureau</u>Board shall also publish a list of administrations whose assignments comply with paragraphs 2.5 and 2.5.1 to 2.5.3 or paragraphs 2.5.1 and]2.5.4.[and yet do not comply with paragraphs 2.5.6.2.5.7 or 2.5.8.]

(Type 4).

Examination of Coordination Data and Agreement Between Administrations

On receipt of the special section referred to in paragraph 2.7.2, an administration shall promptly examine the matter with regard to interference which would be caused to the frequency assignments of its network or terrestrial stations, or caused by these assignments. In so doing, it shall have regard to the proposed date of bringing into use of the assignment for which coordination is sought. It shall then, within-six [four] months from the date of the relevant weekly circular, notify the administration seeking coordination of its agreement. If, however, the administration with which coordination is sought does not agree, it shall, within the same period, send to the administration seeking coordination the technical details of the networks or information on the terrestrial stations concerned upon which its disagreement is based, including the characteristics contained in Section C of Appendix 1 or Appendix 3 and in paragraph A3, as appropriate, which have not previously been notified to the BureauBoard, and make such suggestions as it may be able to offer with a view to a satisfactory solution of the problem. A copy of these comments shall also be sent to the Bureau Board. If the administration concerned has notified its disagreement within the same period, but the information on the fixed service stations upon which its disagreement is based cannot be provided, it shall be assumed that typical parameters, as assumed in [SCP], can be used to determine the need for coordination with this administration.

Addition aimed at facilitating the application of the SCP method when developing countries are involved or when the relevant information cannot be provided.

2.8A Affected administrations, as well as the administration seeking coordination, shall make all possible mutual efforts to overcome the difficulties in a manner acceptable to the parties concerned.

2.8.A1 One month prior to the expiry of the [four] month period mentionned in paragraph 2.8, the Bureau shall dispatch a circular telegramm to all administrations, bringing the matter to their attention.

ADD 4:

- 2.8 B When an administration has not responded either to the notifying administration or to the Bureau within the period of six [four] months referred to in paragraph 2.8, it shall be deemed that this administration has undertaken:
 - a) that no complaint will be made in respect of any harmful interference affecting the services rendered by its space radiocommunication or terrestrial stations which may be caused by the use of the assignment to a station of the satellite network for which coordination was requested;
 - fb) that its space radiocommunication or terrestrial stations will not cause harmful interference to the satellite network assignment for which coordination was requested.}

This provision is intended to attract the administrations attention on the consequences in the case of no reply (see 2.8B).

Addition similar to RR 1142 to 1144, to reflect the RRB Rule of Procedure on Res 46 (Type 3), with the specific difference that the lack of response to the Bureau, rather than to the requesting administration, triggers this situation.

Provision 2.8 B b) could have far reaching consequences and needs to be reviewed.

Results of Coordination

- 2.9 An administration which has initiated a coordination procedure under the provisions of paragraphs 2.1 to 2.6 shall communicate to the <u>BureauBoard</u> the names of the administrations with which agreement has been reached. The <u>BureauBoard</u> shall publish this information in the special section of its weekly circular.
- 2.10 An administration which has sought coordination, as well as any administration which has complied with the provisions of paragraph 2.8, shall communicate to the <u>BureauBoard</u> any modifications to the published characteristics of their respective networks or stations that were required to reach agreement on the coordination. The <u>BureauBoard</u> shall publish this information in accordance with paragraph 2.7.2, indicating that these modifications resulted from the joint efforts of the administrations concerned to reach agreement on the coordination.

Iotification of Frequency Assignments in the Event of Continuing Disagreement

2.11 In the event of continuing disagreement between an administration seeking to effect coordination and any administration with which coordination has been sought, the administration seeking coordination shall, except in the cases where the assistance of the <u>BureauBoard</u> has been requested, defer the submission of its notice concerning the proposed assignment by eight months from the date of publication of the special section referred to in paragraph 2.7.2, taking into account the provisions of No. 1496. When the assistance of the <u>BureauBoard</u> has been requested, the submission of the notice shall be deferred for a further three months.

MOD 4:

Section III. Coordination of Frequency Assignments to Earth Stations of a Non-Geostationnary Satellite Network in Relation to Terrestrial Stations and of a Satellite Network in Relation to other earth stations in the opposite direction of transmission

This section also applies to GSO networks (Type 1) and is proposed to be modified to include the case of feeder link earth stations in relation to other earth stations operating in the opposite direction of transmission (Type 5).

Requirement for Coordination

3.1 Before an administration notifies to the <u>BureauBoard</u> or brings into use any frequency assignment to a fixed earth station or to typical earth stations in a particular band allocated with equal rights to space and terrestrial radiocommunication services, it shall effect coordination of the assignment with each administration whose territory lies wholly or partly within the coordination area¹. The request for coordination may specify all or some of the frequency assignments to the associated space station, but thereafter each assignment shall be dealt with individually.

The coordination area is defined as the service area in which it is intended to operate the typical earth stations, extended in all directions by a coordination distance of [500 km], or as a circular zone with a radius of [500 km] centred on the coordinates of the fixed earth station. For a service area in which aircraft earth stations operate, the coordination area is the service area extended in all directions by a coordination distance of [1 000 km].

See section....

ADD 5:

3.1.1

No coordination under paragraph 3.1 is required when an administration proposes:

Addition from RR 1108 to 1111A (Type 2)

- 3.1.2
- a) to bring into use an earth station the coordination area of which does not include any of the territory of any other country;

3.1.3

b) to change the characteristics of an existing assignment in such a way as not to increase the interference to or from the terrestrial radiocommunication stations of other administrations or the earth station of other administrations operating in the opposite direction of transmission.

(Also type 5)

3.1.4

c) to bring into use a new frequency assignment to a receiving earth station and the notifying administration states that it accepts the interference resulting from existing and future terrestrial station assignments or earth station assignments operating in the opposite direction of transmission. In such case, administrations responsible for the terrestrial stations or the earth stations, are not required to apply the provisions of Section IV or Section III respectively, of this Article.

(Also Type 5)

Coordination Data

3.2 For the purpose of effecting coordination, the administration requesting coordination shall send to each administration concerned under paragraph 3.1 all pertinent information concerning the proposed frequency assignment as listed in Appendix 3, and an indication of the approximate date on which it is planned to begin operations. A copy of this information with the date of dispatch of the request for coordination shall also be sent to the <u>BureauBoard</u> for information.

Acknowledgement of Receipt of Coordination Data

3.3 An administration with which coordination is sought under paragraph 3.1 shall immediately acknowledge receipt of the coordination data.

Examination of Coordination Data and Agreement Between Administrations

3.4 On receipt of the coordination data, an administration shall, having regard to the proposed date of bringing into use of the assignment for which coordination was requested, promptly examine the matter with regard to both:

MOD 5

3.4.1 interference which would affect the service rendered by its terrestrial radiocommunication stations or by its earth stations in the opposite direction of transmission, operating in accordance with the Convention and these Regulations, or to be so operated prior to the planned date of bringing into service of the earth station assignment, or within the next three years, whichever is the longer, and

(Type 5)

MOD 6

3.4.2 interference which would be caused to reception at an earth station by the service rendered by its terrestrial radiocommunication stations or by its earth stations in the opposite direction of transmission, operating in accordance with the Convention and these Regulations, or to be so operated prior to the planned date of bringing into service of the earth station assignment, or within the next three years, whichever is the longer

(Type 5)

- 2.5 The administration with which coordination is sought shall, within four months from dispatch of the coordination data:
 - notify the administration requesting coordination of its agreement with a copy to the <u>BureauBoard</u>, indicating, where appropriate, the part of the allocated frequency band containing the coordinated frequency assignments; or

MOD 7

- 3.5.2. Send to that administration a request for inclusion in coordination of the terrestrial radiocommunication stations or the earth stations in the opposite direction of transmission mentioned in 3.4.1 and 3.4.2: or
- 3.5.23 notify that administration of its disagreement.

Addition taken from RR 1123. This case has been omitted in Resolution 46, although it is the only one to provide for the case of a terrestrial station to be put in service between the next 3 months and the next 3 years (Type 1)

MOD 8:

In the cases mentioned in paragraphs 3.5.2 and 3.5.3, the administration with 3.6 which coordination is sought shall send to the administration requesting coordination a diagram drawn to an appropriate scale indicating the location of those terrestrial radiocommunication stations or earth stations in the opposite direction of transmission which are or will be within the coordination area, together with all other relevant basic characteristics using Appendix 1_or Appendix 3, as appropriate, and make such suggestions as it may be able to offer with a view to a satisfactory solution of the problem.

(Type 1)

(Type 5)

When the administration with which coordination is sought sends to the administration seeking coordination the information required in the case of paragraph 3.5.23, a copy thereof shall also be sent to the Bureau Board. The Bureau shall consider as notifications in accordance with Section I of Article 12 or Section I of Article 13, as appropriate, only that information relating to existing radiocommunication stations, or to those to be brought into use within the next three [months][vears].

Addition from RR 1126 and RR 1536 (Type 1).

ADD 6:

<u>3.7.1</u>

When an agreement on coordination is reached, as a consequence of paragraphs 3.5 to 3.7. the administration responsible for the terrestrial stations or the earth stations in the opposite direction of transmission, may send to the Bureau the information concerning those stations covered by the agreement which are intended to be notified in accordance with Section I of Article 12 or Section I of Article 13. as appropriate. The Bureau shall consider as notifications in accordance with those Sections only that information relating to existing radiocommunication stations or to those to be brought into use within the next three years.

Addition from RR 1127 in order to avoid "paper" terrestrial stations (Type 1), with suitable modifications to accomodate earth stations in the opposite direction of transmission (Type 5).

DD 7

The periods refered to in paragraphs 3.4.1 and 3.4.2 lav be extended by agreement between the administrations concerned in order to take planned terrestrial and space networks into account. The earth station to earth station coordination may commence five and a half years before bringing into use these stations.

Addition fro mRR1120 (Type 1). The period of application of the coordination between two earth stations operating in opposite directions of transmissions needs to be reviewed.

Notification of Frequency Assignments in the Event of Continuing Disagreement

In the event of continuing disagreement between an administration seeking to effect coordination and an administration with which coordination has been sought, the administration seeking coordination shall, except in the cases where the assistance of the BureauBoard has been requested, defer the submission of its notice concerning the proposed assignment by six months from the date of the request for coordination, taking into account the provisions of No. 1496. When the assistance of the BureauBoard has been requested, the submission of the notice shall be deferred for a further three months.

Section IV. Coordination of Frequency Assignments to Terrestrial Stations for Transmission in Relation to Earth Stations of a Non-Geostationary Satellite Network

Requirement for Coordination

- 4.1 Before an administration notifies to the <u>BureauBoard</u>, or brings into use any frequency assignment to a terrestrial station for transmission within the coordination area¹ of an earth station of a non-geostationary satellite network, in a band allocated with equal rights to terrestrial radiocommunication services and space radiocommunication services (space-to-Earth), it shall effect coordination of the proposed assignment with the administration responsible for the earth stations with respect to the frequency assignments:
- The coordination area is defined as the service area in which it is intended to operate the typical earth stations, extended in all directions by a coordination distance of [500 km], or as a circular zone with a radius of [500 km] centred on the coordinates of the fixed earth station. For a service area in which aircraft earth stations operate, the coordination area is the service area extended in all directions by a coordination distance of [1 000 km].
 - 4.1.1 which are in conformity with No. 1503; and
 - 4.1.2 for which coordination has been agreed under paragraph 3.5.1, or

See section ...

Addition from RR 1151 (Type 1)

<u>ADD 8:</u>

which are to be taken into account for coordination with effect from the date of communication of the information referred to in paragraph 3.1.

ADD 9:

4.1.7

(2) No coordination under paragraph 4.1 is required when an administration proposes:

4.1.8

a) to bring into use a terrestrial station which is located, in relation to an earth station, outside the coordination area;

4.1.9

b) to change the characteristics of an existing assignment in such a way as not to increase the interference to the earth stations of other administrations;

4.1.10

c) to bring into use a terrestrial station within the coordination area of an earth station, provided that the proposed terrestrial station assignment is outside any part of a frequency band coordinated under 3.5.1 for reception by that earth station.

Addition from RR 1155 to 1158 (Type 2)

Coordination Data

4.2 For the purpose of effecting coordination, the administration requesting coordination shall send to each administration concerned under paragraph 4.1 all pertinent information. The request for coordination may specify all or some of the frequency assignments expected to be used within the next three years by stations of a terrestrial network wholly or partly within the coordination area of the earth stations. Thereafter each assignment shall be dealt with individually.

Acknowledgement of Receipt of Coordination Data

4.3 An administration with which coordination is sought under paragraph 4.1 shall immediately acknowledge receipt of the coordination data.

Examination of Coordination Data and Agreement Between Administrations

- 4.4 On receipt of the coordination data, the administration with which coordination is sought shall promptly examine the matter with regard to interference which would affect the services rendered by its earth stations covered by paragraph 4.1, which are operating or are to be operated within the next three years.
- 5 The administration with which coordination is sought shall, within an overall period of four months from dispatch of the coordination data, either notify the administration requesting coordination of its agreement to the proposed assignment or, if this is not possible, indicate the reasons for its objection and make such suggestions as it may be able to offer with a view to a satisfactory solution of the problem.

Notification of Frequency Assignments in the Event of Continuing Disagreement

4.6 In the event of continuing disagreement between an administration seeking to effect coordination and an administration with which coordination has been sought, the administration seeking coordination shall, except in the cases where the assistance of the <u>BureauBoard</u> has been requested, defer the submission of its notice concerning the proposed assignment by six months from the date of the request for coordination, taking into account the provisions of Nos. 1230 and 1496. When the assistance of the <u>BureauBoard</u> has been requested, the submission of the notice shall be deferred for a further three months.

Section V. Notification of Frequency Assignments

Notification of Assignments to Space Stations and Earth Stations

- 5.1 An administration shall, for the purpose of notifying an assignment to the <u>BureauBoard</u>, apply the provisions of Article 13. When applying the provisions of Article 13 to frequency assignment notices relating to space stations and earth stations covered by this Resolution, the BureauBoard shall:
 - 5.1.1 in applying No. 1504, also examine the notice with respect to its conformity with the provisions of paragraphs 2.1 or 2.2 relating to coordination of the use of the frequency assignment with the other administrations concerned;
 - 5.1.2 in applying No. 1505, also examine the notice with respect to its conformity with the provisions of paragraph 3.1 relating to coordination of the use of the frequency assignment with the other administrations concerned;
 - 5.1.3 in applying No. 1506, also examine the notice with respect to the probability of harmful interference when the coordination under paragraph 2.1 or 2.2 has not been successfully effected;
 - 5.1.4 in applying No. 1509, also examine the notice with respect to the probability of harmful interference when the coordination under paragraph 3.1 has not been successfully effected;
 - 5.1.5 not apply Nos. 1515 and 1516.
- 5.2 The examination under paragraph 5.1.3 or 5.1.4 shall take into account the frequency assignments for transmission or reception already recorded in the Master Register

Notification of Assignments to Terrestrial Stations

An administration shall, for the purpose of notifying an assignment to the <u>BureauBoard</u>, apply the provisions of Article 12. When applying the provisions of Article 12 the <u>BureauBoard</u> shall, in application of No. 1353, examine frequency assignment notices relating to terrestrial stations covered by this Resolution with respect to their conformity with the provisions of paragraph 4.1 relating to coordination of the use of the frequency assignment with the other administrations concerned

INTERNATIONAL TELECOMMUNICATION UNION



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/4-E 24 October 1995 Original: English only

GENEVA.

23 OCTOBER

17 NOVEMBER 1995

AD HOC GROUP 4B2

Report by the Chairman of Drafting Group 4B2

Further to the discussion held in Working Group 4B on 24 October 1995, the following is a summary of conclusions reached at that meeting.

1 Article S7 of the VGE Report

S7.1 NOC

S7.2 NOC

S7.3 [b) effecting a modification to the world plan (see Article S10);

VGE Note 1]

S7.4 NOC

S7.5 NOC

S7.5bis

F/54/2

ADD S7.5bis

If a frequency assignment is brought into use before commencement of the coordination procedure under Article S9, when coordination is required, or before notification when coordination is not required, the operation in advance of the application of the procedure shall, in no way, afford any priority of the date.

Reasons: Transfer from No. S9.6.2 and clarification to also cover non-GSO systems which begin operation before notification.

VGE Note 1

F/54/1B

SUP

VGE Note 1.

Reasons: Consequential to proposal on S10, see Document 5, EUR/5/24.

S7.6

RUS/7/6 MOD S7.6

If it is they are requested by any administration, particularly by the administration of a country in need of special assistance, the <u>Bureau and the Bureau Board</u> shall use such means at its their disposal as are appropriate in the circumstances and shall render the assistance requested in the application of the procedures of this Chapter.

Reasons: The word "Board" is added in order to clarify that administrations should be provided with assistance in the application of the procedures not only by the Radiocommunication Bureau but also by the Radio Regulations Board, and to bring the provision into line with No. S7.5.

S7.7 NOC

S7.8 NOC

2 Article S8

S8.1 NOC

S8.1.1 NOC

S8.2

EUR/5/19 MOD S8.2

A frequency assignment shall have the right to international protection from harmful interference when it is recorded in the Master Register with the indication that it is in accordance with a frequency allotment or assignment plan adopted by a Conference or as modified following successful application of the procedure, as applicable, for modification of the relevant plan. This protection shall be afforded to allotments or assignments appearing in a world-or a regional plan, to the extent provided by these Regulations-and the relevant agreement. In the case of a regional plan, such protection shall be limited to the countries which are parties to the relevant regional agreement; relations between parties and non-parties are governed by the provision of these Regulations. The right to international protection from harmful interference shall also be afforded to any assignment which is recorded in the Master Register with a favourable finding as a result of the successful application of Article S9. The right to international protection from harmful interference shall also be afforded to any frequency assignment to a station in the fixed service in the high frequency bands selected in accordance with Nos. S11.16 and S13.5 of Articles S11 and S13 respectively.

S8.3

USA/9/16 MOD S8.3

Any frequency assignment recorded in the Master Register with an indication that it is in accordance with the Table of Frequency Allocations and the associated provisions VGE Note 2 of these Regulations other provisions of the Radio Regulations with the exception of those provisions relating to the probability of harmful interference shall have the right to international recognition. For such an assignment, this right means that other administrations shall take it into account when making their own assignments in order to avoid harmful interference. The right to international recognition shall also be afforded to any frequency assignment to a station in the fixed service in the appropriate bands between 3 000 kHz and 27 000 kHz with class of operation B.

EUR/5/20 MOD S8.3

Any frequency assignment recorded in the Master Register with an indication that it is in accordance with the Table of Frequency Allocations and the associated provisions-VGE Note 2½ of these Regulations shall have the right to international recognition. For such an assignment, this right means that other administrations, recognizing Nos. S4.2 and S4.3 in particular, shall take it into account when making their own assignments in order to avoid harmful interference.

EUR/5/21 ADD \$8.3.1

- The "associated provisions" shall be identified and included in the Rules of Procedure.
- * Conclusion on S8.3 will be similarly reflected in S8.4.

S8.4

MOD based on the decision taken in S8.3 so far as the expression "associated provision" is concerned.

S8.5

S8.5 A non-conforming assignment, as defined in No. S8.4, shall neither cause harmful interference to any of the assignments described in Nos. S8.2 and S8.3 nor claim protection from harmful interference from any such assignment. In the event of a complaint of harmful interference to any assignment described in Nos. S8.2 and S8.3 from a non-conforming assignment the administration which notified the latter shall immediately investigate and eliminate the interference.

- 4 -CMR95/DL/4-E

S8.6

ARG/8/19 ADD S8.6

For all world or regional frequency allotment or assignment plans contained in Appendices to these Regulations, or adopted by world or regional conferences convened by the Union, the Bureau shall maintain master copies of the plans, incorporating any agreed modifications, and shall provide such copies in an appropriate form for publication by the Secretary-General when justified by circumstances.

To be transferred to Article S13.



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/5-E 25 October 1995 Original: English only

GENEVA, 23 OCTOBER

17 NOVEMBER 1995

Article S9

A.S9.1 NOC

A.S9.2 NOC

A.S9.3 SUP*

[A.S9.4 SUP**]

S9.1

F/54/16

MOD S9.1

Before initiating any action under this Article in respect of frequency assignments for a satellite network or a satellite system, an administration, or one⁵ acting on behalf of a group of named administrations, shall prior to the coordination procedure described in Section II below, where applicable, send to the Bureau a general description of the network or system for advance publication in the Weekly Circular not earlier than six years and preferably not later than two years before the planned date of bringing into use of the network or system (see also No. S11.44). The characteristics to be provided for this purpose are listed in Appendix S4. The coordination or notification information in cases where coordination under Section II of Article S9 is not required may also be communicated to the Bureau at the same time; it shall be considered as having been received by the Bureau-not earlier than six months after the date of receipt of the complete information for advance publication in cases where coordination under Section II of Article S9 is required and six months after the date of publication of the advance publication information in cases where coordination under Section II of Article S9 is not required.

^{*} Pending the results on Resolution 46.

^{**} Pending decision on Article S10.

- 2 -CMR95/DL/5-E

Reasons: Clarification: (1) incorporation of the substance of the first sentence of S9.5; (2) treatment of non-GSO network without coordination procedure (see also No. 1058C).

USA/9/18 MOD S9.1

Before initiating any action under this Article in respect of frequency assignments for a satellite network or a satellite system, an administration, or one⁵ acting on behalf of a group of named administrations, shall prior to the coordination procedure described in Section II below, where applicable, send to the Bureau a general description of the network or system for advance publication in the Weekly Circular not earlier than six years and preferably not later than two years before the planned date of introduction of the network or system (see also No. S11.44). The characteristics to be provided for this purpose are listed in Appendix S4. The coordination or notification information may also be communicated to the Bureau at the same time; it shall be considered as having been received by the Bureau not earlier than six months after the date of receipt of the information for advance publication where coordination is required by Section II. In the case where coordination is not required by Section II, notification shall be considered as received by the Bureau not earlier than six months after the date of publication of the advance publication information.

S9.1.1

Whenever under this provision an administration acts on behalf of a group of named administrations, all members of that group retain the right to respond in respect of their own networks or systems.

S9.2

RUS/7/7 MOD S9.2

Amendments to the information sent in accordance with the provisions of No. **S9.1** shall also be sent to the Bureau as soon as they become available. Modifications which are of such nature as to significantly change the character of the network may require recommencing the advance publication procedure. A list of modifications which are of such nature as to significantly change the character of the network is given in the Rules of Procedure.

F/54/17 MOD S9.2

Amendments to the information sent in accordance with the provisions of No. **S9.1** shall also be sent to the Bureau as soon as they become available. Modifications which are of such nature as to significantly change the character of involve the use of an additional frequency band to the network mayshall require-recommencing the advance publication-procedure.

Reasons: Clarification taking into account the existing Rules of Procedure (Part A1, AR11, No. 1043).

S9.2A

F/54/18

ADD S9.2bis

If the information is found to be incomplete, the Bureau shall immediately seek from the administration concerned any clarification and information not provided.

Reasons: This provision exists in the current Radio Regulations and should not be transferred to the Rules of Procedure. Indeed, the processing of incomplete advance publication information is crucial for initiating the registration procedure because of the relation between the receipt of complete advance publication information and the date on which networks are then taken into consideration either for purposes of the coordination process in those cases where a formal coordination might be required or for notification when no formal coordination is required (e.g. non-GSO satellite networks or systems in frequency bands not subject to Resolution 46).

S9.3

F/54/20A MOD S9.3

Upon receipt of the Weekly Circular containing information published under No. S9.1, if an administration believes that interference which may be unacceptable may be caused to its existing or planned satellite or terrestrial lbis networks or systems it shall within four months of the date of the Weekly Circular communicate its comments and shall, at the same time, give to the publishing administration its comments on the particulars of the interference foreseen to its existing or planned-satellite systems. A copy of these comments shall also be sent to the Bureau. Thereafter both administrations shall endeavour to cooperate in joint efforts to resolve any difficulties, with if necessary requested by either of the parties, with the assistance of the Bureau, and shall exchange any additional relevant information that may be available. If no such comments are received from an administration within the period mentioned above, it may be assumed that the administration has no basic objections to the planned satellite network(s) of that system on which details have been published.

S9.3.1bis

F/54/21

ADD S9.3.1bis

Terrestrial networks or systems to be taken into account are only those for which the requirement to coordinate is included in a footnote of the Table of Frequency Allocations referring to No. **S9.11bis** and **S9.21**.

Reasons:

- 1 Particulars of the foreseen interference are also required for terrestrial networks or systems, however restricted to Resolution 46 bands.
- 2 To provide for simple bilateral contact between the parties involved unless at least one wishes otherwise.
- 3 At this stage of the procedure, only comments on the particulars of the interference can be provided due to the general nature of the information available.
- 4 The last sentence of No. 1047 might be reintroduced in S9.3 as useful information for administrations.

S9.4

RUS/7/9 MOD S9.4

In the case of difficulties, the administration responsible for the planned network shall explore all possible means to resolve the difficulties without considering the possibility of adjustment to networks of other administrations, especially existing networks whose frequency assignments are recorded in the Master Register. If no such means can be found, the administration responsible for the planned network may request the other administrations to explore all possible means to meet its requirements. The administrations concerned shall make every possible effort to resolve the difficulties by means of mutually acceptable adjustments to their networks.

USA/9/22 MOD S9.4

In the case of difficulties, the administration responsible for the planned satellite network shall explore all possible means to resolve the difficulties without considering the possibility of adjustment to networks of other administrations. If no such means can be found, the administration responsible for the planned network may request the other administrations to explore all possible means to meet its requirements. The administrations concerned shall make every possible effort to resolve the difficulties by means of mutually acceptable adjustments to their networks. An administration on behalf of which details of planned satellite networks have been published in accordance with the provisions of No. S9.2B shall, after the period of four months, inform the Bureau of the progress made in resolving any difficulties. A further report, if necessary, should be provided prior to the commencement of coordination or the sending of notices to the Bureau. The Bureau shall publish this information verbatim in the Special Section of its Weekly Circular.

F/54/22 MOD S9.4

In the case of difficulties, the administration responsible for the planned <u>satellite</u> network shall explore all possible means to resolve the difficulties without considering the possibility of adjustment to networks of other administrations. If no such means can be found, the administration responsible for the planned network may request the other administrations to

- 5 -CMR95/DL/5-E

explore all possible means to meet its requirements. The administrations concerned shall make every possible effort to resolve the difficulties by means of mutually acceptable adjustments to their networks. An administration on behalf of which details of planned satellite networks have been published in accordance with the provisions of No. S9.2ter shall after the period of four months inform the Bureau of the progress made in resolving any difficulties. A further report, if necessary shall be provided prior to the commencement of coordination or the sending of notices to the Bureau.

Reasons: Reintroduction of No. 1056.

S9.5 F/54/23 MOD S9.5

Action under No. S9.1 shall be taken not earlier than six years and preferably not later than two years before the planned date of introduction of the network or system (see also No. S11.43). A response under No. S9.3 shall be made as soon as possible and in any case within four months after the publication of the relevant Weekly Circular. The Bureau shall inform all administrations of the list of administrations having sent comments under No. S9.3 and provide a summary of the comments received. These measures shall be taken solely for the purposes of informing all administrations of developments in the use of space radiocommunications and minimizing any difficulties that might otherwise arise during the coordination stage. (See also S11.47.)

Reasons:

- 1 Consequential to MOD S9.1; duplication with S9.3.
- 2 Reintroduction of RES46/B or AR11/B type circulars. Information of interest particularly in the registration cases for which there is no requirement for formal coordination before the notification.
- 3 Transfer to S9.5bis for sake of clarification.

S9.5A B/17/26 ADD S9.5A

When, upon expiry of a period of six years plus the extension provided for in No. S11.44 after the date of the publication of the Weekly Circular referred to in No. S9.1, the administration responsible for the network has not submitted the Appendix S4 information for coordination under No. S9.29 or for notification under No. S11.2, as appropriate, the information published under No. S9.1 shall be cancelled after the administration concerned has been informed.

- 6 -CMR95/DL/5-E

F/54/24

ADD S9.5bis

The procedure of Section I shall be taken solely for the purposes of informing all administrations of developments in the use of space radiocommunications and minimizing any difficulties that might otherwise arise during the coordination stage.

Reasons: Clarification.

INTERNATIONAL TELECOMMUNICATION UNION



WORLD RADIOCOMMUNICATION CONFERENCE Document DL/6-E 26 October 1995 Original: English

GENEVA, 23 OCTOBER

17 NOVEMBER 1995

AD HOC GROUP 5C1

TOPICS TO BE CONSIDERED BY AD HOC GROUP 5C1

- A. Feeder Links in Frequency Bands from 4 GHz to 8 GHz
- A.1. Feeder Links in Bands Around 5 GHz (Summary of Proposals)
- A.1.1. Bands to be Designated for Non-GSO/MSS Feeder Links

FREQUENCY BAND (MHz)	PROPOSAL	COMMENTS
5150-5250 (11)	EUR/5/57,59	
	ANNEX 1 TO CAN/18	not explicit
	J/22/96,97,100	
	THA/57/5	
	CHN/62/38B	
	CME/73/3	
5150-5250 (⋂ Џ)	RUS/7/34,37	
5100-5250 (11)	MEX/80/10	
5090(1)-5250 (1)	USA/9/17 ⁵ 5,177,179,180	5090-5150: no new assignments after 2010; secondary after 2015
	B/17/173,174,175,177	
	AUS/19/12A,12B,12D	5091-5150: secondary after 2015
	KOR/44/3A,3B,3C,4,5	5090-5150: secondary after 2015

NOTE 1 Proposals MLI/30/8, SEN/39/7 and ICAO (Doc. 24, sections 2.4.5.1, 2.4.5.2) propose that allocations remain unchanged in 5000-5150 and that any additional allocation in 5150-5250 only enter into force after sharing criteria have been developed.

NOTE 2 Proposal NZL/29/5 states that use of the 5 GHz band for feeder links is acceptable.

NOTE 3 Proposal PHL/75/8 supports RR provision No.796 (MLS precedence in 5000-5250 MHz).

- 2 -CMR-95/DL/6-E

A.1.2. RR 2613

Most proposals state that RR 2613 does not apply in the allocations to be designated for use by non-GSO/MSS feeder links. However, Docs. J/22, KOR/44, CHN/62, MEX/80 do not refer to RR2613. Proposals which address the definition of the GSO/FSS networks with respect to which RR 2613 is waived are:

EUR/5/59: ADD 797C - GSO/FSS networks received by the ITU-BR after [xx November 1995].

CME/73/3: ADD 797C - GSO/FSS networks advanced published after 17 November 1995.

A.1.3. Aeronautical Mobile-Satellite (R) Service (Footnote 733)

USA/9/176: MOD 733 - AMS(R)S allocation limited to 5000-5150 MHz.

A.1.4. Precedence of MLS (Footnote 796)

Proposals are consequential to those in A.1.1.

A.1.5. Fixed-Satellite Service and Inter-Satellite Service Used in Conjunction with the Aeronautical Radionavigation and/or Aeronautical Mobile (R) Service (Footnote 797)

USA/9/178: MOD 797 - allocation limited to 5000-5150 MHz.

CAN/18/Annex 1: SUP 797

J/22/98: MOD 797 - stations brought into use after 17 November 1995 shall not claim protection or cause harmful interference.

A.1.6. Radiodetermination-Satellite Service (space-to-Earth) in the Band 5150-5216 MHz (Footnote 797A)

EUR/5/58: MOD 797A - Radiodetermination-Satellite Service feeder links brought into use after [xx November 1995] shall claim protection or cause harmful interference.

RUS/7/36: MOD 797A - Non-GSO/MSS feeder links shall not cause harmful interference to Radiodetermination-Satellite Service feeder links brought into use before xx November 1995 or to aeronautical radionavigation systems operating in 5000-5150 MHz.

B/17/176: SUP 797A

AUS/19/12C: SUP 797A

- 3 -CMR-95/DL/6-E

J/22/99: MOD 797A - Radiodetermination-Satellite Service space stations brought into use after [xx November 1995] shall claim protection or cause harmful interference.

A.1.7. Mobile Service in the Band 5150-5250 MHz (Footnote 797B)

No proposals for change

A.2. Feeder Links in Bands Around 6/7 GHz (Summary of Proposals)

A.2.1. Bands to be Designated for Non-GSO/MSS Feeder Links

FREQUENCY BAND (MHz)	PROPOSAL	ASSOCIATED CONDITIONS
7025-7075 (↓ 1)	RUS/7/39	RUS/7/40,41,42
6975-7075 (↓)	J/22/101	J/22/102,103
	THA/57/5	
	CHN/62/40A,40B	
	CME/73/5	·
	MEX/80/11	
6625-6725 (↓)	CAN/18/Annex 1	ADD 809A, 809B
6875-7075 (∜)	B/17/179	B/17/180,181
	KOR/44/6A,6B	KOR/44/6C
6825-7075 (∜)	EUR/5/60	
6725-7075 (↓)	INS/56/5	
6700-7075 (∜)	AUS/19/13A	AUS/19/13B,14
6650-7075 (↓)	USA/9/182	USA/9/183,184,185

NOTE 1 Proposal NZL/29/5 states that use of the 6/7 GHz band for feeder links using reverse band working is acceptable.

NOTE 2 Proposal PHL/75/7 supports reverse band working in the 6/4 GHz band.

B. Feeder Links in Frequency Bands from 8 GHz to 15 GHz

B.1. Bands to be Designated for Non-GSO/MSS Feeder Links

- 4 -CMR-95/DL/6-E

FREQUENCY BAND (MHz)	PROPOSAL	· ASSOCIATED CONDITIONS
NONE	RUS/7/p.15	
	J/22/1	
	THA/57/6	
10.70-10.95 (11)	AUS/19/16A	AUS/19/16B,16C
11.20-11.45 (ft)		
	USA/9/186	USA/9/187,188,189
12.75-13.25 (∜)	USA/9/190	USA/9/191,192

NOTE 1 USA/9/Addendum 13 states that the band 13.75-14.00 GHz is not viable for space-to-Earth non-GSO/MSS feeder links.

J. ALBUQUERQUE
Chairman of Ad hoc Group 5C1, # 400

INTERNATIONAL TELECOMMUNICATION UNION

WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Addendum 1 to Document DL/7-E 30 October 1995 Original: English only

GENEVA, 23 OCTOBER

17 NOVEMBER 1995

AD HOC GROUP 5B1

Chairman, ad hoc Group 5B1

PROPOSALS ON TECHNICAL CONSTRAINTS (AGENDA ITEM 2.1.a)

Please replace Annex 4 to Document DL/7 by the attached Annex 4 and add the following new Annexes 5, 6 and 7.

ANNEX 4 Article [28] S21

EUR/5/42, B/17/161-164, CAN/18/45, 46, 49, 50, IRN/25/16, CPM Report (Chapter 2, Section I, Part A.2, § 1.4.8 and § 1.4.9)

For reference: page 47 of EUR/5

ANNEX 5 Comparison of proposals relating to RR731E

ANNEX 6 Comparison of proposals relating to RR746B

ANNEX 7 Comparison of proposals relating to RR753F

M. Murotani (Box 230) Chairman of ad hoc Group 5B1

[•] For reasons of economy, this document is printed in a limited number of copies. Participants are therefore kindly asked to bring • their copies to the meeting since no others can be made available.

EUR/5/42 **ADD** 2556.1

(See table.)

Reasons: See EUR/5/40.

Frequency Band (MHz)	Service to be protected	Coordination Threshold Factors				
		Geostationary	Space Stations	Non-Geostatio	onary Space Sta	ations
		pfd calculation (NOTE 2)	factors	pfd calculation (NOTE 2)	n factors	FDP (in 1 MHz)
		P dB(W/m²) in 4 kHz	r dB/deg	P dB(W/m²) in 4 kHz	r dB/deg	(NOTE 1)
2 160 - 2 200	Fixed service	-152	0.5	-147	0.5	
2 483.5 - 2 500	analogue	-152	0.5	-150	0.65	
2 160 - 2 200	Fixed service	-152	0.5	- 1 m		25 %
2 483.5 - 2 500	digital	-152	0.5	-150	0.65	
2 160 - 2 200	Other terrestrial	-152	0.5	-152	0.5	
2 483.5 - 2 500	services	-152	0.5	-152	0.5	

NOTE 1 – The definition for calculation of FDP (Fractional Degradation in Performance) is contained in the relevant ITU-R Recommendations.

NOTE 2 – The following formulae are used for pfd:

 $P dB(W/m^2/4 kHz)$

for $0^{\circ} <= \delta <= 5^{\circ}$

 $P + r(\delta-5) dB(W/m^2/4 \text{ kHz})$ for $5^{\circ} < \delta < 25^{\circ}$

 $P + 20r dB(W/m^2/4 kHz)$ for $25^{\circ} \le \delta \le 90^{\circ}$

The limits relate to the power flux-density which would be obtained under assumed freespace propagation conditions.

B/17/161 ADD

TABLE [AR28bis]

PFD or FDP thresholds for non-GSO MSS satellites

Frequency band	Service	Limit in dB(W/m ² /4 kHz) for angle arrival δ above the horizontal pla		_
		0° - 5°	5° - 25°	25° - 90°
1 492 - 1 525	Mobile-satellite	-1526, 7	-152 + 0.5 $(\delta-5)^{6, 7}$	-1426, 7
2 160 - 2 170 (R2)	Mobile-satellite	-147 ⁷	-147 + 0.5 $(\delta-5)^7$	-137 ⁷
2 170 - 2 200	Mobile-satellite	-1477	-147 + 0.5 $(\delta-5)^7$	-137 ⁷
2 483.5 - 2 500	Mobile-satellite	-1508	-150 + 0.65 $(\delta-5)^8$	-1378

Reasons: To take into account the results of the work of the ITU-R in the cases of the bands 1 492 - 1 525 MHz, 2 160 - 2 200 MHz and 2 483.5 - 2 500 MHz.

B/17/162

ADD S21.16.6

⁶ The power-flux density values to protect analogue systems is to be determined for this band for each non-geostationary mobile-satellite system proposed. Until such time as the new values are available, the values given in this box will apply.

Reasons: Same as in proposal B/17/161.

ADD S21.16.7 The power flux-density values specified for the bands 1 492 - 1 530 MHz and 2 160 - 2 200 MHz provide full protection for analogue radio-relay systems from a non-geostationary mobile-satellite system employing narrow-band TDMA/FDMA techniques. For digital radio-relay systems an FDP (fractional degradation of performance) value of 25% is applicable. The method for calculating the FDP for a fixed service network is contained in the ITU-R Recommendations.

Reasons: Same as in proposal B/17/161.

B/17/164

ADD S21.16.8

8 The power flux-density values given in this box provide full protection for analogue radio-relay systems from multiple non-geostationary mobilesatellite systems employing CDMA techniques. These power flux-density values will not provide full protection for existing digital fixed systems in all cases. However, these power flux-density values are considered to provide adequate protection for digital fixed systems designed to operate in this band,

where high power ISM and possible low power applications are expected to produce a relatively high interference environment.

Reasons: Same as in proposal B/17/161.

CAN/18/45 ADD 2560A

(2bis)Coordination threshold:

- a) Coordination is required with the fixed service if the power-flux density at the Earth's surface produced by emissions from a space station of a non-geostationary system, including emissions from a reflecting satellite, for all conditions and for all methods of modulation, exceeds the following values:
 - -150 dB(W/m²) in any 4 kHz band for angles of arrival between 0 and 5 degrees above the horizontal plane;
 - -150 + 0.65 (δ -5) dB(W/m²) in any 4 kHz band for angles of arrival δ (in degrees) between 5 and 25 degrees above the horizontal plane;
 - -137 dB(W/m²) in any 4 kHz band for angles of arrival between 25 and 90 degrees above the horizontal plane.

These values relate to the power-flux density which would be obtained under assumed free-space propagation conditions.

CAN/18/46 ADD 2560B

b) The coordination thresholds given in No. **2560A** apply in the frequency band 2 483.5 - 2 500 MHz to the mobile-satellite service (space-to-Earth) and the radiodetermination-satellite service (space-to-Earth).

Reasons: To implement the findings of CPM-95 (see Section 3.6.4.8, Chapter 2, Section 1, Part C of the CPM-95 Report).

NOTE – Canada is considering the possibility of creating a single provision in Article 28, Article S21 and Appendix S5 which would encompass such coordination thresholds as provisions distinct from the regular pfd limits of the current Article 28.

and the second of the second o

Control of the second

Consequential modifications to Article S21 and Table [AR28]

TABLE [AR28]

Frequency band	Service	Limit in dB(W/m ²) for angle of arrival above the horizontal plane			Reference
		0° - 5°	5° - 25°	25° - 90°	bandwidth
<u>2 483.5 - 2</u> <u>500</u>	Mobile-Satellite (S-E)	<u>-150</u> <u></u> €	$\frac{-150 + 0.65 (\delta - 10)}{5)6}$	<u>-137</u> <u>6</u>	<u>4 kHz</u>
	Radiodetermination -Satellite (S-E)				

CAN/18/49 MOD

CAN/18/50

ADD S.21.16.6

Reasons: Implement the findings of CPM-95 (see CPM-95 Report, Section 3.6.4.8, Chapter 2, Section 1, Part C).

⁶ These values are to be used to determine whether coordination is required between the non-geostationary space stations and stations in the fixed or mobile services.

IRN/25/16 ADD 2556.1

Frequency band (MHz)	Service to be protected	Coordination threshold factors				
•		Geostationary space stations		Non-geostationary space stations		
		pfd calculation (NOTE 2)	factors	pfd calculation (NOTE 2)	factors	FDP (in 1 MHz)
		P dB (W/m ²) in 4 kHz	r dB/deg	P dB (W/m ²) in 4 kHz	r dB/deg	(NOTE 1)
2 160 - 2 200	Fixed service	-152	0.5	-147	0.5	
2 483.5 - 2 500	analogue	-152	0.5	-150	0.65	
2 160 - 2 200	Fixed service	-152	0.5			25 %
2 483.5 - 2 500	digital	-152	0.5	-150	0.65	
2 160 - 2 200	Other terrestrial	-152	0.5	-152	0.5	
2 483.5 - 2 500	services	-152	0.5	-152	0.5	

NOTE 1 – The definition for calculation of FDP (fractional degradation in performance) is contained in the relevant ITU-R Recommendations.

NOTE 2 – The following formulae are used for pfd:

 $P dB(W/m^2/4 kHz)$

for $0^{\circ} \leq \delta \leq 5^{\circ}$

 $P + r(\delta-5) dB(W/m^2/4 kHz)$

for $5^{\circ} < \delta < 25^{\circ}$

 $P + 20r dB(W/m^2/4 kHz)$

for $25^{\circ} \le \delta \le 90^{\circ}$

The limits relate to the power-flux density which would be obtained under assumed free-space propagation conditions.

Reasons: See IRN/25/15.

CPM Report (Chapter 2, Section I, Part A.2)

1.4.8 Sharing with the mobile service (FPLMTS)

MSS (space-to-Earth) and (Earth-to-space) networks cannot share frequencies with the terrestrial component of FPLMTS in the same and adjacent geographic areas. However, sharing with MSS (space-to-Earth) may be feasible in a non-co-frequency, co-coverage environment.

1.4.9 Sharing with other mobile services

The criteria under development for sharing between the MSS (space-to-Earth) and receiving stations in the fixed service may also adequately protect mobile services, although further study is needed. Some systems in the mobile service (e.g. transportable equipment used for

electronic new gathering) appear to be similar to certain types of fixed systems (e.g. point-to-multipoint systems). It is noted that in RR Article 28 (pfd limits on transmitting space stations) and Appendix 28 (parameters for calculation of coordination area), no distinction is made between systems in the terrestrial services (e.g. fixed and mobile). Thus, for the time being, it is reasonable to assume that sharing between general mobile services and the MSS is no more constraining than sharing between MSS and the fixed service.

FOR REFERENCE (extract from EUR/5):

C Revision of Resolution 46

Background

Above 1 GHz, Resolution 46 applies in the following bands (space-to-Earth):

- 1 492 1 525 MHz (No. **723C**)
- 1 525 1 530 MHz (No. **726D**)
- 1 530 1 559 MHz (No. **726D**)
- 1 613.8 1 626.5 MHz (No. **731F**, secondary)
- 2 160 2 200 MHz (No. **746B**)
- 2 483.5 2 500 MHz (No. **753F**)
- 2 670 2 690 MHz (No. **746A**, plus Article **14** until 2005)
- 2 655 2 670 (No. **766**, plus Article **14**)

In all these allocations to MSS above 1 GHz, Section II paragraph 2.2 of Resolution 46 applies in case of exceedance of pfd levels, except in the band 1 530 - 1 559 (original Inmarsat bands), where there is no pfd threshold since the band is not shared on an equal status with terrestrial services, 1 613.8 - 1 626.5 MHz (secondary status) and 2 655 - 2 690 MHz (under Article 14 at least until 2005).

In summary, in the bands in which the use of the MSS is not limited by a secondary status or by Article 14, and which are shared on an equal basis with terrestrial services, coordination with terrestrial services under Section II paragraph 2.2 of Resolution 46 is foreseen in case of pfd threshold exceedance. The validity of this provision is therefore a general one and the title of Resolution 46 should be amended to reflect this situation.

If this amendment is decided by WRC-95, a consequential change would be the simplification of Nos. Nos. 723C, 726D, 746B and 753F by deletion of the sentence: "In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations."

ANNEX 5

Comparison of proposals relating to RR731E

- 1. First sentence
- 1.1 B/17/142 proposes the deletion of "and by the radiodetermination-satellite service (Earth-to-space)".
- 1.2 Change from "Resolution 46 (WARC-92) to "Resolution 46 (WRC-95) is editorial.
- 2. Second sentence
- 2.1 B/17/142 proposes the change from "either of the services" to "this service".
- 2.2 Change from "an e.i.r.p." to "a peak e.i.r.p." is unanimous.
- 2.3 USA/9/156 proposes the addition of "(No. 953 applies)".
- 3. Third sentence
- 3.1 Six contributions propose the change from "a value" to "a mean value".
- 3.2 RUS/7/28 proposes the change from "a value of -3 dB (W/4 kHz) is applicable" to "the mean e.i.r.p. density of a mobile earth station shall not exceed -3 dB (W/4 kHz)".

 $\mathcal{M}_{\mathcal{A}}(\mathcal{N}_{\mathcal{A}}(\mathcal{N}_{\mathcal{A}}(\mathcal{S}))) = 0$

oranis magazilaran bara bara a Beli este a bara di baran bara Baran baran

- 4. Fourth sentence
- 4.1 USA/9/156 and B/17/142 propose SUP.
- 4.2 EUR/5/39 proposes a modified text.
- 4.3 RUS/7/28, CAN/18/20, IND/41/149 and CHN/52/32 propose NOC.

 $\frac{1}{2} = \frac{1}{2} \left(\frac{1}{2} \left($

ANNEX 6

Comparison of proposals relating to RR746B

- 1. It seems that AUS/19/10 is covered by one of the other proposals.
- 2. First sentence
- 2.1 Some contributions propose amendments to "allocated frequency bands" and "date of entry". But they will not be considered by 5B1.

The report from 5B1 to WG 5B will contain the following Note:

[Note to WG 5B - Some proposals on RR746B contained amendments to "allocated frequency bands" and "date of entry". But Ad Hoc Group 5B1 did not consider them. The conclusion of WG 5B on these matters may lead to consequential amendments to RR746B.]

- 2.2 Change from "Resolution 46 (WARC-92)" to "Resolution 46 (WRC-95) is editorial.
- 3. Second sentence
- 3.1 J/22/95 proposes SUP
- 3.2 Change of "coordination of space stations"

EUR/5/40, CHN/62/37 "coordination of transmitting space stations"

B/17/171 "coordination of assignments to transmitting space stations"

- 3.3 EUR/5/40, B/17/171, IRN/25/14 and CHN/62/37 propose the addition of "under Resolution 46 (WRC-95)" after "the mobile-satellite service".
- 3.4 B/17/171 and CHN/62/37 propose the addition of "assignments to receiving stations of" before "terrestrial services".
- 3.5 Threshold values

EUR/5/40, IRN/25/16: identical with Rec. IS.1141 [Doc. 2/6] for non-GSO satellites and with

Rec. ITU-R IS.1142 [Doc. 2/7] for GSO satellites, with additional pfd

limits for "other terrestrial services"

USA/9/170, KOR/44/1E: identical with Rec. ITU-R IS.1141 [Doc. 2/6] for non-GSO satellites, but

no threshold for GSO satellites (sharing with mobile service not

discussed)

B/17/171: the threshold values for non-GSO satellites are proposed in B/17/161-164

(Article 28)

3.6 Approach for defining thresholds

Reference to Article 28: EUR/5/40, B/17/171, IRN/25/14 [Note 1 by the Chairman of 5B1 - The definition for calculation of FDP proposed in Article 28 seems ambiguous]

Reference to Rec. ITU-R IS.1141: USA/9/170, KOR/44/1E [Note 2 by the Chairman of 5B1 - Rec. ITU-R IS.1141 applies only to sharing between non-GSO satellites and FS]

Relevant provisions of RR: CHN/62/37

[Note 3 by the Chairman of 5B1 - In case of "incorporation by reference", ad hoc Group 5B1 has been requested by the Chairman of Committee 5 to prepare two alternative texts, one based on, and another not based on "incorporation by reference".]

[Note 4 by the Chairman of 5B1 - Note 2 of Rec. ITU-R IS.1141 [Doc. 2/6] is not appropriate from the regulatory standpoint, because a further reference to Rec. ITU-R F.1108 is not desirable. It seems that the following three Recommendations are required to calculate FDP:

Rec. ITU-R F.1108-1:

methodology for FDP calculation

Rec. ITU-R IS.1143 [Doc.2/8]: FS system parameters

Rec. ITU-R F.699-3:

FS antenna reference radiation pattern

In addition, it is noted that Rec. ITU-R F.1108-1 deals only with point-to-point radio-relay systems. It should be decided whether FDP in Rec. ITU-R IS.1141 should be calculated also for point-to-multipoint FS systems.]

[Note 5 by the Chairman of 5B1 - A case study for incorporating Rec. ITU-R IS.1141 by reference reveals the need of the following principle:

"An ITU-R Recommendation which may be incorporated into the Radio Regulations by reference shall not rely on other ITU-R Recommendations for its application, but shall contain all the information necessary for its application".

This may be communicated from Committee 5 to Committee 4.]

- 4. Third sentence
- 4.1 Change from "Resolution 46 (WARC-92)" to "Resolution 46 (WRC-95)" is editorial.
- 4.2 [Note 6 by the Chairman of 5B1 Revision by Committee 4 of Resolution 46 may lead to renumbering of the referenced provisions].
- 5. Fourth sentence
- 5.1 J/22/95 proposes a new fourth sentence.
- 5.2 CHN/62/37 proposes a new fourth sentence.

ANNEX 7

Comparison of proposals relating to RR753F

- 1. First sentence
- 1.1 B/17/159 proposes the deletion of "and the radiodetermination-satellite".
- 1.2 Change from "Resolution 46 (WARC-92)" to "Resolution 46 (WRC-95)" is editorial.
- 2. Second sentence
- 2.1 B/17/159 proposes the deletion of "and radiodetermination-satellite".
- 2.2 USA/9/174 proposes the addition of "per station".
- 2.3 Threshold values

USA/9/174:

single set of values in RR753F for both GSO and non-

GSO

satellites and for all types of terrestrial services

EUR/5/41, B/17/159, IRN/25/15:

reference to Article 28 (the values are identical with Rec.

ITU-R IS.1141 for non-GSO satellites and with Rec.

ITU-R IS.1142 for GSO satellites)

KOR/44/2B:

reference to Rec. ITU-R IS.1141 [Note - Rec. ITU-R

IS.1141 applies only to non-GSO satellites]

CAN/18/45-50:

no proposal to modify RR753F but a proposal to add

provisions in Article 28 (the values are identical with

USA/9/174)

3. Third sentence

See § 4 of Annex 6.

4. Fourth sentence

USA/9/174 proposes a new fourth sentence.



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/7-E 27 October 1995 Original: English only

GENEVA.

23 OCTOBER

17 NOVEMBER 1995

AD HOC GROUP 5B1

Chairman, ad hoc Group 5B1

PROPOSALS ON TECHNICAL CONSTRAINTS (AGENDA ITEM 2.1.a))

Please find attached herewith the texts of the proposals related to the work of ad hoc Group 5B1.

Annex 1 RR 731E

EUR/5/39, RUS/7/28, USA/9/156, B/17/142, CAN/18/20, IND/41/149, CHN/62/32 CPM Report (Chapter 2, Section I, Part A.2, § 1.3.1) (for information only).

Annex 2 RR 746B

EUR/5/40, USA/9/170, B/17/171, AUS/19/10, J/22/95, IRN/25/14, KOR/44/1E, CHN/62/37 CPM Report (Chapter 2, Section I, Part A.2, § 1.3.1) (for information only).

Annex 3 RR 753F

EUR/5/41, USA/9/174, B/17/159, IRN/25/15, KOR/44/2B CPM Report (Chapter 2, Section I, Part A.2, § 1.3.1) (for information only).

Annex 4 Article [28] S21

B/17/161-164, CAN/18/45, 46, 49, 50, IRN/25/16.

NOTE – The following proposals relating to "sharing studies" and "radio astronomy" are not included in this document. EUR/5/43, B/17/165, 166, ALG/23/4, IND/41/151.

M. MUROTANI Chairman of ad hoc Group 5B1

Annexes: 4

ANNEX 1

EUR/5/39 MOD 731E

S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite service (Earth-to-space) and by the radiodetermination-satellite service (Earthto-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). A mobile earth station operating in either of the services in this band shall not produce an peak e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of No. [732] S5.366, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a mean value of -3 dB(W/4 kHz) is applicable. Stations of the mobile-satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, and stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. S5.359.

Reasons: In order to ease the regulatory constraints on the mobile-satellite service in the band 1 610 - 1 626.5 MHz.

RUS/7/28

MOD 731E

S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite WARC-92 service (Earth-to-space) and by the radiodetermination-satellite service (Earthto-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92). A mobile earth station operating in either of the services in this band shall not produce an peak e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of No. [732] S5.366, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a value of the mean e.i.r.p. density of a mobile earth station shall not exceed -3 dB(W/4 kHz) is applicable. Stations of the mobile-satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. [730] S5.359.

> Reasons: In line with the recommendations in the CPM Report for more precise formulation of the requirements for shared use of frequencies with systems operating in accordance with No. [732] S5.366 of the Radio Regulations.

USA/9/156 MOD 731E

S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite service (Earth-to-space) and by the radiodetermination-satellite service (Earthto-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). A mobile earth

- 3 -CMR95/DL/7-E

station operating in either of the services in this band shall not produce an <u>peak</u> e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of No. [732] S5.366 (No. 953 applies), unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a <u>mean</u> value of -3 dB(W/4 kHz) is applicable. Stations of the mobile satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. [730] S5.359.

Reasons: Inclusion of the terms "peak" and "mean" are to clarify how the e.i.r.p. density limit should be measured. The text proposed for deletion at the end of this provision is unnecessary to protect the primary allocation status of the identified services and creates confusion and ambiguity concerning the primary status of the mobile-satellite service in the 1 610 - 1 626.5 MHz band. The addition of text to reference No. 953 is to ensure that special measures to avoid harmful interference to safety communications are taken into account.

B/17/142 MOD 731E S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite service (Earth-to-space) and by the radiodetermination satellite service (Earth-to-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). A mobile earth station operating in either of thethis services in this band shall not produce ana peak e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of No. [732] S5.366, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a mean value of -3 dB(W/4 kHz) is applicable. Stations of the mobile satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. [730] S5.359.

Reasons: A mean value of -3 dB (W/4 kHz) is required for the provision of the mobile-satellite service to lightweight handsets. It is also proposed that the mobile-satellite service operates on an equal basis with the aeronautical radionavigation service, the fixed service and those stations operating in accordance with the provisions of No. [732] S5.366. The reference to the radiodetermination-satellite service is also deleted for the reason in proposal B/17/141.

CAN/18/20 MOD 731E S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite service (Earth-to-space) and by the radiodetermination-satellite service (Earth-to-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92). A mobile earth station operating in either of the services in this band shall not produce an <u>peak</u> e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by

- 4 -CMR95/DL/7-E

systems operating in accordance with the provisions of No. [732] S5.366, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a mean value of -3 dB(W/4 kHz) is applicable. Stations of the mobile-satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. [730] S5.359.

Reasons: This modification is consistent with the recommendations of the CPM-95 Report (Chapter 2, Section I, Part A.2, item 1.3.1), and is intended to clarify the method of measurement of the e.i.r.p. density applied to earth stations of the mobile-satellite and radionavigation services.

IND/41/149 MOD 731E S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite service (Earth-to-space) and by the radiodetermination-satellite service (Earth-to-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). A mobile earth station operating in either of the services in this band shall not produce ana peak e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of No. [732] S5.366, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a mean value of -3 dB(W/4 kHz) is applicable. Stations of the mobile-satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. [730] S5.359.

Reasons: To clarify the reference to peak and mean values of the e.i.r.p..

CHN/62/32 MOD 731E S5.364

The use of the band 1 610 - 1 626.5 MHz by the mobile-satellite service (Earth-to-space) and by the radiodetermination-satellite service (Earth-to-space) is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). A mobile earth station operating in either of the services in this band shall not produce an peak e.i.r.p. density in excess of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of No. [732] S5.366, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a mean value of -3 dB(W/4 kHz) is applicable. Stations of the mobile-satellite service shall not cause harmful interference to, or claim protection from, stations in the aeronautical radionavigation service, stations operating in accordance with the provisions of No. [732] S5.366 and stations in the fixed service operating in accordance with the provisions of No. [730] S5.359.

Reasons: To clarify how the e.i.r.p. density limit should be measured.

CPM Report (Chapter 2, Section I, Part A.2)

1.3.1 Constraints on MSS Earth-to-space allocations

RR 731E

There is a need to clarify how the e.i.r.p. density limits given in RR 731E should be measured. In assessing whether the limit of RR 731E is exceeded, it is recommended that these limits should be understood to be a peak e.i.r.p. density of -15 dB(W/4 kHz) in the part of the band used by systems operating in accordance with the provisions of RR 732, unless otherwise agreed by the affected administrations. In the part of the band where such systems are not operating, a mean value of -3 dB(W/4 kHz) is applicable.

In some MSS allocations subject to Resolution 46, the MSS service is subject to various regulatory constraints via footnotes which may qualify the status of the MSS with respect to other services. For example, RR 731E states that the MSS, a primary service in the Table of Frequency Allocations (Article 8 of the RR), "shall not cause harmful interference to or claim protection from," some of the other primary services. This situation is not unique, however, since there are similar cases in which one primary allocation should protect another primary allocation if required by a footnote to the entry in the Table. Further consideration of the impact of such footnotes qualifying some primary MSS allocations within the context of Resolution 46 coordination between primary services will be necessary.

ANNEX 2

EUR/5/40 MOD 746B* \$5,389

The use of the bands 1 970 - 2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite service shall not commence before 1 January 20052000 and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). In the band 2 160 - 2 200 MHz coordination of transmitting space stations of the mobile-satellite service under Resolution 46 (WRC-95) with respect to terrestrial services is required only-if the power flux-density produced at the Earth's surface exceeds the limits in [No. 2566] Article S21, Table [AR28] if the limits in No. 2556.1 are exceeded. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: The reference to No. **2566** is updated taking into account the latest work of TG 2/2 to develop new coordination triggers in MSS space-to-Earth allocations with respect to the fixed service. See ITU-R draft new Recommendations 2/6 and 2/7. A new Article **28** No. **2556.1** is proposed taking into account TG-2/2's proposed coordination thresholds.

USA/9/170 MOD 746B

S5.389

The use of the bands 1 970 - 2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite service shall not commence before 1 January 2005 and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC 92 WRC-95). In the band 2 160 - 2 200 MHz coordination of space stations of the mobile-satellite service with respect to terrestrial services is required only if the power flux-density or Fractional Degradation Percentage produced at the Earth's surface exceeds the limits in No. 2566 threshold in Recommendation ITU-R IS.[Document 2/6]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92 WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: The modification to No. 746B will facilitate the introduction of mobile-satellite systems in this band by providing updated technical coordination limits.

B/17/171 MOD 746B S5.389

The use of the bands 1 970 1 990 - 2 010 MHz and 2 160 2 180 - 2 200 MHz by the mobile-satellite service shall not commence before 1 January 2005 2000 (and the bands 1 970 - 1 990 MHz and 2 160 - 2 180 MHz not before 1 January 2005) and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). In the band 2 160 - 2 200 MHz coordination of assignments to transmitting space stations of the mobile-satellite service under

^{*} See also EUR/5/53.

- 7 -CMR95/DL/7-E

Resolution 46 (WRC-95) with respect to assignments to receiving stations of terrestrial services is required only if the power flux-density produced at the Earth's surface exceeds the limits in Article S21, Table [AR28]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: To make 2 x 20 MHz available to the mobile-satellite service in the 2 GHz frequency range by the year 2000.

AUS/19/10

The coordination threshold reference in No. **746B** [S5.389] to "the limits in No. **2566**" [S21.16] should be replaced by an appropriate reference based on ITU-R Recommendation IS.[Document 2/6].

J/22/95 MOD 746B \$5,389

The use of the bands 1 970 - 2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite service shall not commence before 1 January 20052000 and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). In the band 2 160 - 2 200 MHz coordination of space stations of the mobile satellite service with respect to terrestrial services is required only if the power flux density produced at the Earth's surface exceeds the limits in No. [2566] Article S21, Table [AR28]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations. The use of the bands 1 970 - 2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite service does not preclude the use of these bands by other existing systems to which the frequencies within these bands are assigned before 17 November 1995.

Reasons: The CPM Report says that the frequency sharing between MSS (Earth-to-space) and FS is difficult. Since many fixed stations are used in these bands in the world, priority should be given to the existing services. Therefore, it is adequate to introduce MSS systems by using frequency segments which are lightly used by the FS, in order to avoid severe FS interference.

IRN/25/14 MOD 746B S5.389

The use of the bands 1 970 - 2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite service shall not commence before 1 January 2005 and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). In the band 2 160 - 2 200 MHz coordination of space stations of the mobile-satellite service under Resolution 46 (WRC-95) with respect to terrestrial services is required only if the power flux density produced at the Earth's surface exceeds the limits in [No. 2566] Article S21, Table [AR28] are exceeded. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of

Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: The reference to [No. 2566] Article S21, Table [AR28] is updated taking into account the latest work of TG 2/2 to develop new coordination triggers in MSS space-to-Earth allocations with respect to the Fixed Service. See ITU-R draft new Recommendations 2/6 and 2/7. A new Article 28, No. 2556.1 is proposed, taking into account TG 2/2's proposed coordination thresholds.

KOR/44/1E MOD 746B S5.389

The use of the bands 1 970 - 2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite service shall not commence before 1 January 20052000 and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). In the band 2 160 - 2 200 MHz coordination of space stations of the mobile-satellite service with respect to terrestrial services is required only if the power flux-density or fractional degradation in performance produced at the Earth's surface per space station exceeds the limits in Article [2566] S21, Table [AR28]ITU-R IS Recommendation [Document 2/6]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC 92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: This proposal is to facilitate the timely introduction of worldwide mobile-satellite services and to reduce the number of coordinations between the mobile-satellite service and fixed service, while providing adequate protection to the fixed service.

CHN/62/37

MOD 746B The use of the bands 1-970 - 2 010 MHz and 2-160 - 2 200 MHz

S5.389

WARC-92 <u>1 980 - 2 010 MHz and 2 170 - 2 200 MHz in Regions 1 and 3 and</u> 1 970 - 2 010 MHz and 2 160 - 2 200 MHz in Region 2 by the mobile-satellite service shall not commence before 1 January 20052000 and is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). In the band 2160 - 2200 MHz coordination of transmitting space stations of the mobile-satellite service under Resolution 46 (WRC-95) with respect to assignments to receiving stations of terrestrial services is required only if the power flux-density produced at the Earth's surface exceeds the limits in No. 2566. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations, the relevant provisions of the Radio Regulations. Before 1 January 2005, the use of the bands 1 970 -2 010 MHz and 2 160 - 2 200 MHz by the mobile-satellite services shall not cause harmful interference to the use of these bands by, and shall not claim protection from harmful interference caused by other services to which these bands are allocated.

- 9 -CMR95/DL/7-E

Reasons:

- 1 To advance the date of entry into force of allocations in the bands 1 980 2 010 MHz and 2 170 2 200 MHz to 1 January 2000 to satisfy the need of the MSS.
- 2 Resolution 46 (WARC-92) would be modified at WRC-95.
- 3 To emphasize the need to protect already existing FS systems in the 2 GHz band.

CPM Report (Chapter 2, Section I, Part A.2)

1.3.1 Constraints on MSS Earth-to-space allocations

RR 746B and RR 746C

The bands 1 970 - 1 980 MHz in Region 2, and 1 980 - 2 010 MHz in all Regions, are allocated to the MSS under RR 746B and RR 746C (see Part B of Section I).

- 10 -CMR95/DL/7-E

ANNEX 3

EUR/5/41 MOD 753F

S5.402

The use of the band 2 483.5 - 2 500 MHz by the mobile-satellite and the radiodetermination-satellite services is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). Coordination of space stations of the mobile-satellite and radiodetermination-satellite services with respect to terrestrial services is required only if the power flux-density produced at the Earth's surface exceeds the limits in [No. 2556] Article S21, Table [AR28] No. 2556.1. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: See EUR/5/40.

USA/9/174 MOD 753F S5.402

The use of the band 2 483.5 - 2 500 MHz by the mobile-satellite and the radiodetermination-satellite services is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). Coordination of space stations of the mobile-satellite and radiodetermination-satellite services with respect to terrestrial services is required only if the power flux-density produced at the Earth's surface per space station exceeds the limits in No. 2566.

- = $\frac{-150 \text{ dB (W/m}^2) \text{ in any 4 kHz band for angles of arrival between 0 and 5}^{\Omega}}{\text{above the horizontal plane;}}$
- $= \frac{-150 + 0.65 (\underline{\delta} 5) dB (W/m^{2}) in any 4 kHz band for angles of arrival \underline{\delta}}{(\underline{\circ}) \text{ between 5 and 25}^{\underline{\circ}} \text{ above the horizontal plane;}}$
- = $\frac{-137 \text{ dB (W/m}^2) \text{ in any 4 kHz band for angles of arrival between 25 and}}{90^{\circ} \text{ above the horizontal plane.}}$

These values relate to the power-flux density which would be obtained under assumed free-space conditions. See Recommendation ITU-R IS.[Document 2/6]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations. Administrations are urged to take all practicable steps to prevent harmful interference to the radio astronomy service from emissions in the 2 483.5 - 2 500 MHz band, especially those caused by second-harmonic radiation that would fall into the 4 990 - 5 000 MHz band allocated exclusively to the radio astronomy service worldwide.

Reasons: As stated in the CPM-95 Report, this modification will reduce the number of coordinations required between the mobile-satellite service and the fixed service, while providing adequate protection to the fixed service systems in this band.

- 11 -CMR95/DL/7-E

B/17/159 MOD 753F

S5.402

The use of the band 2 483.5 - 2 500 MHz by the mobile-satellite and the radiodetermination satellite services is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). Coordination of space stations of the mobile-satellite and radiodetermination-satellite services with respect to terrestrial services is required only if the power flux-density produced at the Earth's surface exceeds the limits in Article S21, Table [AR28]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: Same as in proposal B/17/158.

IRN/25/15 MOD 753F S5.402

The use of the band 2 483.5 - 2 500 MHz by the mobile-satellite and the radiodetermination-satellite services is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). Coordination of space stations of the mobile-satellite and radiodetermination-satellite services with respect to terrestrial services is required only if the power flux-density produced at the Earth's surface exceeds the limits in [No. 2566] Article S21, Table [AR28]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: See IRN/25/16.

KOR/44/2B MOD 753F S5.402

The use of the band 2 483.5 - 2 500 MHz by the mobile-satellite and the radiodetermination-satellite services is subject to the application of the coordination and notification procedures set forth in Resolution 46 (WARC-92WRC-95). Coordination of space stations of the mobile-satellite and radiodetermination-satellite services with respect to terrestrial services is required only if the power flux-density produced at the Earth's surface per space station exceeds the limits in Article [2566] S21, Table [AR28]ITU-R IS Recommendation [Document 2/6]. In respect of assignments operating in this band, the provisions of Section II, paragraph 2.2 of Resolution 46 (WARC-92WRC-95) shall also be applied to geostationary transmitting space stations with respect to terrestrial stations.

Reasons: This modification is to reduce the number of coordinations between the mobile-satellite service and the fixed service, while providing adequate protection to the fixed service.

CPM Report (Chapter 2, Section I, Part A.2)

1.3.1 Constraints on MSS Earth-to-space allocations

RR 753F

The pfd thresholds currently applicable to MSS (space-to-Earth) in the 2 483.5 - 2 500 MHz band are those contained in RR 2566. These thresholds may be more stringent than is required to protect analogue terrestrial fixed services. Some MSS systems may plan to operate at slightly higher pfds in order to provide the levels of capacity needed to meet service requirements. Raising the pfd thresholds may reduce the number of coordinations between MSS and FS systems, while still providing adequate protection to the analogue FS systems. With regard to protection of digital systems, refer to § 1.4.6.4a. For proposed new pfd levels in this band, refer to draft new Recommendation ITU-R IS.[Document 2/6].

ANNEX 4

B/17/161 ADD

TABLE [AR28bis]

Frequency band	requency band Service		Limit in dB(W/m ² /4 kHz) for angle of arrival δ above the horizontal plane			
		0° - 5°	5° - 25°	25° - 90°		
1 492 - 1 525	Mobile-satellite	-1526, 7	-152 + 0.5 $(\delta-5)^6$, 7	-1426, 7		
2 160 - 2 170 (R1, R2)	Mobile-satellite	-1477	$-147 + 0.5$ $(8-5)^7$	-137 ⁷		
2 170 - 2 200	Mobile-satellite	-147 ⁷	-147 + 0.5 $(\delta-5)^7$	-1377		
2 483.5 - 2 500	Mobile-satellite	-1508	-150 + 0.65 $(\delta-5)^8$	-1378		

Reasons: To take into account the results of the work of the ITU-R in the cases of the bands 1 492 - 1 525 MHz, 2 160 - 2 200 MHz and 2 483.5 - 2 500 MHz.

B/17/162 ADD S21.16.6

⁶ The power-flux density values to protect analogue systems is to be determined for this band for each non-geostationary mobile-satellite system proposed. Until such time as the new values are available, the values given in this box will apply.

Reasons: Same as in proposal B/17/161.

B/17/163 ADD S21.16.7

The power flux-density values specified for the bands 1 492 - 1 530 MHz and 2 160 - 2 200 MHz provide full protection for analogue radio-relay systems from a non-geostationary mobile-satellite system employing narrow-band TDMA/FDMA techniques. For digital radio-relay systems an FDP (fractional degradation of performance) value of 25% is applicable. The method for calculating the FDP for a fixed service network is contained in the ITU-R Recommendations.

Reasons: Same as in proposal B/17/161.

B/17/164 ADD S21.16.8

⁸ The power flux-density values given in this box provide full protection for analogue radio-relay systems from multiple non-geostationary mobile-satellite systems employing CDMA techniques. These power flux-density values will not provide full protection for existing digital fixed systems in all cases. However, these power flux-density values are considered to provide adequate protection for digital fixed systems designed to operate in this band, where high power ISM and possible low power applications are expected to produce a relatively high interference environment.

Reasons: Same as in proposal B/17/161.

- 14 -CMR95/DL/7-E

CAN/18/45 ADD 2560A

(2bis)Coordination threshold:

- a) Coordination is required with the fixed service if the power-flux density at the Earth's surface produced by emissions from a space station of a non-geostationary system, including emissions from a reflecting satellite, for all conditions and for all methods of modulation, exceeds the following values:
 - -150 dB(W/m²) in any 4 kHz band for angles of arrival between 0 and 5 degrees above the horizontal plane;
 - -150 + 0.65 (δ -5) dB(W/m²) in any 4 kHz band for angles of arrival δ (in degrees) between 5 and 25 degrees above the horizontal plane;
 - -137 dB(W/m²) in any 4 kHz band for angles of arrival between 25 and 90 degrees above the horizontal plane.

These values relate to the power-flux density which would be obtained under assumed free-space propagation conditions.

CAN/18/46 ADD 2560B

b) The coordination thresholds given in No. **2560A** apply in the frequency band 2 483.5 - 2 500 MHz to the mobile-satellite service (space-to-Earth) and the radiodetermination-satellite service (space-to-Earth).

Reasons: To implement the findings of CPM-95 (see Section 3.6.4.8, Chapter 2, Section 1, Part C of the CPM-95 Report).

NOTE – Canada is considering the possibility of creating a single provision in **Article 28**, **Article S21** and **Appendix S5** which would encompass such coordination thresholds as provisions distinct from the regular pfd limits of the current **Article 28**.

er tiga. Innersia

the and the second of

THE REPORTS

and experiences of the second of the second

The area of the state of the s

LINTAF GOL

Consequential modifications to Article S21 and Table [AR28]

TABLE [AR28]

CAN/18/49 MOD

Frequency band	Service	Limit in dB(V	Reference		
		0° - 5°	5° - 25°	25° - 90°	bandwidth
2 483.5 - 2 500	Mobile-Satellite (S-E)	<u>-150</u> <u>6</u>	<u>-150 + 0.65 (δ-5)</u> 6	<u>-137</u> 6	4 kHz
	Radiodetermination -Satellite (S-E)				

CAN/18/50

ADD S.21.16.6

⁶ These values are to be used to determine whether coordination is required between the non-geostationary space stations and stations in the fixed or mobile services.

Reasons: Implement the findings of CPM-95 (see CPM-95 Report, Section 3.6.4.8, Chapter 2, Section 1, Part C).

IRN/25/16 **ADD** 2556.1

Frequency band (MHz)	Service to be protected	Coordination threshold factors				
		Geostationary space stations		Non-geostationary space stations		
		pfd calculation factors (NOTE 2)		pfd calculation factors (NOTE 2)		FDP (in 1 MHz)
		P dB (W/m²) in 4 kHz	r dB/deg	P dB (W/m ²) in 4 kHz	r dB/deg	(NOTE 1)
2 160 - 2 200	Fixed service	-152	0.5	-146	0.5	
2 483.5 - 2 500	analogue	-152	0.5	-150	0.65	
2 160 - 2 200	Fixed service	-152	0.5			25 %
2 483.5 - 2 500	digital	-152	0.5	-150	0.65	
2 160 - 2 200	Other terrestrial	-152	0.5	-152	0.5	1 1
2 483.5 - 2 500	services	-152	0.5	-152	0.5	

NOTE 1 – The definition for calculation of FDP (fractional degradation in performance) is contained in the relevant ITU-R Recommendations.

NOTE 2 – The following formulae are used for pfd:

 $P dB(W/m^2/4 kHz)$

for $0^{\circ} \leq \delta \leq 5^{\circ}$

 $P + r(\delta-5) dB(W/m^2/4 kHz)$ for $5^{\circ} < \delta < 25^{\circ}$

 $P + 20r dB(W/m^2/4 kHz)$

for $25^{\circ} \le \delta \le 90^{\circ}$

The limits relate to the power-flux density which would be obtained under assumed free-space propagation conditions.

Reasons: See IRN/25/15.

UNIÓN INTERNACIONAL DE TELECOMUNICACIONES



CMR-95

CONFERENCIA MUNDIAL DE RADIOCOMUNICACIONES

Documento DL/8-S 2 de noviembre de 1995 Original: inglés

GINEBRA, 23 DE OCTUBRE – 17 DE NOVIEMBRE DE 1995

GRUPO DE TRABAJO 4B

Nota del Presidente del Grupo de Trabajo 4B

ARTÍCULO S9

F/54/25A MOD

Sección II. Procedimiento de coordinación 2,3

[A.S9.II.1

Estos procedimientos <u>pueden ser aplicados también son</u> <u>aplicables</u> a estaciones terrenas del servicio de exploración de la Tierra por satélite, del servicio de investigación espacial, del servicio de operaciones espaciales y del servicio de radiodeterminación por satélite destinadas a su utilización en desplazamientos o durante paradas en puntos no determinados.]

F/54/26

ADD A.S9.II.2

En todo este artículo, "coordinación" se refiere también al proceso de búsqueda del acuerdo de otras administraciones cuando así se requiera de conformidad con el número **S9.21**.

Subsección IIA. Necesidad y solicitud de coordinación

F/54/27A MOD S9.6

Antes de notificar a la Oficina o poner en servicio una asignación de frecuencia en cualquiera de los casos seguidamente enumerados, la administración interesada^{1,2} deberá efectuar una coordinación, según se requiera, con las administraciones identificadas de acuerdo con el número \$9.27:

NOC S9.6.1

En el caso de coordinación de una asignación en una red de satélites en relación con otra red de satélites, la administración interesada puede actuar en nombre de un grupo de administraciones nominadas. Cuando en el marco de esta disposición una administración actúe en nombre de un grupo de administraciones nominadas, todos los miembros de ese grupo tendrán derecho a responder con respecto a sus propias redes o sistemas.

F/54/27B SUP S9.6.2

F/54/28

MOD S9.7

a) para una estación de una red de satélite que utiliza la órbita de los satélites geoestacionarios, con respecto a cualquier otra red de satélite que utilice dicha órbita, para todos los de cualesquiera servicios de radiocomunicación espacial y todas las bandas de frecuencias excepto las abarcadas por un Plan mundiallos Planes de los apéndices S30, S30A y S30B;

F/54/29

[MOD S9.8

b)* para las estaciones espaciales transmisoras del servicio fijo por satélite que utilicen la órbita de los satélites geoestacionarios en una banda de frecuencias compartida a título primario con igualdad de derechos con el servicio de radiodifusión por satélite, con respecto a las estaciones espaciales de este último servicio que estén sujetas a unal Plan del apéndice \$30;]

F/54/30

[MOD S9.9

c)* para una estación <u>espacial transmisora</u> del servicio fijo por satélite <u>que</u> <u>utilice la órbita de los satélites geoestacionarios</u> en una banda de frecuencias compartida a título primario con igualdad de derechos con enlaces de conexión del servicio de radiodifusión por satélite que estén sujetos <u>a unal</u> Plan <u>del apéndice S30A;</u>]

F/54/31

SUP S9.10

F/54/32

(MOD) S9.11

ed) para una estación espacial del servicio de radiodifusión por satélite, en una banda de frecuencias compartida a título primario con igualdad de derechos con servicios terrenales para las que no exista un plan relativo a aquel servicio, con respecto a los servicios terrenales;

F/54/33

ADD S9.11bis

e) para una estación con respecto a la cual se estipula el requisito de efectuar coordinación en una nota del Cuadro de atribuciones de frecuencias que haga referencia a esta disposición;

F/54/34

MOD S9.12

f)i) para una estación de en una red de satélite que utiliza una órbita no geoestacionaria en las bandas de frecuencias a las que se aplica la Resolución 46, con respecto a cualquier otra red de satélite que emplee una órbita no geoestacionaria y, con respecto a cualquier otra red de satélite que utilice la órbita de los satélites geoestacionarios, excepto la coordinación con arreglo a lo dispuesto en el número \$9.17bis;

^{*} La aplicación de estas disposiciones se aplaza hasta la decisión de la CMR-97 sobre la revisión de los apéndices 30 y 30A.

F/54/35

MOD S9.13

g)ii) para una estación de en una red de satélites que emplea la órbita de los satélites geoestacionarios en las bandas de frecuencias a las que se aplica la Resolución 46, con respecto a cualquier otra red de satélite que utilice una órbita no geoestacionaria;

F/54/36A

MOD S9.14

h)iii) para una estación de en una red de satélite que utiliza una órbita no geoestacionaria¹, en las bandas de frecuencias a las que se aplica la Resolución 46-con respecto a las estaciones de los servicios terrenales donde se rebasan los límites valores umbral de densidad de flujo de potencia;-GVE Nota 3

F/54/36B

SUP

GVE Nota 3.

F/54/37

ADD S9.14.1

Las disposiciones del número S9.14 se aplicarán también a una estación espacial transmisora de una red de satélites que utilice la órbita de los satélites geoestacionarios, con respecto a las estaciones terrenales de los números S5.348, S5.354, S5.389, S5.402 y S5.414.

F/54/38

(ADD) S9.15

iv) si bien una estación terrena específica o bien una estación terrena típica con respecto a las estaciones terrenales en bandas de frecuencias atribuidas con igualdad de derechos a servicios espaciales y terrenales y cuando la zona de coordinación de la estación terrena incluye el territorio de cualquier otro país;

F/54/39

(**ADD**) S9.16

 v) si una estación transmisora o receptora de un servicio terrenal dentro de la zona de coordinación de una estación terrena en una red de satélites no geoestacionarios;

F/54/40A

MOD S9.4517

para cualquier estación terrena individualespecífica o estación terrena móvil típica, en bandas de frecuencias por encima de 1 GHz atribuidas con igualdad de derechos a servicios espaciales y terrenales con respecto a las estaciones terrenales, y cuando la zona de coordinación de la estación terrena incluye el territorio de cualquier otro país-GVE Nota 4, excepto la coordinación con arreglo a lo dispuesto en el número S9.15;

F/54/40B

SUP

GVE Nota 4.

F/54/38

(SUP) S9.16

F/54/41

ADD S9.17bis

g) para cualquier estación terrena específica, con respecto a otras estaciones terrenas que operan en el sentido opuesto de la transmisión en bandas de frecuencias atribuidas con igualdad de derechos a servicios de radiocomunicación espaciales en ambos sentidos de la transmisión y cuando la zona de coordinación de la estación terrena incluye el territorio de cualquier otro país;

F/54/42

MOD S9.1718

k)h) para cualquier estación transmisora de un servicio terrenal en las bandas mencionadas en el número S9.1517 dentro de la zona de coordinación de una estación terrena excepto la coordinación con arreglo a lo dispuesto en el número S9.16;

F/54/39

(SUP) S9.18

F/54/43

[(MOD)S9.19*

m)i) para cualquier estación transmisora de un servicio terrenal en una banda de frecuencias compartida a título primario con igualdad de derechos con el servicio de radiodifusión por satélite;]

F/54/44

SUP S9.20

Será analizada por el Grupo 4C.

F/54/45A

MOD S9.21

e)j) para cualquier estación de un servicio con respecto al cual se estipula el requisito de efectuar la coordinación u obtener buscar el acuerdo de otras administraciones en una nota del Cuadro de atribución de bandas de frecuencias que haga referencia a esta disposición, o en cualquier otra disposición del presente Reglamento. GVE Nota 5

F/54/45B

SUP

GVE Nota 5.

F/54/46

SUP S9.22

F/54/47

MOD S9.23

Cuando sea necesario efectuar más de una forma de coordinación de acuerdo con el número S9.30, las correspondientes solicitudes se identificarán de una manera apropiada por referencia a los números S9.7 a S9.21 y, en la medida posible, se enviarán y se publicarán simultáneamente, cuando sea conveniente.

F/54/48

SUP S9.24

^{*} La aplicación de esta disposición se aplaza hasta la decisión de la CMR-97 sobre la revisión del apéndice 30.

B/17/32

SUP S9.25

F/54/50

MOD S9.26

La coordinación podrá efectuarse para una red de satélites que utilice la información relativa a la estación espacial incluyendo su zona de servicio y los parámetros de una o más estaciones terrenas típicas situadas en <u>la totalidad o parte de zona de servicio de la estación espacial. Lo mismo se aplicará La coordinación también puede ser efectuada para redes terrenales que utilicen la información relativa a las estaciones terrenales típicas, exceptuadas las mencionadas en los números S11.18 a S11.23.</u>

F/54/51

MOD S9.27

Las asignaciones de frecuencia que han de tenerse en cuenta al efectuar la coordinación se especifican para cada caso en el apéndice S5. La administración solicitante identificará, aplicando el método de cálculo y los eriterios definidos en el mencionado apéndice a dichas asignaciones de frecuencia, las administraciones con las cuales haya de efectuarse la coordinación.

F/54/52

MOD S9.28

En el caso de peticiones de coordinación de acuerdo con el número S9.29, la administración solicitante deberá, aplicando el método de cálculo y los criterios contenidos en el apéndice S5 a esas atribuciones de frecuencia, identificar, en la medida de lo posible, las administraciones con las que ha de efectuarse la coordinación.

USA/9/35 MOD S9.29

La administración solicitante enviará entonces a las administraciones identificadas, solicitudes de coordinación con arreglo a los números S9.15 - S9.19 con copia a la Oficina¹, una petición de coordinación junto con la información apropiada enumerada en el apéndice S4 a este Reglamento.

F/54/54A MOD S9.30

De forma alternativa, la administración solicitante puede enviar la información adecuada a la Oficina para publicación en la circular semanal² como una solicitud de coordinación. Las peticiones de coordinación efectuadas de acuerdo con los números S9.7 - S9.14 [S9.20] y S9.21 deberán ser enviadas por la administración solicitante a la Oficina junto con la información apropiada enumerada en el apéndice S4 a este Reglamento.

F/54/55 MOD S9.31

La información enviada en virtud del número S9.29 incluirá también, en los casos previstos por los números S9.15 e, S9.1617 o S9.17bis, diagramas en una escala adecuada que indiquen, para la transmisión y recepción, la ubicación de la estación terrena y su correspondiente zona de coordinación, o la zona de coordinación relacionada con la zona de servicio en la que vaya a funcionar la estación terrena móvil, y los datos en que se basan

- 6 -CMR95/DL/8-S

los diagramas. Con respecto a las estaciones terrenales, <u>en los casos previstos</u> <u>por los números S9.16 y S9.18</u> la información deberá incluir las ubicaciones de las estaciones de una red terrenal que se hallen total o parcialmente dentro de la zona de coordinación de la correspondiente estación terrena.

F/54/56

MOD S9.32

Si la administración responsable, tras la aplicación de los números [S9.2223 a S9.27], llega a la conclusión de que no es necesaria la coordinación, podrá enviar a la Oficina la información correspondiente del según el apéndice S4 para tomar las acciones correspondientes en virtud del número S9.34 o de la sección I del artículo S11.

F/54/57

ADD S9.32bis

Si la administración responsable, tras la aplicación de los números \$9.15 a \$9.19, llega a la conclusión de que no es necesaria la coordinación, podrá enviar a la Oficina la información correspondiente según el apéndice \$4 para tomar las acciones correspondientes en virtud de la sección I del artículo \$.11.

F/54/58

MOD S9.33

Si por cualquier motivo una administración no puede actuar de acuerdo con lo dispuesto en el número \$9.29 deberá recabar la asistencia de la Oficina. Esta última enviará entonces la petición a las administraciones interesadas y ofrecerá la ayuda precisa-tomando. La Oficina tomará también cualquier acción ulterior necesaria en virtud de los números \$9.45 y \$9.46.

F/54/59

MOD S9.34

Al recibir la información completa enviada con arreglo al número \$9.29\$\frac{S9.30}{29}\$ o \$9.32, la Oficina deberá proceder rápidamente a:

NOC S9.35

a) examinar la información con respecto a su conformidad con el número S11.31;

F/54/60

MOD S9.36

b) identificar de acuerdo con el número S9.27, cualquier administración cuyos servicios puedan resultar afectados con la que se tenga que efectuar una coordinación⁴;

F/54/61

ADD S9.36.1

Las administraciones identificadas por la Oficina en virtud de los números S9.12 a S9.14 y S9.21 sólo lo serán a efectos informativos, para ayudar a las administraciones a cumplir con este procedimiento.

F/54/62

MOD S9.37

c) incluir los nombres de éstas en la <u>publicación en virtud del número S9.38</u> información enviada en virtud del número S9.29;

SUP S9.38.1

IND/41/68

MOD S9.38

d) publicar de manera adecuada³ la información completa en la circular semanal en un plazo inferior a cuatro meses;

F/54/64

SUP S9.39

- 7 -CMR95/DL/8-S

NOC S9.40

f) informar a las administraciones interesadas de su actuación, comunicar los resultados de sus cálculos y señalar a su atención la correspondiente circular semanal

IND/41/69

ADD S9.40bis

Si las informaciones comunicadas se consideran incompletas, la Oficina pedirá inmediatamente a la administración interesada las aclaraciones necesarias así como la información que falte.

F/54/66 MOD S9.41

Si tras la recepción de la circular semanal, en la que se hace referencia a peticiones de coordinación con arreglo a lo dispuesto en los números \$9.7 a \$9.9, una administración considera que hubiese tenido que ser incluida en la solicitud, deberá informar de ello a la administración que solicita el acuerdo y a la Oficina en un plazo de cuatro meses a partir de la fecha de publicación*, indicando los motivos técnicos de su solicitud y solicitando que se incluya su nombre.

F/54/67 MOD S9.42

La Oficina estudiará esta información sobre la base del apéndice S5½ y las Reglas de Procedimiento y comunicará sus conclusiones a ambas administraciones. Si la Oficina estuviera de acuerdo en incluir la administración en la solicitud, publicará un addéndum a la publicación en virtud del número S9.38.

F/54/68

SUP S9.42.1

F/54/69

MOD S9.43

Cuando este Reglamento no permita identificar a todas las administraciones afectadas, Después de tomar las medidas indicadas en el número S9.41, las administraciones que no respondan dentro del plazo especificado en el número S9.41 se considerarán no afectadas y se aplicarán las disposiciones de los número S9.48 y S9.49.

USA/9/50 MOD S9.44

La administración que solicita el acuerdo coordinación y las administraciones interesadas, o la Oficina cuando actúe con arreglo al artículo \$7, número \$7.6, podrán pedir cuantas informaciones adicionales consideren necesarias. Se enviará a la Oficina copia de todas las solicitudes de esta naturaleza y de las respuestas a las mismas.

^{*} De la correspondiente circular semanal.

- 8 -CMR95/DL/8-S

Subsección IIB. Acuse de recibo de una solicitud de coordinación

F/54/7	1				
MOD	S9.45	Cuando una administración reciba una solicitud de coordinación con arreglo a lo dispuesto en el número S9.29 deberá, en un plazo de 30 días desde la fecha de la circular semanal correspondiente solicitud, acusar recibo de ella por telegrama a la administración solicitante, con copia a la Oficina. Si no se acusa recibo de su solicitud en el plazo de 30 días, la administración solicitante enviará un telegrama solicitando acuse de recibo, con copia a la Oficina.			
NOC	S9.46	Si en un plazo de 15 días después del envío de esta segunda solicitud con arreglo al número S9.45 no se acusa recibo de la misma, la administración solicitante recabará la asistencia de la Oficina. En este caso, la Oficina enviará enseguida un telegrama a la administración que no ha respondido para pedirle un acuse de recibo inmediato.			
NOC	S9.47	Si después de tomar las medidas previstas en el número S9.46 la Oficina no recibe un acuse de recibo en un plazo de 30 días, se considerará que la administración que no ha acusado recibo se compromete:			
NOC	S9.48	 a) a no formular ninguna reclamación con respecto a ninguna interferencia perjudicial que pudiera causar a sus propias asignaciones la asignación para la cual se ha solicitado la coordinación; y 			
NOC	S9.49	 a utilizar sus propias asignaciones de manera tal que no causen interferencia perjudicial a la asignación para la cual se ha solicitado la coordinación. 			
		Subsección IIC Respuesta a una solicitud de coordinación**			

Subsección IIC. Respuesta a una solicitud de coordinación**

F/54/72A MOD S9.50

Cuando una administración reciba una solicitud de coordinación o haya sido incluida en el procedimiento tras las medidas descritas en el número S9.41, deberá examinar a la mayor brevedad posible el asunto con respecto a la interferencia que podrían sufrir o, en ciertos casos, causar sus propias asignaciones. Para las asignaciones que han de tenerse en cuenta, véase de acuerdo con el apéndice S5³.

SUP S9.50.1

NOC S9.50.2

A falta de disposiciones concretas en el presente Reglamento con respecto a la evaluación de la interferencia, deben aplicarse métodos de cálculo y criterios basados en las correspondientes Recomendaciones UIT-R aceptadas por las administraciones interesadas. En caso de desacuerdo con respecto a una Recomendación, o de inexistencia de tal Recomendación, los métodos y criterios deberán ser acordados entre las administraciones interesadas. Estos acuerdos no deberán perjudicar a otras administraciones.

- 9 -CMR95/DL/8-S

NOC S9.50.3

Cuando el apéndice S5 especifica un periodo durante el cual se pueden tener en cuenta las asignaciones planificadas, se puede ampliar dicho periodo por acuerdo entre las administraciones interesadas.

F/54/73A MOD S9.51

Después de tomar las medidas indicadas en el número \$9.50, la administración a la que se solicite la coordinación con arreglo a lo dispuesto en los números \$9.7 a \$9.9 deberá comunicar su acuerdo a la administración solicitante y a la Oficina en el plazo de cuatro meses a partir de la fecha de la correspondiente circular semanal GVE Nota 6, o actuar de acuerdo con el número \$9.52. En el caso de una solicitud de coordinación que no sea necesario publicar en la circular semanal (es decir, con arreglo a los números \$9.15 a \$9.19) el periodo de cuatro meses empezará a contar a partir de la fecha de la solicitud.

SUP

GVE Nota 6.

F/54/74

ADD S9.51bis

Después de tomar las medidas indicadas en el número S9.50, la administración a la que se solicita la coordinación con arreglo a lo dispuesto en los números S9.15 a S9.19 deberá comunicar su acuerdo a la administración solicitante en el plazo de cuatro meses a partir de la fecha de despacho de los datos de coordinación, o actuar de acuerdo con el número S9.52.

F/54/75 MOD S9.52

Si una administración, tras tomar las medidas indicadas en el número S9.50, no está de acuerdo con la solicitud de coordinación, comunicará su desacuerdo a la administración solicitante dentro del mismo de un plazo de cuatro meses y le facilitará información sobre sus propias asignaciones que motivan su desacuerdo. Formulará asimismo cuantas sugerencias pueda ofrecer para resolver satisfactoriamente el asunto. Se enviará a la Oficina copia de esta información. Cuando esta información se refiera a estaciones terrenales o estaciones terrenas que operan en el sentido opuesto de la transmisión situadas dentro de la zona de coordinación de una estación terrena, solo la información relativa a las estaciones de radiocomunicaciones existentes o a las que se han de poner en servicio en los tres meses siguientes, en el caso de las estaciones terrenales, se tratará como las notificaciones efectuadas con arreglo a lo dispuesto en los números S11.2 o S11.9.

F/54/76 ADD S9.52bis

Cuando se alcance un acuerdo sobre coordinación, la administración responsable de las estaciones terrenales o de la estación terrena que operan en el sentido opuesto de la transmisión, pueden enviar a la Oficina la información relativa a las estaciones abarcadas por el acuerdo, que se ha de

- 10 -CMR95/DL/8-S

notificar con arreglo a lo dispuesto en los números S11.2 o S11.9. La Oficina sólo considerará como notificaciones la información relativa a las estaciones terrenales o terrenas que operen en el sentido opuesto de la transmisión o que se vayan a poner en servicio en los tres años siguientes.

F/54/77

ADD S9.52ter

En el caso de una solicitud de coordinación con arreglo a lo dispuestos en los números [12] S9.11 a S9.14 y S9.21, una administración que no responda de conformidad con el número S9.52 dentro del plazo especificado será considerada como no afectada y se aplicarán las disposiciones de los números S9.48 y S9.49.]

NOC S9.53

Seguidamente, la administración que solicita el acuerdo y la que ha respondido harán todo lo posible, para solucionar las dificultades de forma aceptable por ambas partes afectadas.

NOC S9.54

La administración que busca la coordinación o una administración cuyas asignaciones puedan resultar afectadas podrán solicitar cuantas informaciones adicionales necesiten para evaluar la interferencia causada a sus propias asignaciones o para resolver el asunto.

F/54/78

MOD S9.55

Todas las administraciones podrán comunicarse por correspondencia, por cualquier medio de telecomunicación adecuado o celebrar, de ser necesario, reuniones para resolver el asunto, y los resultados deberán comunicarse a la Oficina y ser publicados por la misma en la circular semanal, según proceda.

SUP S9.56 SUP S9.56.1 SUP S9.57

F/54/80

MOD S9.58

La administración que haya iniciado la coordinación, así como aquellas con las que se trate de efectuar la coordinación, comunicarán a la Junta Oficina toda modificación de las características publicadas de sus redes respectivas que se haya realizado para llegar a un acuerdo sobre la coordinación. La Junta Oficina publicará esta información de conformidad con el número S9.38 indicando que esas modificaciones son el resultado del esfuerzo común de las administraciones interesadas para llegar a un acuerdo sobre la coordinación, y que por este motivo deben ser objeto de especial consideración. Las modificaciones pueden entrañar la aplicación de la subsección IIA del artículo S9 con respecto a otras administraciones.

NOC S9.59

Si la administración que solicita la coordinación y una administración interesada no pueden ponerse de acuerdo sobre el nivel de interferencia aceptable, cualquiera de ellas podrá recabar la asistencia de la Oficina; en tal caso facilitará la información necesaria para que la Oficina pueda tratar de efectuar la coordinación.

Subsección IID. Procedimiento que ha de seguirse cuando <u>no se da una</u> <u>respuesta,</u> no se toma una decisión o persiste el desacuerdo tras una solicitud de coordinación

F/54/81 MOD S9.60

Si una administración a la que se solicita la coordinación no comunica su decisión al respecto con arreglo a lo dispuesto en el número S9.51 o, después de tomar las medidas indicadas en el número S9.52, no contesta, no indica su decisión o no proporciona información respecto a sus propias asignaciones, en las que se basa su desacuerdo en el plazo de cuatro meses especificado en el número S9.51, la administración que solicite el acuerdo puede recabar la asistencia de la Oficina.

NOC S9.61

La Oficina, en respuesta a una solicitud de asistencia con arreglo al número S9.60, solicitará de inmediato a la administración interesada que comunique a la mayor brevedad posible su decisión al respecto.

NOC S9.62

Si la administración interesada no comunica su decisión al respecto en el plazo de treinta días tras la petición de la Oficina con arreglo al número S9.61, se aplicarán las disposiciones de los números S9.48 y S9.49.

NOC S9.63

Si persiste el desacuerdo o si cualquier administración interesada en el asunto recaba la asistencia de la Oficina, ésta solicitará toda la información necesaria para permitir a dicha administración evaluar la interferencia. La Oficina comunicará sus conclusiones a las administraciones interesadas.

F/54/82 MOD S9.64

Si después de que la Oficina ha comunicado sus conclusiones a las administraciones el desacuerdo sigue sin resolverse, la administración que ha solicitado la coordinación deberá, habida cuenta de las demás disposiciones de la presente sección, aplazar la presentación de sus notificaciones de asignación de frecuencia a la Oficina en virtud de lo dispuesto en el artículo S11 durante seis meses a partir de la fecha de la solicitud o de la circular semanal que contiene la solicitud de coordinación, según proceda.

F/54/83 MOD S9.65

Si en la fecha de recepción de una notificación en virtud del número S9.64 la Oficina tiene ya conocimiento de que persiste el desacuerdo, <u>la Oficina</u> deberá examinar la notificación con arreglo <u>al número a los números [S11.32bis] o S11.33¹</u> y actuar de acuerdo con el número S11.38.

F/54/84 MOD S9.65.1

Una notificación de asignación de frecuencia para la que se ha solicitado coordinación con arreglo a lo dispuesto en el número S9.21 y sobre la que continúa habiendo desacuerdo, no se examinará según lo estipulado en-el número los números [S11.32bis] o S11.33 sino que se considerará de acuerdo con el número S9.31S11.31 teniendo en cuenta el número S5.45.

INTERNATIONAL TELECOMMUNICATION UNION



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/9-E 6 November 1995 Original: English only

GENEVA.

23 OCTOBER

17 NOVEMBER 1995

AD HOC GROUP 5C1

Draft

REPORT FROM AD HOC GROUP 5C1 TO WORKING GROUP 5C

1 Feeder links in bands around 5 GHz

Proposed changes to the Table of Frequency Allocations in the band 5 000 - 5 250 MHz

MHz 4 800 - 5 725

Allocation to Services									
Region 1	Region 2	Region 3							
5 000 – 5 250 5 150 AERONAUTICAL RADIONAVIGATION									
733 [MOD 796] <u>[796A]</u> —797—797A—797B									
<u>5-0005 150</u> – 5 250	AERONAUTICAL RADIONA	VIGATION							
	FIXED-SATELLITE SERVICE (Earth-to-space)								
	733 796 MOD 797 797A 7	97B <u>[797C] 797D 797E</u>							

MOD 733

The bands 1 610 - 1 626.5 MHz, 5 000 - 5 2505 150 MHz and 15.4 - 15.7 GHz are also allocated to the aeronautical mobile-satellite (R) service on a primary basis. Such use is subject to agreement obtained under the procedure set forth in Article 14.

IMOD 796

See Document DT/65.]

[ADD 796A

See Document DT/65.]

MOD 797

The bands 5 000 – 5 250 MHz and 15.4 - 15.7 GHz are is also allocated to the fixed-satellite service and the inter-satellite service, for connection between one or more earth stations at specified fixed points on the Earth and space stations, when these services are used in conjunction with the aeronautical radionavigation and/or aeronautical mobile (R) service. Such use shall be subject to agreement obtained under the procedure set forth in Article 14.

- 2 -CMR95/DL/9-E

NOC 797A

NOC 797B

ADD 797C

This allocation is limited to feeder links for non-geostationary-satellite systems of the mobile-satellite service and is subject to the coordination and notification procedures set forth in Resolution 46.

ADD 797D

The band 5 150 - 5 216 MHz is also allocated to the fixed-satellite service (space-to-Earth) on a primary basis. This allocation is limited to feeder links for non-geostationary-satellite systems of the mobile-satellite service and is subject to the coordination and notification procedures set forth in Resolution 46. The power flux-density at the Earth's surface produced by space stations of the fixed-satellite service operating in the space-to-Earth direction in the band 5 150 - 5 216 MHz shall in no case exceed -164 dB(W/m²) in any 4 kHz band for angles of arrival.

ADD 797E

Administrations responsible for fixed-satellite service networks in the band 5 150 - 5 250 MHz operated under Nos. **797C** and **797D** shall coordinate on an equal basis in accordance with Resolution **46** with administrations responsible for non-geostationary satellite networks operated under No. **797A** and brought into use prior to 17 November 1995. Satellite networks operated under No. **797A** brought into use after 17 November 1995 shall not claim protection and shall not cause harmful interference to stations of the fixed-satellite service operated under Nos. **797C** and **797D**.

2 Feeder links in bands around 6/7 GHz

Proposed changes to the Table of Frequency Allocations in the band 5 925 - 7 075 MHz

MHz 5 725 - 7 300

	Allocation to Services										
Region 1	Region 2	Region 3									
5 925 – 7 075 [XXXX]	FIXED										
	FIXED-SATELLITE (Earth-to-space) MOD 792A										
	MOBILE										
	791 [MOD 809]										
<u>5-925[XXXX</u> - 7 075]	FIXED										
	FIXED-SATELLITE (Earth-to-space) (space-to-Earth) MOD 792A [809A] [809B]										
	MOBILE										
	791 —[MOD 809]										

NOC 791

MOD 792A The use of the bands 4 500 - 4 800 MHz (space-to-Earth),

Orb-88 6 725 - 7 025 MHz (Earth-to-space), 10.7 - 10.95 GHz (space-to-Earth),

11.2 - 11.45 GHz (space-to-Earth), and 12.75 - 13.25 GHz (Earth-to-space) by the fixed-satellite service shall be in accordance with the provisions of

Appendix 30B.

IMOD 809 In the band 6 425 - 7 075 MHz, passive microwave sensor

measurements are carried out over the oceans. In the band 7 075 - 7 250 MHz, passive microwave sensor measurements are carried out. Administrations should bear in mind the needs of the earth exploration-satellite (passive) and

space research (passive) services in their future planning of this band.]

[ADD 809A The space-to-Earth allocation to the fixed-satellite service is limited

to feeder links for non-geostationary-satellite systems of the mobile-satellite service and is subject to the coordination and notification procedures set forth in Resolution 46. [The use of this space-to-Earth allocation by feeder links is

not subject to the provisions of No. 2613]*.]

[ADD 809B Pending discussions related to the Earth-to-space allocation to the

fixed-satellite service in the band 7 025 - 7 075 MHz.]

3 Feeder links in the band 8 GHz to 15 GHz

Pending discussions in Drafting Group 5C1-5.

J. ALBUQUERQUE Chairman of ad hoc Group 5C1, Box No. 400

The issue of existing GSO/FSS systems will be discussed in Working Group 5C and this footnote has to be modified accordingly.



CMR-95

CONFERENCIA MUNDIAL DE RADIOCOMUNICACIONES

Documento DL/10(Rev.1)-S 7 de noviembre de 1995 Original: inglés

GINEBRA, 23 DE OCTUBRE – 17 DE NOVIEMBRE DE 1995

GRUPO DE REDACCIÓN 5B-a

Presidente del Grupo de Redacción 5B-a

PROYECTO DE INFORME DEL GRUPO DE REDACCIÓN 5B-a SOBRE LAS DISPOSICIONES TRANSITORIAS EN LA BANDA DE 2 GHZ

El Grupo de Redacción se creó para elaborar disposiciones transitorias para el acceso del SMS a las bandas de 2 GHz según el mandato definido en el Documento DT/49(Rev.1).

La fecha de entrada en vigor de las bandas atribuidas al SMS en la gama de 2 GHz ha sido objeto de repetidos debates en varios foros. Estas atribuciones al SMS están altamente restringidas por la fecha de acceso del 1 de enero de 2005. Un grupo de administraciones considera que es necesario adelantar la fecha al año 2000 para el desarrollo oportuno de los sistemas del SMS. Otras administraciones, que tienen sistemas terrenales importantes para sus comunicaciones gubernamentales y no gubernamentales, consideran que el adelanto de la fecha les causaría graves dificultades.

Después de la segunda sesión del Grupo de Redacción 5B-a celebrada el 4 de noviembre de 1995, se acordó que se constituyese un Subgrupo para tratar el alcance y contenido de las posibles notas del artículo 8 con respecto a las disposiciones transitorias relativas a la banda 2 GHz.

El Subgrupo se reunió el 4 de noviembre de 1995 bajo la coordinación del Sr. Carlos Merchan de México y llegó a la conclusión de que sería apropiado reflejar el alcance del proyecto de nota número **746D** en el contexto del proyecto de Resolución [COM5-X]. Esto se aceptó sobre la base de simplificar la naturaleza de las notas del artículo **8** de acuerdo con las recomendaciones del GVE y que los elementos transitorios se reflejen mejor en el proyecto de Resolución, especialmente debido a los puntos delicados de este asunto. El texto propuesto del número **746D** sería el siguiente:

"ADD 746D(Rev.) La utilización de las bandas [...] está sujeta a las disposiciones de la Resolución [COM5-X]."

El Subgrupo completó también el texto propuesto para las medidas transitorias relacionadas con los sistemas de dispersión troposférica en la banda del enlace ascendente del SMS en 2 GHz, pero no pudo llegar a la conclusión de si este texto se debe incluir en la Resolución [COM5-X] o como una nota del artículo 8 (por ejemplo, número 746Dbis). El texto es el siguiente:

- 2 -CMR95/DL/10(Rev.1)-S

"Se insta a las administraciones a que adopten todas las medidas prácticas para eliminar gradualmente los sistemas de dispersión troposférica que funcionan en la banda [enlace ascendente] MHz en [fecha XXX]. En la banda antes mencionada no se [pondrán/deben poner] en servicio nuevos sistemas de dispersión troposférica después de [fecha XXX]."

Con miras a facilitar el trabajo del Grupo de Redacción 5B-a y tomando en consideración el trabajo del Subgrupo, el Presidente del GR 5B-a presenta un proyecto revisado de la Resolución [COM5-X] que tiene en cuenta los debates celebrados durante la segunda sesión del GR 5B-a el 4 de noviembre de 1995 y los resultados del trabajo del Subgrupo.

Este proyecto de Resolución revisado contiene esencialmente los cinco elementos principales siguientes e intenta abarcar de una manera más exhaustiva el mandato definido en el Documento DT/49(Rev.1):

- aplicación de la Resolución 46 y situación de los sistemas terrenales;
- protección de los sistemas terrenales existentes;
- facilitar la introducción del SMS y las disposiciones transitorias para el SF;
- tratar las cuestiones de interés para los países en desarrollo;
- necesidad de ulteriores estudios en el UIT-R.

RESOLUCIÓN [COM5-X](Rev.1)

UTILIZACIÓN DE LAS BANDAS DE FRECUENCIA [ENLACE ASCENDENTE] Y [ENLACE DESCENDENTE] POR LOS SERVICIOS FIJO Y MÓVIL POR SATÉLITE Y DISPOSICIONES TRANSITORIAS ASOCIADAS

La Conferencia Mundial de Radiocomunicaciones (Ginebra, 1995),

considerando

- a) que la utilización de las bandas de frecuencias [enlace ascendente] MHz y [enlace descendente] MHz por el servicio móvil por satélite está sujeta a una fecha de entrada en vigor de [fecha];
- b) que estas bandas están compartidas con los servicios fijo y móvil a título primario y que se utilizan ampliamente por el servicio fijo en numerosos países;
- c) que los estudios efectuados han demostrado que, si bien la compartición del servicio móvil por satélite con el servicio fijo sería generalmente viable a corto y medio plazo, a largo plazo la compartición será compleja y difícil en ambas bandas por lo que sería aconsejable transferir las estaciones del servicio terrenal que funcionan en las bandas en cuestión a otros segmentos del espectro;
- d) que para muchos países en desarrollo la utilización de la banda 2 GHz ofrece una ventaja sustancial para sus redes de radiocomunicaciones y no es factible transferir estos sistemas a bandas de frecuencia superiores debido a las consecuencias económicas que ello acarrearía;
- e) que en respuesta a la Resolución 113 (CAMR-92) el UIT-R ha elaborado un nuevo plan de frecuencias para el servicio fijo en la banda de 2 GHz, establecido en la Recomendación UIT-R F.1098 (que podría ser modificada para dar cabida a la atribución al servicio móvil por satélite decidida en esta Conferencia), el cual facilitará la introducción de nuevos sistemas del servicio fijo en segmentos de la banda que no se superponen con las atribuciones al servicio móvil por satélite anteriormente mencionadas en 2 GHz;
- f) que no es generalmente viable la compartición entre sistemas del servicio fijo que utilizan la dispersión troposférica y los enlaces Tierra-espacio del servicio móvil por satélite en los mismos segmentos de la banda de frecuencias;

reconociendo

a) que la CAMR-92 ha identificado las bandas 1 885 - 2 025 MHz y 2 110 - 2 200 MHz para utilización mundial por el FSPTMT, con el componente de satélite limitado a las frecuencias 1 980 - 2 010 y 2 170 - 2 200 MHz, y que el desarrollo del FSPTMT puede ofrecer grandes posibilidades para ayudar a los países en desarrollo a que preparen con mayor rapidez su infraestructura de telecomunicaciones;

reconociendo además

que en la Resolución 22 (CAMR-92), "Asistencia a los países en desarrollo para facilitar la implantación de los cambios de atribuciones de bandas de frecuencias que necesitan la transferencia de asignaciones existentes", la CAMR-92 resolvió solicitar a la BDT que al formular sus planes inmediatos de asistencia a los países en desarrollo considere la introducción de modificaciones precisas en las redes de radiocomunicaciones de éstos y que una futura conferencia mundial de desarrollo considere las necesidades de los países en desarrollo y les ayude con los recursos necesarios para introducir las modificaciones precisas en sus redes de radiocomunicaciones;

resuelve

- 1 que para facilitar la aplicación de las disposiciones de la Resolución 46 relativas a la coordinación de las redes del servicio móvil por satélite con los servicios terrenales actuales o en proyecto:
- 1.1 las administraciones notifiquen a la Oficina de Radiocomunicaciones las asignaciones de frecuencias que requieran protección o que sean típicas de aquellas que requieren protección de sus estaciones del servicio fijo y móvil existentes o proyectadas que funcionan en las bandas de frecuencias [enlace ascendente] MHz y [enlace descendente] MHz antes de [XXX]; las notificaciones recibidas por la Oficina de Radiocomunicaciones después de [XXX] llevarán un asterisco en el registro indicando que no están en conformidad con esta Resolución [y se registran únicamente para fines de información];
- 1.2 aquellas estaciones de los servicios fijo y móvil terrenales existentes a las que se aplican las disposiciones del artículo 48 de la Constitución sean objeto de negociaciones bilaterales entre las administraciones interesadas;
- que, en cuanto a las estaciones de los servicios fijo y móvil tenidas en cuenta en la aplicación de la Resolución 46, las administraciones responsables de las redes del servicio móvil por satélite en la banda [enlace descendente] MHz intenten por todos los medios asegurar que no se produce una interferencia inaceptable a las estaciones del servicio fijo y móvil notificadas antes de [XXX];
- que para facilitar la introducción y la utilización en el futuro de las bandas de 2 GHz por el servicio móvil por satélite:
- 3.1 se inste a las administraciones a que las asignaciones de frecuencia a nuevos sistemas del servicio fijo que hayan de entrar en servicio después de [XXX] estén de acuerdo con los planes de canales que no se superpongan con las bandas de [enlace ascendente] MHz y [enlace descendente] MHz, por ejemplo como se señala en la Recomendación UIT-R F.1098;
- [3.2] se inste a las administraciones a que adopten todas las medidas posibles para suspender el funcionamiento de los sistemas por dispersión troposférica en la banda [enlace ascendente] MHz hasta [XXX]];

(Texto acordado en el Grupo de Redacción para su posible incorporación en la Resolución [COM5-X] o número **746Dbis**);

3.3 se inste a las administraciones a que redacten planes para la transferencia gradual de las asignaciones de frecuencia a sus estaciones de los servicios fijo y móvil en las bandas [enlace ascendente] MHz y [enlace descendente] MHz a bandas que no se superpongan, dando prioridad a la transferencia de sus asignaciones de frecuencia de la banda [enlace ascendente] MHz;

- 4 que cuando la coordinación de las redes del servicio móvil por satélite implica a redes del servicio fijo en países en desarrollo:
- 4.1 las administraciones responsables de la introducción de los sistemas móviles por satélite reconozcan la inquietud de los países en desarrollo por reducir al mínimo el posible efecto económico de las medidas transitorias sobre los sistemas actuales;
- 4.2 que se invite a la BDT, conjuntamente con la Oficina de Radiocomunicaciones, a que proporcione asistencia a los países en desarrollo que la soliciten para la introducción de modificaciones específicas en sus redes de radiocomunicaciones que habrán de facilitar su acceso a las nuevas tecnologías en desarrollo para la banda de 2 GHz;
- [5 que el UIT-R urgentemente desarrolle ulteriores estudios, conjuntamente con la Oficina de Radiocomunicaciones, para:
- **5.1** elaborar lo antes posible los instrumentos de evaluación de interferencia necesarios para asistir a las administraciones afectadas en la coordinación bilateral detallada de los sistemas móviles por satélite;
- **5.2** desarrollar lo antes posible los instrumentos de planificación necesarios para asistir a las administraciones que examinan una replanificación gradual de sus redes fijas terrenales en la banda de 2 GHz;]

encarga al Director de la Oficina de Radiocomunicaciones

que presente un Informe sobre la aplicación de esta Resolución a la Conferencia Mundial de Radiocomunicaciones de 1999.

W. TALLAH Presidente del Grupo de Redacción 5B-a

INTERNATIONAL TELECOMMUNICATION UNION



WORLD RADIOCOMMUNICATION CONFERENCE Document DL/10-E 6 November 1995 Original: English only

GENEVA. 23 OCTOBER

17 NOVEMBER 1995

DRAFTING GROUP 5B-a

Chairman, Drafting Group 5B-a

DRAFT REPORT OF DRAFTING GROUP 5B-A ON TRANSITIONAL ARRANGEMENTS IN THE 2 GHZ BANDS

The drafting group was established to develop transitional arrangements for MSS access to the 2 GHz bands under the terms of reference defined in Doc DT/49(Rev 1).

The date of entry into force of the bands allocated to MSS in the 2 GHz range has been the subject of repeated discussions in several fora. These MSS allocations are presently constrained by a date of access of 1 January 2005. A group of administrations considers that the advancement of the date to the year 2000 is necessary for the timely development of MSS systems. Others, having terrestrial systems important for their governmental and non-governmental communications, consider that the advancement of the date will cause them serious difficulties.

Following the second meeting of drafting group 5B-a on November 4, 1995, it was agreed that a subgroup be constituted to address the scope and content of possible Article 8 footnotes in respect of 2 GHz transition arrangements.

The sub-group met on November 4, 1995 under the coordination of Mr. Carlos Merchan of Mexico and concluded that it would be appropriate to reflect the scope of the draft RR 746D footnote into the context of the draft Resolution [COM5-X]. This was taken on the basis of simplifying the nature of Article 8 footnotes pursuant to the VGE recommendations and that the transitional elements should be better reflected in the draft Resolution, especially given the sensitivities involved. The proposed text for RR746D would be the following:

"ADD 746D (Rev.) The use of the bands [...] is subject to the provisions of Resolution [COM5/X]."

The sub-group also concluded proposed text for transition measures related to troposcatter systems in the 2 GHz MSS uplink band, but could not conclude whether such text should be included in the Resolution [COM5-X] or as an Article 8 footnote (eg RR 746D bis). The agreed text is as follows:

"Administrations are urged to take all practicable steps to phase out troposcatter systems operating in the band [uplink] MHz by [date XXX]. New troposcatter systems [shall/should] not be brought into operation in the above mentioned band after [date XXX]."

- 2 -CMR95/DL/10-E

In order to expedite the work of 5B-a and taking into consideration the work of the sub-group, the chairman of 5B-a submits for your consideration a revised draft version of Resolution [COM5-X] taking into account the discussions held during the second 5B-a meeting on November 4, 1995 and the results of the work of the sub-group.

This revised draft Resolution essentially contains the following five major elements and seeks to address in a more exhaustive manner the terms of reference defined in Doc. DT/49(Rev 1):

- application of Resolution 46 and filing of terrestrial systems;
- protection of existing terrestrial systems;
- facilitating MSS introduction and transition arrangements for the FS;
- addressing concerns of developing countries;
- need for further studies in ITU-R.

RESOLUTION [COM5-X] (Rev 1)

USE OF THE FREQUENCY BANDS [UPLINK] AND [DOWNLINK] BY THE FIXED, MOBILE AND MOBILE-SATELLITE SERVICES & ASSOCIATED TRANSITION

 ,	 ARRANGEMEN		

The World Radiocommunication Conference (Geneva, 1995),

considering

- a) [that the date for the use of the frequency bands [uplink] MHz and [downlink] MHz by the mobile satellite service has been advanced to 1 January 2000];
- a) that the use of the frequency bands [uplink] MHz and [downlink] MHz by the mobile-satellite service is subject to a date of entry into force of [Date];
- b) that these bands in question are shared with the fixed and mobile services on a primary basis and that they are widely used by the fixed service in many countries;
- c) that the studies made have shown that, while sharing of mobile satellite service with the fixed service in the short to medium term would be generally feasible, in the long term sharing will be complex and difficult in both bands, so that it would be advisable to transfer the terrestrial service fixed and mobile service stations operating in the bands in question to other segments of the spectrum;
- d) that for many developing countries, the use of the 2 GHz band offers a substantial advantage for their radiocommunication networks and that it is not feasible to transfer these systems to higher frequency bands because of the economic consequences that would entail;
- e) that in response to Resolution 113 (WARC-92) the ITU-R has developed a new frequency plan for the Fixed Service in the 2 GHz band, set out in the Recommendation ITU-R F.1098 (which could be modified to accommodate the Mobile Satellite Service allocation decision of this Conference), which will facilitate the introduction of new Fixed Service systems in band segments that do not overlap with the above mentioned mobile satellite service allocations at 2 GHz;
- f) that sharing between fixed-service systems using tropospheric scatter and Earth-to-space links in the mobile-satellite service in the same frequency band segments is generally not feasible;

recognizing

- 4 -CMR95/DL/10-E

a) that WARC-92 earmarked identified the bands 1 885 - 2 025 MHz and 2 110 - 2 200 MHz for world-wide use by FPLMTS, the satellite component being limited to the frequencies 1 980 - 2 010 and 2 170 - 2 200 MHz, and that the development of FPLMTS can offer great potential in helping the developing countries develop more rapidly their telecommunications infrastructure to bridge the gap between them and the developed countries more effectively;

recognizing further

that in Resolution 22 (WARC-92), "Assistance to the Developing Countries to Facilitate the Implementation of Changes in Frequency Band Allocations Which Necessitate the Transfer of Existing Assignments", WARC-92 resolved to request the BDT, when formulating its immediate plans for assistance to the developing countries, to consider the introduction of specific modifications in the radiocommunication networks of the developing countries and that a future world development conference should consider the needs of developing countries and should assist them with the resources needed to implement the required modifications to their radiocommunication networks;

resolves

- 1. that a procedure should be applied for the gradual transfer of the frequency assignments to fixed and mobile service stations operating the frequency bands [uplink] MHz and downlink MHz, as follows:
- 1. that to facilitate the application of the provisions of Resolution 46 for coordination of mobile satellite service networks with existing or planned terrestrial services:
- administrations shall notify the Radiocommunication Bureau of frequency assignments either requiring protection or typical of those requiring protection to their existing or planned fixed and mobile service stations operating in the frequency bands [uplink] MHz and [downlink] MHz before [Date];
- 1.2 those existing fixed and mobile terrestrial stations to which the provisions of Article 48 of the Constitution applies should be the subject of bilateral negotiations between the concerned Administrations:
- 1.3 notifications received by the Radiocomunications Bureau after [Date] shall carry a remark in the register that they are not in conformity with this Resolution and are registered solely for information purpose.
- that in respect of stations of the fixed and mobile services taken into account in the application of Resolution 46, administrations planning to establish satellite networks for the mobile satellite service responsible for mobile satellite service networks in the band [downlink] MHz shall make all practicable efforts to ensure that unacceptable interference is not caused to fixed and mobile service stations in operation before [Date];

- 5 -CMR95/DL/10-E

4. that Administrations should be urged to refrain from assigning frequencies to new fixed and mobile service stations
in the bands [uplink]MHz and [downlink] MHz and that, as far as possible, frequency assignments to new fixed service
systems in the 2 GHz band should be in accordance with channel plans which do not overlap with the above mentioned
bands, for example as given in Recommendation ITU-R F.1098;

- 3 that to facilitate the introduction and future use of the 2GHz bands by the mobile satellite service:
- 3.1 Administrations are urged that frequency assignments to new fixed service systems, to be brought into operation after [Date], shall be in accordance with channel plans which do not overlap with the [uplink] MHz and [downlink] MHz bands, for example as given in Recommendation ITU-R F.1098;
- [3.2 administrations are urged to take all practicable steps to cease the operation of troposcatter systems in the band [uplink] MHz by [Date]]

(Text agreed in drafting group for possible incorporation in Resolution [COM5-X] or RR 746Dbis)

- 3.3 administrations shall draw up plans for the gradual transfer, of the frequency assignments to their fixed and mobile service stations in the bands [uplink] MHz and [downlink] MHz to non-overlapping bands, giving priority to the transfer of their frequency assignments in the band [uplink] MHz.
- 2 that when coordination of mobile satellite service network involves developing countries terrestrial fixed service networks, measures should be considered that minimise economic implications for them;
- 4 that when coordination of mobile satellite service networks involves developing countries fixed service networks:
- 4.1 Administrations responsible for the introduction of mobile satellite systems should recognise the concerns of developing countries to minimize the possible economic impact of transition measures in respect to existing systems;
- 4.2 that the BDT should be invited, in conjunction with the Radiocommunication Bureau, to provide assistance to developing countries requesting it for the introduction of specific modifications to their radiocommunication networks that will facilitate their access to the new technologies being developed in the 2 GHz band;
- [5. that the ITU-R should urgently conduct further studies, in conjunction with the Radiocommunication Bureau, to:
- 5.1 develop the necessary interference assessment tools as soon as possible to assist concerned administrations in the detailed bilateral coordination of mobile satellite systems;

replanning of their terrestrial fixed networks in the 2 GHz range;]
instructs the Director of the Radiocommunication Bureau
to submit a report on the implementation of this Resolution to the 1999 World Radiocommunication Conference.
•
•
W. TALLAH
Chairman of Drafting Group 5B-a

INTERNATIONAL TELECOMMUNICATION UNION



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/11-E 6 November 1995 English only

GENEVA.

23 OCTOBER

17 NOVEMBER 1995

COMMITTEE 5

Note by the Chairman of Committee 5

INFORMATION RECEIVED BY THE RADIOCOMMUNICATION BUREAU CONCERNING SPACE STATIONS WITH FREQUENCY ASSIGNMENTS IN THE BANDS 18.8 - 19.7 AND 28.5 - 29.5 GHZ

Attached is a list of geostationary and non-geostationary satellite networks with frequency assignments in the bands 18.8 - 19.7 and 28.5 - 29.5 GHz for which information has been received by the Radiocommunication Bureau up until 1 November 1995.

This list provides the indication of the date of receipt of information with respect to the different stages of the coordination and notification process. Information is given for cases received for Advance publication (Section I of Article 11) and not yet published, those which have been advance published, those received for the publication of a coordination request (Section II of Article 11) and not yet published, those which have had the coordination request published, those received for examination under Article 13 and not yet recorded in the MIFR or returned, and those already recorded in the MIFR.

G. F. JENKINSON

Chairman of Committee 5

LIST OF SPACE STATIONS WITH FREQUENCY ASSIGNMENTS IN THE 18.8-19.7 GHz AND 28.5-29.5 GHz BANDS

CASES RECEIVED UP TO 1.11.95 - DATES IN COLUMNS WITH AN "*" ARE PROVISIONAL PENDING CONFIRMATION OF NOTICE COMPLETENESS

Nr	ADM	NAME OF	ORB POS	Date of receipt of AP4 - RR1042	AR11/A Date of receipt	Date of receipt of AP3 - RR1064	AR11/C Date of receipt	Date of receipt - RR1488	IN the	18.8 19.7	28.5 29.5
		SPACE STATION	103	(for AR11/A)*	of AP4	(for AR11/C)*	of AP3 - RR1064	(for Art. 13)*	MIFR	GHz	GHz
5	CAN	CANSAT KA-1	107.3W	(100)	01.11.94	20.10.95		(14111111111111111111111111111111111111		X	X
	CAN	CANSAT KA-2	111.1W		01.11.94	20.10.95				Х	X
	CAN	CANSAT KA-3	118.7W		01.11.94	20.10.95				Х	X
ŀ	CAN	CANSAT-KA-4	111.1W	20.10.95						Х	X
	CAN	CANSAT-KA-5	118.7W	20.10.95						Х	Х
7	D	DFS-1	023.5E				09.01.84		Х	Х	
	D	DFS-2	028.5E				09.01.84		Х	Х	
	D	DFS-5	033.5E				15.04.91			Х	
	D	DFS-6	026.0E				11.11.91			Х	
	D	QUASIGEO-L1	9 99 9. 9 E		19.01.93					Х	Х
	D	QUASIGEO-L2	9999.9E		19.01.93					Х	X
İ	D	QUASIGEO-L3	9999.9E		19.01.93					Х	X
2	F	F-SAT ICO	9999.9E		19.11.93					Х	Х
	F	F-SAT LEO	9999.9E		19.11.93					Х	X
8	F/ESA	ARTEMIS-16.4E DR	016.4E		07.05.93	25.07.94				Х	Х
	F/ESA	ARTEMIS-21.5E DR	021.5E		07.05.93	25.07.94				Х	X
	F/ESA	EDRSS-E	059.0E		02.04.90	08.05.95				Х	Х
	F/ESA	EDRSS-EC	047.0E		02.04.90	08.05.95				Х	Х
}	F/ESA	EDRSS-W	044.0W		02.04.90	08.05.95				Х	Х
	F/ESA	EDRSS-WC	032.0W		02.04.90	08.05.95				Х	Х
1	F/ESA	EURECA	9999.99		03.02.86					Х	X
İ	F/ESA	L-SAT	019.0W				31.10.83		Х	Х	X
20	F/EUT	EUTELSAT-KA-4E	004.0E	04.09.95						Х	Х
	F/EUT	EUTELSAT-KA-7E	007.0E	04.09.95						Х	Х
	F/EUT	EUTELSAT-KA-10E	010.0E	04.09.95						Х	Х
i	F/EUT	EUTELSAT-KA-13E	013.0E	04.09.95						Х	X
l	F/EUT	EUTELSAT-KA-16E	016.0E	04.09.95	<u>l</u>					Х	X
İ	F/EUT	EUTELSAT-KA-21.5E	021.5E	04.09.95		<u> </u>				Х	Х
ļ	F/EUT	EUTELSAT-KA-25.5E	025.5E	04.09.95						Х	X
	F/EUT	EUTELSAT-KA-33E	033.0E	04.09.95						Х	Х
	F/EUT	EUTELSAT-KA-36E	036.0E	04.09.95						Х	Х
I	F/EUT	EUTELSAT-KA-44E	044.0E	04.09.95						Х	X
	F/EUT	EUTELSAT-KA-48E	048.0E	04.09.95						Х	Х
	F/EUT	EUTELSAT-KA-70.5E	070.5E	04.09.95						Х	X
	F/EUT	EUTELSAT-KA-76E	076.0E	04.09.95						Х	Х
l	F/EUT	EUTELSAT-KA-80.5E	080.5E	04.09.95						Х	Х
l	F/EUT	EUTELSAT-KA-83.5E	083.5E	04.09.95						Х	X

Nr	ADM	NAME	ORB	Date of receipt	AR11/A	Date of receipt	AR11/C	Date of	IN	18.8	28.5
	Į	OF	POS	of AP4 - RR1042	Date of receipt	of AP3 - RR1064	Date of receipt	receipt - RR1488	the	19.7	29.5
		SPACE STATION		(for AR11/A)*	of AP4	(for AR11/C)*	of AP3 - RR1064	(for Art. 13)*	MIFR	GHz	GHz
	F/EUT	EUTELSAT-KA-88.5E	088.5E	04.09.95						Х	X
	F/EUT	EUTELSAT-KA-12.5W	012.5W	04.09.95						Х	Х
	F/EUT	EUTELSAT-KA-14.8W	014.8W	04.09.95						Х	X
	F/EUT	EUTELSAT-KA-37.2W	037.2W	04.09.95						Х	Х
	F/EUT	EUTELSAT-KA-64W	064.0W	04.09.95						Х	Х
19	G	AFRISAT-1	019.0E		27.02.95	27.08.95				Х	X
	G	AFRISAT-2	032.0E		27.02.95	27.08.95				Х	Х
	G	AFRISAT-3	038.0E		27.02.95	27.08.95				Х	Х
	G	AFRISAT-4	064.5E		27.02.95	27.08.95				Х	Х
	G	SAMSAT-1	075.0W		27.02.95	27.08.95				Х	Х
	G	SAMSAT-2	082.0W		27.02.95	27.08.95		<u></u>		Х	X
	G	SAMSAT-3	089.0W		27.02.95	27.08.95				Х	Х
	G	SKYSAT-A1	118.3E		05.08.94	05.02.95				Х	X
	G	SKYSAT-A2	121.5E		05.08.94	05.02.95				Х	X
	G	SKYSAT-A3	124.7E		05.08.94	05.02.95				Х	Х
	G	SKYSAT-B1	133.2E		05.08.94	05.02.95				Х	Х
	G	SKYSAT-B2	136.4E		05.08.94	05.02.95				Х	X
l	G	SKYSAT-B3	139.6E		05.08.94	05.02.95				Х	Х
1	G	SKYSAT-B4	142.8E		05.08.94	05.02.95				Х	Х
	G	SKYSAT-C1	080.0E		28.09.94	28.03.95				Х	Х
l	G	SKYSAT-C2	090.0E		28.09.94	28.03.95				Х	Х
	G	SKYSAT-C3	101.5E		28.09.94	28.03.95				X	Х
	G	SKYSAT-C4	104.8E		28.09.94	28.03.95				Х	Х
	G	SKYSAT-C5	169.2E		28.09.94	28.03.95				Х	Х
1	HOL	PETALRING 30C-K	9999.99		22.11.93					Х	X
12	1	EUROSKYWAY-10.2E	010.2E	28.09.95						Х	X
l	1	EUROSKYWAY-13.2E	013.2E	28.09.95						Х	X
	1	EUROSKYWAY-16.4E	016.4E	28.09.95						Х	X
	1	EUROSKYWAY-22E	022.0E	28.09.95						Х	X
		EUROSKYWAY-30E	030.0E	28.09.95	<u> </u>					Х	X
	1	EUROSKYWAY-39E	039.0E	28.09.95						Х	X
	1	EUROSKYWAY-5E	005.0E	28.09.95	L					Х	Х
	1	ITALSAT	013.2E				25.05.85		Х	X	Х
		ITALSAT-10.2E	010.2E				24.09.93			X	X
		ITALSAT-13.2E	013.2E				24.09.93			Х	X
		ITALSAT-16.4E	016.4E				24.09.93			Х	Х
	1	SARIT-19W	019.0W		27.03.95					Х	
15	IND	INSAT-108W	108.0W	16.10.95						Х	Х
	IND	INSAT-13W	013.0W	16.10.95					ļ	Х	X
	IND	INSAT-75.2W	075.2W	16.10.95	<u> </u>					Х	X

Nr	ADM	NAME	ORB	Date of receipt	AR11/A	Date of receipt	AR11/C	Date of	IN	18.8	28.5
		OF	POS	of AP4 - RR1042	Date of receipt	of AP3 - RR1064	Date of receipt	receipt - RR1488	the	19.7	29.5
		SPACE STATION		(for AR11/A)*	of AP4	(for AR11/C)*	of AP3 - RR1064	(for Art. 13)*	MIFR	GHz	GHz
	IND	INSAT-KA-100	100.0E	16.10.95						Х	Х
	IND	INSAT-KA-50	050.0E	16.10.95						Х	Х
	IND	INSAT-KA-54	054.0E	16.10.95						Х	Х
	IND	INSAT-KA-63	063.0E	16.10.95						X	_ X
	IND	INSAT-KA-67	067.0E	16.10.95						X	Х
1	IND	INSAT-KA-74	074.0E	16.10.95						X	Х
	IND	INSAT-KA-76	076.0E	16.10.95						Х	Х
	IND	INSAT-KA-78	078.0E	16.10.95						Х	Х
	IND	INSAT-KA-83	083.0E	16.10.95						Х	X
	IND	INSAT-KA-87	087.0E	16.10.95						Х	Х
	IND	INSAT-KA-93.5	093.5E	16.10.95						Х	X
L	IND	INSAT-KA-98	098.0E	16.10.95						Х	Х
12	J	COMETS	121.0E		30.09.94	08.06.95				Х	X
1	J	CS-3A	132.0E				30.04.87		Х	Х	Х
	J	CS-3B	136.0E				30.04.87		Х	Х	X
	J	ETS-6-FS	154.0E				21.10.91			Х	X
	J	KIKU-6	9999.99		22.12.94			<u> </u>		X	X
	J	MTSAT-135E	135.0E	01.09.95	l		<u> </u>	<u> </u>		Х	X
	J	MTSAT-140E	140.0E	01.09.95	<u></u>		<u> </u>	<u> </u>	<u> </u>	Х	X
	J	MTSAT-145E	145.0E	01.09.95					<u> </u>	X	Х
	J	N-STAR-A	132.0E				16.11.91	06.04.95		Х	X
	J	N-STAR-B	136.0E				16.11.91	04.10.95		X	X
	J	SUPERBIRD-A	158.0E				18.03.88		X	Х	X
	J	SUPERBIRD-B	162.0E				18.03.88		Х	X	X
6	KOR	DACOMSAT-4	140.0E	03.10.95			<u></u>		L	Х	X
	KOR	EASTSAT	164.0E	01.08.95					<u> </u>	X	X
l	KOR	GLOBALSAT	177.5E	01.08.95					<u> </u>	X	Х
	KOR	INFOSAT-A	103.0E	01.08.95	ļ				<u> </u>	Х	X
	KOR	INFOSAT-B	113.0E	01.08.95					L	Х	X
	KOR	INFOSAT-C	116.0E	01.08.95				·		X	X
21	LUX	LUX-KA-19.0E	019.0E	24.04.95	1	24.10.95			<u> </u>	X	X
	LUX	LUX-KA-24.2E	024.2E	25.09.95						Х	X
	LUX	LUX-KA-26.2E	026.2E	25.09.95						Х	X
Ī	LUX	LUX-KA-28.2E	028.2E	25.09.95						Х	X
	LUX	LUX-KA-31.5E	031.5E	25.09.95						Х	X
	LUX	LUX-KA-35.5E	035.5E	25.09.95						Х	X
	LUX	LUX-KA-37.5E	037.5E	25.09.95						Х	Х
	LUX	LUX-KA-41.2E	041.2E	25.09.95						Х	X
	LUX	LUX-KA-43.2E	043.2E	25.09.95					L	Х	X
l	LUX	LUX-KA-60.2E	060.2E	20.10.95						Х	Х

Nr	ADM	NAME	ORB	Date of receipt	AR11/A	Date of receipt	AR11/C	Date of	IN	18.8	28.5
		OF	POS	of AP4 - RR1042	Date of receipt	of AP3 - RR1064	Date of receipt	receipt - RR1488	the	19.7	29.5
		SPACE STATION		(for AR11/A)*	of AP4	(for AR11/C)*	of AP3 - RR1064	(for Art. 13)*	MIFR	GHz	GHz
	LUX	LUX-KA-78.2E	078.2E	20.10.95						Х	Х
	LUX	LUX-KA-107.2E	107.2E	20.10.95						Х	X
	LUX	LUX-KA-123.2E	123.2E	20.10.95						Х	X
	LUX	LUX-KA-145.2E	145.2E	20.10.95						Х	Х
	LUX	LUX-KA-33.2W	033.2W	20.10.95						Х	Х
	LUX	LUX-KA-42.2W	042.2W	20.10.95	_					Х	Х
	LUX	LUX-KA-56.2W	056.2W	20.10.95						Х	X
	LUX	LUX-KA-79.2W	079.2W	20.10.95						Х	X
	LUX	LUX-KA-93.2W	093.2W	20.10.95						Х	X
	LUX	LUX-KA-123.2W	123.2W	20.10.95						Х	X
	LUX	LUX-KA-177.2W	177.2W	20.10.95						Х	Х
5	PAK	PAKSAT-A	038.0E	27.10.95						Х	X
	PAK	PAKSAT-B	041.0E	27.10.95						Х	X
	PAK	PAKSAT-C	030.0E	27.10.95						Х	X
	PAK	PAKSAT-D	088.0E	27.10.95						Х	X
	PAK	PAKSAT-E	101.0E	27.10.95						Х	X
4	RUS	GOMS-1M	014.5W				26.09.89	16.12.94			Х
	RUS	GOMS-2M	166.0E				08.11.89	09.06.95			X
	RUS	GOMS-M	076.0E				23.11.89	09.06.95			X
	RUS	LOUTCH-1	014.0W		22.04.88					R	R
1	S	SIRIUS-4	005.0E	29.06.95						Х	X
5	TON	TONGASAT-ELL-1	9999.99		31.03.92					Х	Х
	TON	TONGASAT-LEO-10000	9999.99		27.03.92			*******		Х	X
	TON	TONGASAT-LEO-1200	9999.99		27.03.92					Х	X
	TON	TONGASAT-LEO-1300	9999 99		27.03.92					Х	X
	TON	TONGASAT-RADIO/TV-8	9999.99		06.04.92				f		X
22	URS	TOR-1	026.5W				09.03.88			Х	1
	URS	TOR-11	003.0W				06.07.88			Х	1
	URS	TOR-12	015.0E				30.08.88			Х	
	URS	TOR-13	058.0E				14.07.88			Х	1
	URS	TOR-14	069.0E				06.10.88	4111	<u> </u>	X	
	URS	TOR-15	001.0E				14.07.88			Х	1
	URS	TOR-16	049.0E				06.10.88			Х	1
	URS	TOR-17	070.0E				06.12.88			Х	1
	URS	TOR-18	012.0E		,		06.12.88			Х	1
	URS	TOR-19	005.0E				03.01.89			Х	T
	URS	TOR-2	035.0E				09.03.88			Х	1
	URS	TOR-20	027.0E				06.12.88			Х	†

R - only frequencies 19.237GHz and 28.856 Ghz as propagation beacons in the space to Earth direction.

Nr	ADM	NAME	ORB	Date of receipt	AR11/A	Date of receipt	AR11/C	Date of	IN	18.8	28.5
	ŀ	OF	POS	of AP4 - RR1042	Date of receipt	of AP3 - RR1064	Date of receipt	receipt - RR1488	the	19.7	29.5
		SPACE STATION	<u> </u>	(for AR11/A)*	of AP4	(for AR11/C)*	of AP3 - RR1064	(for Art. 13)*	MIFR	GHz	GHz
	URS	TOR-21	032.0E				23.02.89			Х	
	URS	TOR-22	040.0E				21.02.89			Х	
	URS	TOR-23	053.0E				23.02.89			Х	
	URS	TOR-24	062.0E				28.02.89			Х	
	URS	TOR-25	065.0E				28.02.89			Х	
	URS	TOR-26	019.0E		19.04.88					Х	
	URS	TOR-3	045.0E				18.05.88		Х	Х	
	URS	TOR-4	085.0E				18.05.88		Х	X	
	URS	TOR-5	170.0W				02.06.88		Х	Х	
	URS	TOR-6	128.0E				19.04.88			Х	
25	USA	ACTS	100.0W				12.08.91		Х	Х	X
	USA	HIBLEO-2	9999.99		05.02.92					Х	Х
	USA	LEOSAT-1	9999.99		15.02.95			20.10.95		Х	Х
	USA	MSSLEO-1	9999.99		19.11.93					Х	Х
	USA	MSSLEO-2	9999 99		19.11.93					Х	Х
i i	USA	USASAT-27A	090.0W		28.10.92					Х	Х
	USA	USASAT-29A	101.0W		07.03.95					Х	Х
	USA	USASAT-29B	099.0W		07.03.95					Х	Х
	USA	USASAT-29C	050.0W		07.03.95					Х	T
	USA	USASAT-29D	025.0E		07.03.95					Х	Х
	USA	USASAT-29E	110.0E		07.03.95					Х	Х
	USA	USASAT-29F	175.0E		07.03.95					Х	X
	USA	USCSID-A1	0.00	07.08.95						Х	
	USA	USCSID-A2	044.0E	07.08.95						Х	
	USA	USCSID-A3	075.0E	07.08.95						Х	T
	USA	USCSID-A4	082.0E	07.08.95						Х	I
	USA	USCSID-A5	092.0E	07.08.95						Х	T
	USA	USCSID-A6	110.0E	07.08.95						X	T
	USA	USCSID-E1	010.0W	07.08.95						Х	T
Ī	USA	USCSID-E2	013.0W	07.08.95						Х	
	USA	USCSID-E3	024.0W	07.08.95						Х	
	USA	USCSID-E4	030.0W	07.08.95						Х	
	USA	USCSID-P	9999.99	07.08.95						Х	
	USA	USCSID-W1	141.0W	07.08.95						Х	1
	USA	USCSID-W2	144.0W	07.08.95						Х	

Nr	ADM	NAME	ORB	Date of receipt	AR11/A	Date of receipt	AR11/C	Date of	IN	18.8	28.5
i		OF	POS	of AP4 - RR1042	Date of receipt	of AP3 - RR1064	Date of receipt	receipt - RR1488	the	19.7	29.5
		- SPACE STATION		(for AR11/A)*	of AP4	(for AR11/C)*	of AP3 - RR1064	(for Art. 13)*	MIFR	GHz	GHz
7	USA/IT	INTELSAT-KA-33E	033.0E	14.06.95						Х	X
1	USA/IT	INTELSAT-KA-66E	066.0E	14.06.95			,			Х	X
	USA/IT	INTELSAT-KA-95E	095.0E	14.06.95						Х	X
	USA/IT	INTELSAT-KA-157E	157.0E	14.06.95						Х	Х
1	USA/IT	INTELSAT-KA-307E	053.0W	14.06.95	}			1		Х	X
	USA/IT	INTELSAT-KA-319.5E	040.5W	14.06.95						Х	X
1	USA/IT	INTELSAT-KA-359E	001.0W	14.06.95						Х	Х
S	19	197		100	56	30	41	6	12	193	156

INTERNATIONAL TELECOMMUNICATION UNION



WORLD RADIOCOMMUNICATION CONFERENCE Document DL/12-E 6 November 1995 English only

GENEVA. 23 OCTOBER

17 NOVEMBER 1995

DRAFTING GROUP 5B-a

Note by the Chairman of Drafting Group 5B-a

NON-GSO FSS

Basic Considerations

- The Conference should recognize that the rights and obligations of Administrations acquired in accordance with the Radio Regulations in force at the time they were required cannot be taken away from Administrations retroactively.
 - (No decision should be taken retroactively which affects the rights and obligations of Members).
- The provision for the identification of appropriate spectrum for the feeder links of non-GSO MSS systems in the Ka-band needs to be taken into account.
- No further debate should be undertaken on the need for the service; reference should be made, in general, to non-GSO FSS and not to a single system.
- Pursuant to Document DT/38 (Rev.2), the proposed solution consists of the inclusion of a footnote against the two sub-bands in the 20/30 GHz spectrum range for use by non-GSO FSS systems, and the adoption of a Resolution specifying the conditions of the use of these sub-bands. In addition, the following points are included in the resolves and further resolves provisions of the Draft Resolution contained in Doc. DT/38 (Rev.2):
 - * Resolution [46 (WRC-95)] shall be applied in the bands [] to frequency assignments of GSO and non-GSO satellite systems of the FSS communicated to the BR [after ...], according to the date of receipt of coordination information complete as required on the date that the information is communicated to the Bureau and to those of the terrestrial services to which these bands are allocated (see DT/42 -- note from the BR Director);
 - * when applying Resolution [46 (WRC-95)], [No.2613] shall not be taken into account;
 - * administrations should be urged to communicate to the BR their satellite systems in the bands [] prior to the date of WRC-95, to make every effort to reach agreement on the coordination of their respective systems;
 - * the ITU-R be requested, as a matter of urgency, to study the criteria to be applied for the sharing of the Ka-band between GSO and non-GSO satellite systems, on the one hand, and among those systems and the terrestrial services, on the other hand, taking account of existing and planned systems and to recommend the required revisions to the RRs;

- * the BR Director be instructed to ensure, in consultation with the chairmen of the SGs, Committees and meetings, that the results of these studies be available in due time for their consideration by WRC-97;
- * it be recommended that WRC-97 review the results of the sharing studies and take appropriate action for the harmonious development of GSO and non-GSO systems and the terrestrial services in the Ka-band;
- * administrations concerned be urged to cooperate to the maximum extent practicable in initiating the required coordination and to conduct them with the view to reaching results acceptable to all the parties concerned.
- Committee 5 is to: identify an appropriate bandwidth of spectrum; identify specific bands where the non-GSO FSS system(s) can be located, taking into account existing services, and the fact that sharing studies have not been completed.

 1

¹ If improvements can be identified with respect to sharing possibilities, recommendations can be provided to Committee 5 from the Ad Hoc Group.



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE

Document DL/13-E 7 November 1995 <u>Original</u>: English only

GENEVA,

23 OCTOBER

17 NOVEMBER 1995

COMMITTEE 5 AD HOC 2

NOTE BY THE CHAIRMAN OF 5 AD HOC 2

NON-GSO FSS: Points for Discussion

- 1 The focus of the discussions is on non-GSO FSS as a service to be implemented in Kaband, and not on any one system or systems.
- 2 As such, debates with respect to exclusivity, and/or the application of 2613, and/or the technical justification of any one system are beyond the purview and mandate of the Ad Hoc Group.
- 3 The determination of an appropriate amount of spectrum (bandwidth) to permit the operation of non-GSO FSS is dependent on the following factors:
- **3.1** the number of requirements based on the number of subscribers to be served, taking into account regional and demographic differences; and
- **3.2** the nature of the service to be provided.
- 4 The range of bandwidth considered to be appropriate for the implementation of non-GSO FSS in Ka-band (at the moment) is 50 to 500 MHz.
- In order to provide more specific guidance to the ITU-R on sharing studies to be undertaken by the relevant Study Groups, the Ad Hoc Group considered that, in addition to sharing between GSO and non-GSO systems, between non-GSO and terrestrial services, and GSO and terrestrial services, studies should also be focused on sharing between non-GSO FSS and non-GSO FSS.



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/14-E only 8 November 1995 Original: English

GENEVA. 2

23 OCTOBER

17 NOVEMBER 1995

COMMITTEE 5

TERMS OF REFERENCE OF 5 AD HOC 3 TO CONSIDER TECHNICAL ASPECTS IN RELATION WITH RESOLUTION 46

- A Taking into account the CPM Report and the technical results achieved concerning Resolution 46 in Committee 5, the 5 Adhoc Group 3 will be entrusted, in liaison with Committee 4:
 - to prepare draft texts concerning the achieved technical results of Committee 5 with adequate format to be introduced or referred to in Resolution 46,
 - to prepare texts with relevant formats for the following technical subjects which need further consideration:
 - coordination distances applicable between the different services
 - coordination thresholds (Annex 1 of Rec. ITU-R IS 1143)
 - affected region (doc ITU-R [8/1034]
 - to prepare texts for future work concerning the three above items, as necessary
- **B** The Adhoc Group will be chaired by
- C The Adhoc Group is required to finalize its report by



WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/15-E 9 November 1995 Original: English only

GENEVA,

23 OCTOBER

17 NOVEMBER 1995

C5 AD HOC 3

Note by the Chairman of C5 ad hoc 3

DRAFT APPENDIX XYZ

Determination methods for the need of coordination between space stations of the MSS (space-to-Earth) and terrestrial services in the 1 - 3 GHz frequency bands shared between MSS (space-to-Earth) and terrestrial services

1 Objectives

Generally, the necessity for coordination between terrestrial services and space stations of the MSS (space-to-Earth) was performed using power flux-density limits. However, to facilitate the sharing between digital FS stations and NGSO MSS space stations, the concept of fractional degradation in performance (FDP) was adopted. This concept involves new methodologies described in this Appendix.

As a consequence of this new concept, in this Appendix, the need for coordination between space stations of the MSS and terrestrial services is determined using two methods:

- the need for coordination is determined using FDP trigger value (typical characteristics of FS stations are used with the simplest definition of the MSS system) or power flux-density;
- the need for coordination is determined using a system specific methodology (SSM), typical characteristics of FS stations are used with specific characteristics of the MSS system.

There is a need for coordination in the situation where both methodologies give a negative result. If one of the two methods gives a positive result there is no need for coordination.

2 General considerations

2.1 Method for calculating the value of fractional degradation in performance (FDP)

FDP is used in case of sharing between digital FS stations with non-GSO MSS stations (space-to-Earth).

For calculating the value of the FDP, the following parameters are needed:

technical characteristics of digital FS station; and

- technical characteristics of non-GSO MSS constellation.

The FDP is calculated:

- by simulating the proposed MSS constellation;
- by positioning the FS station at a certain latitude (each station is assumed to operate at an elevation angle of 0°);
- by calculating for each pointing azimuth (Azi) varying between 0° and 360°;
- at each time instance of the simulation, the aggregate interference from all visible space stations received at the FS station.
- the FDP_{Azi} for the azimuth Azi using the following formula:

$$FDP_{Azi} = \sum_{I_i = \min}^{\max} \frac{I_i f_i}{N_T}$$

by the following formula:

$$FDP = max(FDP_{Azi})$$

(the formula for FDP applies to the 1 - 3 GHz frequency range only. A different formula may apply at frequencies above 3 GHz),

where:

 I_i = interference noise power level

 f_i = the fractional period of time that the interference power equals I_i

 N_T = station receiving system noise power level = kTB

 $k = boltzmann's constant = 1.38 \cdot 10^{-23}$

T =station receiving system effective noise temperature (T should be calculated by the following formula: $10 \log T = NF + 10 \log T_0$ where NF (dB) is the receiver noise figure given in Annex 1 and T_0 should be assumed as 290 K)

B = reference bandwidth = 1 MHz.

NOTE – For the purpose of FDP calculation according to this Appendix, it should be assumed that all space stations in the same MSS constellation operate on the same frequencies.

2.2 Characteristics of reference systems in the fixed service

The following parameters represent the set of reference parameters of the fixed service.

2.2.1 Characteristics of reference digital point-to-point systems

Three different digital systems are described in this table:

- 64 kbit/s capacity used for example for outside-plant (individual subscriber connection);
- 2 Mbit/s capacity used for example for business subscriber connections for the local part of the inside-plant;
- 45 Mbit/s capacity used for example for trunk networks.

- 3 -CMR95/DL/15-E

Capacity	64 kbit/s	2 Mbit/s	45 Mbit/s
Modulation	4-PSK	8-PSK	64-QAM
Antenna gain (dB)	33	33	33
Transmit power (dBW)	7	7	1
Feeder/multiplexer loss (dB)	2	2	2
e.i.r.p. (dBW)	38	38	32
Receiver IF bandwidth (MHz)	0.032	0.7	10
Receiver noise figure (dB)	4	4.5	4
Receiver input level for a BER of 10 ⁻³ (dBW)	-137	-120	-106
Maximum long-term interference	-165	-151	-136
Total power (dBW)			
Maximum long-term interference	-174	-173	-170
Power spectral density (dB(W/4 kHz))			

Antenna pattern:

$$\begin{split} G(\phi) &= G_{max} - 2.5 \text{ x } 10^{-3} \text{ } (D\phi/\lambda)^2 & \text{ for } 0 < \phi < \phi_{\textbf{m}} \\ G(\phi) &= G_1 & \text{ for } \phi_{\textbf{m}} \leq \phi < 75.86(\lambda/D) \\ G(\phi) &= 49 \text{ - } 10 \log \text{ } (D/\lambda) \text{ - } 25 \log \phi & \text{ for } 75.86(\lambda/D) \leq \phi < 48^{\circ} \\ G(\phi) &= 7 \text{ - } 10 \log \text{ } (D/\lambda) & \text{ for } 48^{\circ} \leq \phi \end{split}$$

 $G(\varphi)$ = gain relative to an isotropic antenna

 $\varphi = \text{ off-axis angle}$

D = antenna diameter

 λ = wavelength expressed in the same unit as D

 $G_1 = \text{gain of the first side lobe} = 2 + 15 \log (D/\lambda)$

 $(D/\lambda \text{ may be estimated from } 20 \log D/\lambda \approx G_{max} - 7.7)$

G_{max} = main-lobe antenna gain (dBi);

$$\varphi_{\rm m} = 20 \, (\lambda/{\rm D}) \, {\rm x} \, \sqrt{({\rm G}_{\rm max} - {\rm G}_{\rm I})} \, ({\rm degrees})$$

It should be noted that the above antenna radiation pattern corresponds to the "average" side-lobe pattern and it is recognized that individual side lobes may exceed the reduced side-lobe radiation pattern by up to 3 dB.

2.2.2 Characteristics of reference analogue point-to-point systems

Antenna gain (dBi)	33
e.i.r.p. (dBW)	36
Feeder/multiplexer loss (dB)	3
Receiver noise figure (referred to input of receiver) (dB)	8
Maximum long-term interference per link (20% of time) (dB(W/4kHz))	-170

Antenna pattern:

$$G(\phi) = G_{\text{max}} - 2.5 \text{ x } 10^{-3} (D_{\phi}/\lambda)^2$$
 for $0 < \phi < \phi_{\text{m}}$

$$G(\phi) = G_1 \qquad \qquad \text{for } \phi_m \leq \phi < 75.86(\lambda/D)$$

$$G(\phi) = 49 - 10 \log (D/\lambda) - 25 \log \phi \qquad \qquad \text{for } 75.86(\lambda/D) \le \phi < 48^{\circ}$$

$$G(\varphi) = 7 - 10 \log (D/\lambda)$$
 for $48^{\circ} \le \varphi$

 $G(\varphi)$ = gain relative to an isotropic antenna

 $\varphi = \text{ off-axis angle}$

D = antenna diameter

 λ = wavelength expressed in the same unit as D

 $G_1 = gain of the first side lobe = 2 + 15 log (D/<math>\lambda$)

(D/ λ may be estimated from 20 log D/ $\lambda \approx G_{max}$ - 7.7)

G_{max} = main-lobe antenna gain (dBi);

$$\varphi_{\rm m} = 20 \ (\lambda/{\rm D}) \ {\rm x} \ \sqrt{(G_{\rm max} - G_1)} \ ({\rm degrees})$$

It should be noted that the above antenna radiation pattern corresponds to the "average" side-lobe pattern and it is recognized that individual side lobes may exceed the reduced side-lobe radiation pattern by up to 3 dB.

2.2.3 Characteristics of reference point-to-multipoint systems

PARAMETER	CENTRAL STATION	OUTSTATION Dish/horn 20 analogue 27 digital		
Antenna type	Omni/sector			
Antenna gain (dBi)	10/17			
e.i.r.p. (max) (dBW) analogue digital	12 24	21 34		
noise figure (dB)	3.5	3.5		
feeder loss (dB)	2	2		

- 5 -CMR95/DL/15-E

IF bandwidth (MHz)	3.5	3.5
Maximum permissible long-term interference power (20% time)		
Total (dBW)	- 142	- 142
dB(W/4 kHz)	- 170	- 170
dB(W/MHz)	- 147	- 147

Antenna pattern:

For the outstation antenna pattern, the reference pattern described in Sections 2.2.1 or 2.2.2 has to be used.

The reference radiation pattern for omnidirectional antennas is the following:

$$G(\theta) = G_0 - 12 (\theta/\phi_3)^2$$
, dBi $0 \le \theta < \phi_3$
 $G(\theta) = G_0 - 12 - 10 \log (\theta/\phi_3)$, dBi $\phi_3 \le \theta \le 90^\circ$

where

 G_0 = maximum gain in the horizontal plane (dBi)

 θ = the radiation angle above the horizontal plane (degrees).

 φ_3 (degrees) is given by:

$$\phi_3 = \frac{1}{\alpha^2 - 0.818}$$
, degrees
$$\alpha = \frac{10^{0.1G_0} + 172.4}{191.0}$$

It should be noted that the above antenna pattern is provisional and that further study is under way in the ITU-R.

3 Determination of the need for coordination between MSS space stations and stations of the FS

The two following methods show the possibility of sharing between the two services involved and thus that it is not necessary to coordinate.

3.1 First method for the determination of the need for non-coordination between MSS space stations (space-to-Earth) and other terrestrial services sharing the same frequency band

The following table gives the coordination thresholds (in case of sharing with MSS downlink in the 1 - 3 GHz band):

- 6 -CMR95/DL/15-E

Frequency Band (MHz)	Service to be protected	Coordination threshold factors				
		Geostationary space stations power flux-density (per space station) calculation factors (NOTE 2)		Non-geostationary space stations		
				power flux-density (per space station) calculation factors (NOTE 2)		% FDP (in 1 MHz)
		P dB(W/m²) in 4 kHz	r dB/deg.	P dB(W/m²) in 4 kHz	r dB/deg.	
1 492 - 1 525	analogue FS	-152	0.5	-152	0.5	
	digital FS	-152	0.5			25
	other terrestrial services	-152	0.5	-152	0.5	
1 525 - 1 530	analogue FS	-152	0.5	-152	0.5	
	digital FS	-152	0.5			25
	other terrestrial services	-152	0.5	-152	0.5	
2 160 - 2 200	analogue FS	-152	0.5	-147	0.5	
	digital FS	-152	0.5			25
	other terrestrial services	-152	0.5	-147	0.5	
2 483.5 - 2 500	fixed	-152	0.5	-150	0.65	
	other terrestrial services	-152	0.5	-150	0.65	
2 500 - 2 520	analogue FS	-152	0.5	-152	0.5	
	digital FS	-152	0.5			25
	other terrestrial services	-152	0.5	-152	0.5	
2 520 - 2 535	analogue FS	-160	0.75	-152	0.5	
	digital FS	-160	0.75			25
	other terrestrial services	-160	0.75	-152	0.5	

NOTE 1 – The calculation of FDP (Fractional Degradation in Performance) is contained in Section 2.1, using reference FS parameters contained in Sections 2.2.1 and 2.2.3

NOTE 2 – The following formula are used for power flux-density:

 $P dB(W/m^2/4 kHz)$

for $0^{\circ} \le \delta \le 5^{\circ}$

 $P + r(\delta-5) dB(W/m^2/4 kHz)$

for $5^{\circ} < \delta < 25^{\circ}$

 $P + 20r dB(W/m^2/4 kHz)$

for $25^{\circ} \le \delta \le 90^{\circ}$

where δ is the angle of arrival (degrees).

- 7 -CMR95/DL/15-E

The limits relate to the power flux-density which would be obtained under assumed free-space propagation conditions.

NOTE 3 — The coordination threshold in the band 2 160 - 2 200 MHz to protect other terrestrial services do not apply to the terrestrial component of the FPLMTS as the satellite and the terrestrial components are not intended to operate on common frequencies within this band.

[NOTE 4 – The coordination threshold factors applicable to other terrestrial services may be reviewed at a future conference as necessary.]

3.2 A system specific methodology (SSM) to be used in determining the need for coordination of NGSO MSS systems in the space-to-Earth MSS allocations with the fixed service

3.2.1 Introduction

The purpose of the system specific methodology (SSM) is to allow a detailed assessment of the need to coordinate frequency assignments to transmitting non-GSO MSS space stations with frequency assignments to receiving FS stations in an FS network of a potentially affected administration. The SSM takes into account most specific characteristics of the non-GSO MSS system and reference FS characteristics.

The SSM requires as input a characterization of the reference FS system as well as that of the non-GSO MSS satellite system as described in Section 2.2.

The computation methodology is based on the data relevant statistics of interference caused by the non-GSO MSS constellation to the given reference FS system. For a given administration the SSM is exercized at a suitable sample of latitudes (e.g. every 5 degrees) covering the latitude range covered by the territory of that administration.

If the applicable maximum interference criteria are not exceeded coordination is not considered to be necessary.

3.2.2 FS and MSS data requirements

3.2.2.1 Analogue FS system data

It is assumed that there are M = 51 analogue stations on a route centred at a given latitude with a given trendline for the given non-GSO MSS constellation. The routes span a distance of $D = 2\,500$ km with stations spaced exactly d = 50 km apart. The azimuth angle for each station is specified by the given trendline angle and a variable angle that is uniformly distributed between $V = \pm 12.5$ degrees. Each FS station is assumed to use a high gain antenna pointed at the next station at an elevation angle of 0 degrees. The point-to-point FS station antenna gain conforms to the antenna pattern having averaged side-lobe levels as defined in Section 2.2.2.

The characteristics of the reference analogue FS system are taken to be as given in Section 2.2.2.

3.2.2.2 Digital FS system data

Only one digital FS receiver is required for the analysis as opposed to a complete route. The FS station is positioned at a given latitude pointing in a given azimuth direction. The FS station is assumed to use an antenna at an elevation angle of 0 degrees. The FS station antenna gain conforms to the antenna pattern having averaged side-lobe levels as defined in Section 2.2.1.

The characteristics of the reference digital FS system are taken to be as given in Section 2.2.1.

Point-to-multipoint systems, which use antennas that are omnidirectional in azimuth, may also need to be considered in some bands. The characteristics of the reference digital point-to-multipoint system are given in Section 2.2.3.

3.2.2.3 Non-GSO MSS data to be used

The following data (in addition to Appendix 3 to Annex 2 data) is to be supplied by the administration responsible for the non-GSO MSS system:

- a) Full information to characterize the orbital elements of each satellite in the non-GSO constellation (i.e. orbital altitude, orbital inclination, right ascension of ascending node, argument of perigee, mean anomaly, mean motion and epoch) in accordance with an appropriate reference inertial coordinate system.
- b) Full information to characterize the antenna pattern (maximum gain, antenna roll-off pattern and polarization) of each spot beam of each satellite in the non-GSO MSS constellation. The MSS spot beam antenna pattern may for example be characterized by a parabolic roll-off with a floor, any of the applicable WARC-88 or WARC-77 satellite antenna patterns or as given in Recommendation ITU-R S.672.
- c) Full information on the maximum e.i.r.p. density /4 kHz or 1 MHz in any active beam with potential carrier frequency overlap with the assumed FS receiver of any satellite visible to the FS system at all sample points in time. This information should implicitly reflect the intrasatellite and inter-satellite frequency reuse plans as well as satellite spot beam traffic loading taking into account the expected geographical distribution of traffic for the MSS system.

 In general if a CDMA/FDMA access scheme is employed on the non-GSO MSS constellation, then potentially all beams of all visible satellites may operate co-frequency. However if a TDMA/FDMA or FDMA access scheme is employed on the non-GSO MSS constellation, then only a subset of beams on visible satellites will operate co-frequency.
- d) In the absence of such information c) above, each spot beam of each satellite is to be assumed to have frequency overlap with the FS receiver and to be loaded to a level given by:
 - i) either the absolute worst case maximum MSS traffic loading (MAX) for any spot beam (which would result in a considerable overestimation of interference power into the FS system);
 - ii) or the mean MSS loading (MEAN) for any spot beam obtained by dividing the total instantaneous maximum satellite traffic capacity by the number of spot beams on the satellite (which would underestimate the effect of interference power into the FS system);
 - iii) or by a random variation between the above MAX and MEAN levels (which would mediate between i) and ii) to more accurately reflect the effect of interference power into the FS system);
- e) For non-GSO MSS systems which employ CDMA/FDMA, all visible satellites could be considered in the computation of aggregate interference to the victim FS station. For non-GSO MSS systems which employ TDMA/FDMA or FDMA, one satellite randomly selected among visible satellites or the highest elevation angle satellite could be considered in the computation of aggregate interference to the victim FS station.

3.2.3 Methodology for calculating interference

The SSM simulates the interference into the FS network from the non-GSO satellite constellation(s) as follows.

The SSM calculates the position and velocity vectors of the satellites of the NGSO satellite system and stations of FS system at each time instance.

At each time sample the SSM calculates the total interfering power at each victim FS station from all active spots with active carrier frequency overlap to the given FS station receiver bandwidth from all visible and appropriately selected MSS satellites. The aggregate interference power from all active spot beams of all visible satellites visible to the FS station(s) is determined using the following equation:

$$I = \sum_{k=1}^{M} \sum_{i=1}^{N} \sum_{j=1}^{S} \frac{E_{ijk}}{L_{ik}} G^{3}(\alpha_{ijk}) G^{4}(\theta_{ik}) \frac{B_{w}}{B_{ij}} \frac{1}{F_{k}} \frac{1}{P_{ik}A}$$
(1)

where:

I = interference power (W)

i = 1 of N visible satellites to the current k^{th} FS station;

j = 1 of S of active spot beams on the visible selected MSS satellite with frequency overlap to the current FS station receiver;

k = 1 of M FS stations on an FS route;

 E_{ijk} = the maximum e.i.r.p. density per reference bandwidth input to the antenna for the jth active spot beam in its boresight direction of the ith visible selected satellite (W/reference bandwidth):

 B_{ij} = the reference bandwidth for the interfering signal from the jth active spot beam of the ith visible selected satellite (kHz);

 $G^3(\alpha_{ijk})$ = the antenna discrimination of the jth active spot beam of the ith visible selected satellite towards the kth FS station;

 α_{ijk} = the angle between the boresight pointing vector of the jth active spot beam of the ith visible selected satellite and the range vector from the kth FS station to this satellite (degrees);

 L_{ik} = free space loss at the given reference frequency from the ith visible selected satellite to the kth FS station;

 $G^4(\theta_{ik}) = k^{th}$ FS station antenna gain in the direction of the ith visible selected satellite;

 θ_{ik} = angle between the k^{th} station antenna pointing vector and the range vector from the k^{th} station to the i^{th} visible selected satellite (degrees);

 B_w = receiver bandwidth of the victim FS station;

A = averaging factor to take into account MSS carrier frequency, power or time variability;

 F_k = feed loss for the k^{th} FS station;

 P_{ik} = polarization advantage factor between i^{th} MSS satellite and k^{th} FS station.

The averaging factor A may be applicable to reflect dynamic frequency, time or power variations in MSS traffic levels in a given reference bandwidth (due to for example use of voice activation, duty cycle, power control etc. as appropriate for the concerned non-GSO MSS system).

The polarization advantage P_{ik} of 3 dB is to be used only if the range vector from the k^{th} FS station to the i^{th} MSS satellite is within the 3 dB beamwidth of the FS antenna pointing vector.

3.2.4 Applicable interference criteria

3.2.4.1 Analogue FS

The SSM calculates the interference statistics based on the aggregate interference noise power accumulated over all stations as calculated at each sample point. The interference statistics indicate the probability that the aggregate received interference noise power exceeds a given interference level.

It was assumed that there are 51 analogue stations on a route centred at a given latitude. The routes span a distance of 2 500 km with stations spaced exactly 50 km apart. The azimuth angle for each station is specified by a given trendline angle and a variable angle that is uniformly distributed between ± 12.5 degrees. The analysis considers trendline angles that vary between 10 degrees and 170 degrees in 20 degree steps. Each station is assumed to use a high gain antenna pointed at the next station at an elevation angle of 0 degrees.

The SSM calculates the interference statistics based on the aggregate interference noise power calculated at each sample point. The interference statistics show the probability that the aggregate received interference noise power exceeds a given interference level. The interference interval is then mapped to the interference noise power in a 4 kHz telephony channel by:

$$N_{ch} = \frac{N_T}{kTB}I\tag{3}$$

where:

 N_T = thermal noise power introduced in a 4 kHz telephony channel at a station

= 25 picowatts psophometrically weighted at a point of zero relative level (pW0p)

k = Boltzmann's constant

T = station receiving system noise temperature

B = reference BW = 4 kHz

I = aggregate received interference noise power in the reference BW.

In order to assess if coordination is triggered or not with respect to analogue FS systems, the computed N_{ch} is compared with respect to the following criteria:

- 50 000 pW0p for 0.01% of time;
- 1 000 pW0p for 20% of time.

3.2.4.2 Digital FS system

For the digital FS case the SSM calculates the fractional degradation in performance FDP (as described in Section 2.1).

In order to assess if coordination is triggered or not with respect to digital FS systems, the computed FDP is compared with respect to the applicable criterion of 25%.

Jean MAHÉ Chairman 5C ad hoc 3, Box No. 819

INTERNATIONAL TELECOMMUNICATION UNION



WORLD RADIOCOMMUNICATION CONFERENCE Document DL/16-E 10 November 1995 Original: English

GENEVA,

23 OCTOBER

17 NOVEMBER 1995

English only

C5 AD HOC 3

Note by the Chairman of C5 ad hoc 3

DRAFT ANNEX AAA

Coordination Areas for Mobile Earth Stations Operating Below 3 GHz and Earth Stations
Providing Feeder Links for Non-Geostationary Satellites Operating
in the Mobile-Satellite Service

1. Objectives

[In regard to Section III and IV, Paragraphs 3.1 and 4.1, of the Annex to Resolution 46 (WRC-95),] this [Annex] specifies the coordination area (see No. [165]) for mobile earth stations as well as earth stations providing feeder links for non-geostationary satellite networks operating in the mobile-satellite service. In both cases, the coordination contour (see No. [166]) associated with the coordination area is drawn to scale on an appropriate map in order to depict the coordination area and the extent to which it overlaps the territory of administrations that may be affected. Tables 1-3 specify coordination distances (see No. [167]) for certain of the frequency sharing situations and frequency bands in which the provisions of Resolution 46 (WRC-95) are applied.

The coordination area of a mobile earth station is determined as the service area in which it is intended to operate the typical earth stations, extended in all directions by the coordination distance. Tables 1 and 2 specify coordination distances for mobile earth stations operating below 1 GHz and in the 1 - 3 GHz frequency range, respectively. In the case of feeder link earth stations, the coordination contour is determined as the end points of coordination distances measured from the earth station location. Coordination distances for feeder link earth stations operating below 1 GHz are specified in Table 1. Coordination distances for feeder link earth stations operating above 5 GHz are specified in Table 3 with respect to stations in terrestrial services and, where applicable, earth stations of other satellite networks operating in the opposite direction of transmission.

2. General considerations

Two types of coordination distances are specified in Tables 1 - 3: (1) predetermined distances, and (2) distances that are to be calculated on a case-by-case basis taking into account specific parameters of the earth station for which the coordination area is being determined. Neither of these distances indicate required separation distances.

It must be emphasized that the presence or installation of another station within the coordination area of an earth station would not necessarily preclude the satisfactory operation of either the earth station or the other station, since coordination distances are based on the most unfavourable case assumptions as regards interference.

- 2 -CMR95/DL/16-E

TABLES 1 - 3 (General Description)

Tables 1-3 of Document 189 will be edited as follows to create Tables 1-3 of this Annex. These tables will be available on Monday, 13 November.

- Combine Tables 1a and 1b of Doc. 189 to form Table 1 having the heading "Table 1 Earth Stations Operating Below 1 GHz" and delete the third column ("Applicable ITU-R Recommendation").
- Combine Tables 2a and 2b of Doc. 189 to form Table 2 having the heading "Table 2 Mobile Earth Stations Operating in the 1 3 GHz Frequency Range" and delete the third column ("Applicable ITU-R Recommendation").
- Re-label the heading of Table 3 of Doc. 189 to read "Table 3 Feeder Link Earth Stations Operating with Non-Geostationary Satellite Networks" and delete the third column ("Applicable ITU-R Recommendation"). Also delete Note 1 to Table 3.
- In all Tables, edit all entries in the 1st column to obtain language more suitable for inclusion in the RR, and make any references to existing ITU-R Recommendations comply with the guidance given in Doc. 124 (Rev. 1).
- In all tables update the last column in accordance with the latest decisions of COM 5 and make the following changes to Table 3 (as agreed at the first meeting of COM 5 ad hoc 3):
 - in the 1st row (i.e., the row with a 1st-column entry "Bands in which the FSS is already allocated; earth station operating co-directionally"), modify the existing entry:

<u>A)</u> [19.2-19.7 GHz and 29-29.5 GHz]: predetermined distance - value not given (<u>in</u> DT/26);

B) [19.2-19.7 GHz and 29-29.5 GHz]: case-by-case calculation using Recommendations ITU-R IS.847 and 849.

- in the last row (i.e., the row with a 2nd column entry "Bands in which the FSS is already allocated; earth station operating in reverse direction"), modify the existing entries to read as follows:

A) [19.2-19.7 GHz]: [170 km]

B) [13/11 GHz]: 225 km

C) [7 GHz]: 300 km

INTERNATIONAL TELECOMMUNICATION UNION

WRC-95

WORLD RADIOCOMMUNICATION CONFERENCE Document DL/17-E 13 November 1995 Original: English

GENEVA,

23 OCTOBER

17 NOVEMBER 1995

English ONLY

COMMITTEE 4

DRAFT

CONSEQUENTIAL MODIFICATIONS TO RESOLUTION 21

As a consequence of the deletion of provision S11.16 the following modifications are proposed to the attached text of Resolution 21.

RESOLUTION NO. 21 (WARC-92)(REV. WRC-95)

IMPLEMENTATION OF CHANGES IN FREQUENCY ALLOCATIONS BETWEEN 5 900 KHZ AND 19 020 KHZ

The World Administrative Radio<u>communication</u> Conference, for Dealing with Frequency Allocations in Certain Parts of the Spectrum (Malaga-Torremolinos, 1992) Geneva, 1995,

considering

- a) that parts of the frequency bands between 5 900 kHz and 19 020 kHz which were previously allocated on an exclusive or shared basis to the fixed and mobile services have been reallocated to the broadcasting service;
- b) that some existing fixed and mobile assignments may need to be removed progressively from those reallocated bands to make way for the broadcasting service;
- c) that the assignments to be removed, termed "displaced assignments", must be reaccommodated in other appropriate frequency bands;
- d) that developing countries may require special assistance from the IFRB <u>Bureau</u>, as well as in application of Resolution 22 (WARC-92), in replacing their displaced assignments with appropriate protection;
- e) that procedures already exist in Article 12 5.11 of the Radio Regulations that may be used to this effect;

recognizing

the difficulties that administrations and the IFRB Bureau might encounter during the period of transition from the previous allocations to those made by this Conference;

resolves

- 1. that the duration of the transition period shall be from 1 April 1992 to 1 April 2007;
- 2. that, as of 1 April 1992, administrations should no longer notify any frequency assignments to stations of the fixed and mobile services in the reallocated bands. Assignments notified in these bands after 1 April 1992 shall bear a symbol to indicate that the finding will be examined by the IFRB Bureau as of 1 April 2007 in accordance with the provisions of No. 1240 S.11.31 of the Radio Regulations;
- 3. that, as of 1 April 1992, the FRB Bureau shall undertake a continuing action to review the Master International Frequency Register with the help of administrations. In this respect the FRB Bureau shall periodically consult the administrations concerning the frequency assignments to links for which another satisfactory means of telecommunication exists, with a view to either downgrading assignments of class of operation A or deleting such assignments;

- 4. that administrations shall, for assignments of class of operation A in the reallocated bands, either notify the replacement frequencies to the IFRB Bureau or request the IFRB Bureau's assistance in selecting the replacement frequencies in application of No. 1218 Articles S.7 and S.13 of the Radio Regulations; and Resolution 103 (WARC-79);
- 5. that the IFRB <u>Bureau</u> shall develop in due time a draft procedure to be used for the replacement of remaining frequency assignments and shall consult administrations in accordance with No. 1001.1 <u>Article S.14</u> of the Radio Regulations;
- 6. that the IFRB <u>Bureau</u> should modify the draft procedures taking into account, to the extent practicable, comments received from administrations and propose replacement assignments at the latest three years before 1 April 2007. In so doing, the <u>IFRB Bureau</u> shall request administrations to take appropriate action to bring their assignments in conformity with the Table of Frequency Allocations by the due date;
- 7. that a replacement frequency assignment whose basic characteristics, with the exception of the assigned frequency, have not been modified in the above process, shall keep its original date. However, if these basic characteristics of a replacement frequency assignment are different from those of the displaced assignment, the replacement assignment shall be treated in accordance with Nos. 1376 to 1380 the relevant provisions of Section II of Article S.11 of the Radio Regulations;

invites administrations

when seeking reaccommodation of the displaced assignments for their fixed and mobile services in the bands between 5 900 kHz and 19 020 kHz which have been reallocated to the broadcasting service, to make every effort to find replacement assignments in the bands allocated to the fixed and mobile services concerned.